# CONTRACT DOCUMENTS FOR:



#### ITB #18-010

# REBID: TRANSIENT RESTROOM/ DOCKMASTER BUILDING CITY MARINA at GARRISON BIGHT

JANUARY 2018

(FWC Grant Agreement #14154)

## CITY OF KEY WEST

MAYOR: CRAIG CATES

**COMMISSIONERS:** 

JIMMY WEEKLEY SAMUEL KAUFMAN

BILLY WARDLOW RICHARD G. PAYNE

MARGARET ROMERO CLAYTON LOPEZ

PREPARED BY: William P. Horn Architect / City of Key West

ITB DOCUMENTS

#### CITY OF KEY WEST

#### CONTRACT DOCUMENTS

For

## CITY MARINA at GARRISON BIGHT

# REBID: TRANSIENT RESTROOM/ DOCKMASTER BUILDING

\*\*\*\*

#### CONSISTING OF:

PROCUREMENT REQUIREMENTS
CONTRACT FORMS
CONDITIONS OF THE CONTRACT
SPECIFICATIONS
SUPPLEMENTAL INFORMATION (GRANT REQUIREMENTS)
DRAWINGS

\*\*\*\*

KEY WEST, FLORIDA

JANUARY 2018

#### **TABLE OF CONTENTS**

		<u>Page</u>
	MENT REQUIREMENTS	
	TON TO BID	
	CTIONS TO BIDDERS	
	MS	
	CKBACK AFFIDAVIT	
	ENTITIES CRIME FORM	
	KEY WEST INDEMNIFICATION	
EQUAL I	BENEFITS FOR DOMESTIC PARTNERS AFFIDAVIT	27
	F SILENCE AFFIDAVIT	
LOCAL V	/ENDOR CERTIFICATION	29
BIDDER'	S CHECKLIST	30
PART 2 - CONTRAC	ΓFORMS	
		32
FLORIDA	A PERFORMANCE BOND	34
	A PAYMENT BOND	
LICENSE	REQUIREMENT AND COST	38
PART 3 - CONDITIO	NS OF THE CONTRACT	
	L CONDITIONS	42
	MENTARY CONDITIONS	
PART 4 - <u>SPECIFICA</u>	<u>TIONS</u>	
DIVISION 1 - GENER	RAL REQUIREMENTS	
01001	GENERAL REQUIREMENTS	67
01010	SCOPE OF WORK	
01025	MEASUREMENT AND PAYMENT	
01300	SUBMITTALS	
01700	CONTRACT CLOSEOUT	
	ENTEAL DIFORMATION (DIG C ( D )	
	ENTAL INFORMATION (BIG Grant Requirements)	0.1
	NT CERTIFICATIONS AND ASSURANCES	
	NT ADDITIONAL COMPLIANCE INFORMATION	
BIG GRA	NT REGULATION 50CFR86	90
PART 6 - DRAWING		
BID DOC	UMENTS (DRAWINGS)	104

# PART 1 PROCUREMENT REQUIREMENTS

#### **INVITATION TO BID**

Sealed Bids for City of Key West (CITY) "ITB #18-010 REBID: TRANSIENT RESTROOMS / DOCKMASTER BUILDING," addressed to the City of Key West, will be received at the office of the City Clerk, 1300 White Street, Key West, Florida until 3:00 p.m., local time, on the 14th day of FEBRUARY 2018, and then will be publicly opened and read. Any bids received after the time and date specified will not be considered.

Please submit one (1) original bid package and two (2) electronic copies on USB drives with a single PDF file of the entire bid package. Bid package shall be enclosed in a sealed envelope, clearly marked on the outside "ITB #18-010 REBID: TRANSIENT RESTROOMS / DOCKMASTER BUILDING — CITY MARINA at GARRISON BIGHT" addressed and delivered to the City Clerk at the address noted above.

The CITY is seeking BIDS from qualified individuals or firms for a project consisting of construction and permitting of a new Dockmaster building with transient restrooms and laundry facility and maintenance shop located at the City Marina at Garrison Bight. All improvements must meet requirements of the Florida Building Code, current edition, the Boating Infrastructure Grant (BIG), and must be constructed to meet Florida Green Building Coalition (FGBC) certification.

The full Invitation to Bid may be obtained from Demand Star by Onvia and The City of Key West website. Please contact Demand Star at www.demandstar.com or call 1-800-711-1712 or <a href="https://www.cityofkeywest-fl.gov">www.cityofkeywest-fl.gov</a>

Each Bid must be submitted on the prescribed form. The Contractor shall be a licensed contractor by the State of Florida and submit proof of such with the Bid.

A MANDATORY PRE-BID BE HELD AT THE CITY MARINA AT GARRISON BIGHT, 1801 N ROOSEVELT BLVD., KEY WEST, FL 33040 ON JANUARY 31, 2018 @ 2:30PM

EACH BID MUST BE SUBMITTED ON THE PRESCRIBED FORM AND ACCOMPANIED BY BID SECURITY AS PRESCRIBED IN THE INSTRUCTIONS TO BIDDERS, PAYABLE TO THE CITY OF KEY WEST, FLORIDA, IN AN AMOUNT NOT LESS THAN FIVE (5) PERCENT OF THE AMOUNT BID.

The Bidder shall furnish documentation showing that they are in compliance with the licensing requirements of the State and the provisions of Chapter 66 Section 87 of the Code of Ordinances of the City of Key West; within 10 days the following the Notice of Award and must demonstrate that he holds at a minimum, the following licenses & certificates;

- A. City of Key West License, as defined in Code of Ordinances, Chapter 66, enabling the Contractor to perform the work stated herein.
- B. City of Key West Business Tax License Receipt

# THE BIDDER SHALL BE A LICENSED CONTRACTOR BY THE STATE OF FLORIDA AND SUBMIT PROOF OF SUCH WITH THE BID.

All insurance contracts and certificates of insurance shall be either executed by or countersigned by a licensed resident agent of the Surety or Insurance Company having his place of business in the State of Florida, and in all ways complying with the insurance laws of the State of Florida. Further, the said Surety or Insurance Company shall be duly licensed and qualified to do business in the State of Florida.

Before a Contract will be awarded for the work contemplated herein, the CITY will conduct such investigation as is necessary to determine the performance record and ability of the apparent low Bidder to perform the size and type of work specified under this Contract. Upon request, the Bidder shall submit such information as deemed necessary by the CITY to evaluate the Bidder's qualifications.

The CITY hereby notifies all Bidders that it will affirmatively ensure that in any contract entered into pursuant to this advertisement, Disadvantaged Business Enterprise will be afforded full opportunity to submit bids in response to this invitation and will not be discriminated against on the grounds of race, color, gender, religion, age, disability, marital status or national origin in consideration for this award.

For information concerning the proposed work please contact Karen Olson, Deputy Director Port and Marine Services by email at kolson@cityofkeywest-fl.gov. Verbal communications, per the City's "Cone of Silence" ordinance are not allowed.

As stated above at the time of the BID submittal the Bidder must provide satisfactory documentation of State Licenses. The Bidder shall furnish documentation showing that he is in compliance with the licensing requirements of County, and City licenses as would be required within ten days of the award. The successful Bidder must also be able to satisfy the City Attorney as to such insurance coverage and legal requirements as may be demanded by the Proposal in question. The CITY may reject BID for any and/or all of the following reasons: (1) for budgetary reasons, (2) if the proposer misstates or conceals a material fact in its bid, (3) if the proposal does not strictly conform to the law or is non-responsive to the bid requirements, (4) if the proposal is conditional, or (5) if a change of circumstances occurs making the purpose of the proposal unnecessary to the CITY. (6) if such rejection is in the best interest of the CITY. The CITY may also waive any minor formalities or irregularities in any proposal.

The CITY retains the right to award bid to the bidder that best meet the needs of the City.

\* \* \* \* \* \*

#### **INSTRUCTIONS TO BIDDERS**

#### 1. CONTRACT DOCUMENTS

#### A. FORMAT

The Contract Documents are divided into parts, divisions, and sections for convenient organization and reference. Generally, there has been no attempt to divide the Specification sections into work performed by the various building trades, work by separate subcontractors, or work required for separate facilities in the Project.

#### B. DOCUMENT INTERPRETATION

Separate sections contained within these Contract Documents are intended to be mutually cooperative and provide all details reasonably required for the execution of proposed work.

Should there be any doubt as to the meaning or intent of said Contract Documents, the Bidder should request of the Deputy Port and Marine Services Director, in writing (at least 8 calendar days prior to Bid opening) an interpretation thereof. Any interpretation or change in said Contract Documents will be made only in writing in the form of Addenda to the Documents, which will be furnished to all registered holders of Contract Documents. Bidders shall submit with their Bids, or indicate receipt of all Addenda. The Owner will not be responsible for any other explanation or interpretations of said Documents.

The intent of these Documents is to describe a functionally complete project (or part thereof) to be constructed in accordance with the Contract Documents. Any work, materials, or equipment that may reasonably be inferred from the Contract Documents, as being required to produce the intended result shall be supplied whether or not specifically called for.

#### C. DRAWINGS

Full set of Bid Documents (Drawings) are provided with this Invitation to Bid.

#### 2. SCOPE OF SERVICES

A general description of the work to be done is contained in the Scope of Services section.

#### 3. REQUIRED QUALIFICATIONS

The prospective Proposers must meet the statutorily prescribed requirements before award of Contract by the CITY. Proposers must hold or obtain all licenses and/or certificates as required by the State and Local Statutes in order to bid and perform the work specified herein. Additionally all proposers must meet the minimum qualifications as described in Required Qualifications section.

#### 4. BIDDER'S UNDERSTANDING

Each Proposer must inform himself of the conditions relating to the execution of the work and make himself thoroughly familiar with all the Contract Documents. Failure to do so will not relieve the successful Proposer of his obligation to enter into a Contract and complete the contemplated work in strict accordance with the Contract Documents.

Each Proposer shall inform himself of, and the Proposer awarded a Contract shall comply with, federal, state, and local laws, statutes, and ordinances relative to the execution of the work. This requirement includes, but is not limited to, applicable regulations concerning minimum wage rates, nondiscrimination in the employment of labor, protection of public and employee safety and health, environmental protection, the protection of natural resources, fire protection, permits, fees, and similar subjects.

#### 5. TYPE OF BID

#### **LUMP SUM**

The Proposal for the work is to be submitted on a lump sum basis. All items required to complete the work specified or shown on the Drawings but not included in the Proposal shall be considered incidental to those set forth in the Proposal. Payment to the Contractor will be made on the measurement of the work actually performed by the Contractor, as specified in the Contract Documents.

The Bidder further proposes to accept as full payment for the Work proposed herein, the amounts computed under the provisions of the Contract Documents and based on the following lump sum amounts. The Bidder agrees that the lump sums include all allowances for overhead and profit for each type and unit of work called for in these Contract Documents.

#### 6. <u>PREPARATION OF BIDS</u>

#### A. GENERAL

All blank spaces in the BID form must be filled in for all schedules and associated parts, as required, preferably in BLACK INK. All price information shall be clearly shown in both words and figures where required. No changes shall be made in the phraseology of the forms. Written amounts shall govern in case of discrepancy between the amounts stated in writing and the amounts stated in figures. In case of discrepancy between unit prices and extended totals, unit prices shall prevail.

Any BID shall be deemed informal which contains omissions, erasures, alterations, or additions of any kind, or prices uncalled for, or in which any of the prices are obviously unbalanced, or which in any manner shall fail to conform to the conditions of the published Invitation to BID.

Only one BID from any individual, firm, partnership, or corporation, under the same or different names, will be considered. Should it appear to the Owner that any Bidder is interested in more than one BID for work contemplated; all Bids in which such Bidder is interested will be rejected.

#### B. **SIGNATURE**

The Bidder shall sign his BID in the blank space provided therefor. If Bidder is a corporation, the legal name of the corporation shall be set forth above together with the signature of the officer or officers authorized to sign Contracts on behalf of the corporation. If Bidder is a partnership, the true name of the firm shall be set forth above together with the signature of the partner or partners authorized to sign Contracts in behalf of the partnership. If signature is by an agent, other than an officer of a corporation or a member of a partnership, a notarized power-of-attorney must be on file with the Owner prior to opening of Bids or submitted with the Bid otherwise the Bid will be regarded as not properly authorized.

#### C. SPECIAL BIDDING REQUIREMENTS

The Bidder's attention is brought to the hiring practices and licenses and permits of the City of Key West. These are defined in the addition to Article 39, ORDINANCES, PERMITS, AND LICENSES, as set forth in the Supplementary Conditions.

The Bidder shall submit with his Bid his experience record showing his experience and expertise in site construction and related work. Such experience record shall provide at least five current or recent projects of similar work, not more than 5 years old within the State Florida and preferably Monroe County. For each project the following information shall be provided:

- 1.Description and location of work.
- 2.Contract amount.
- 3.Dates work was performed.
- 4 Owner
- 5. Name of Owner's contact person and phone number.
- 6.Engineer.
- 7. Name of Engineer's contact person and phone number.

The bidder shall submit with his bid a list of items to be performed by his own labor and that performed by subcontractors or others.

#### D. ATTACHMENTS

Bidder shall complete and submit the following forms with his bid:

- 1. Anti-Kickback Affidavit
- 2. Public Entity Crimes Form
- 3. City of Key West Indemnification Form
- 4. Equal Benefits for Domestic Partners Affidavit
- 5. Cone of Silence Affidavit

- 6. Local Vender Certification
- 7. Grant Requirements

#### E. CITY OF KEY WEST LICENSE REQUIRED

Contractor is required to have a Certified or Registered General Contractors City of Key West license. License fees not to exceed \$410.00. License shall be obtained within ten (10) days of Notice of Award.

#### 7. STATE AND LOCAL SALES AND USE TAXES

Unless the Supplementary Conditions contains a statement that the CITY is exempt from state sales tax on materials incorporated into the work due to the qualification of the work under this Contract, all state and local sales and use taxes as required by the laws and statutes of the state and its political subdivisions shall be paid by the Contractor. Prices quoted in the Bid shall include all nonexempt sales and use taxes, unless provision is made in the Bid form to separately itemize the tax.

#### 8. SUBMISSION OF BID

All BIDS must be submitted no later than the time prescribed, at the place, and in the manner set forth in the Invitation to Bid. Bids must be made on the Bid forms provided herewith submit one (1) ORIGINAL bid package and two (2) FLASH DRIVES containing a single PDF file of the entire bid package.

Each Bid must be submitted in a sealed envelope, so marked as to indicate the Bidder's name and its contents without being opened, and addressed in conformance with the instructions in the Invitation to Bid.

#### 9. MODIFICATION OR WITHDRAWAL OF BID

Prior to the time and date designated for receipt of Bids, any Bid submitted may be withdrawn by notice to the party receiving Bids at the place designated for receipt of Bids. Such notice shall be in writing over the signature of the Bidder or by telegram. If by telegram, written confirmation over the signature of the Bidder shall be mailed and postmarked on or before the date and time set for receipt of Bids. No Bid may be withdrawn after the time scheduled for opening of Bids unless the time specified in paragraph AWARD OF CONTRACT of these Instructions to Bidders shall have elapsed.

#### 10. BID SECURITY

BIDS must be accompanied by cash, a certified check, or cashier's check drawn on a bank in good standing, or a bid bond issued by a Surety authorized to issue such bonds in the state where the work is located, in the amount of five (5) percent of the total amount of the Proposal submitted. This bid security shall be given as a guarantee that the Bidder will not withdraw his BID for a period of sixty (60) days after bid opening, and that if awarded the Contract, the successful Bidder will execute the attached Contract and furnish properly executed Performance and Payment Bonds, each in the full amount of the Contract price within the time specified. Agent and Surety phone numbers must be provided.

The Attorney-in-Fact who executes this bond in behalf of the Surety must attach a notarized

copy of his power-of-attorney as evidence of his authority to bind the Surety on the date of execution of the bond. Where State Statute requires, certification by a resident agent shall also be provided.

If the Bidder elects to furnish a Bid Bond, he shall use the Bid Bond form bound herewith, or one conforming substantially thereto in form and content.

#### 11. RETURN OF BID SECURITY

Within 15 days after the award of the Contract, the CITY will return the bid securities to all Bidders whose BIDS are not to be further considered in awarding the Contract. Retained bid securities will be held until the Contract has been finally executed, after which all bid securities, other than Bidders' bonds and any guarantees, which have been forfeited, will be returned to the respective Bidders whose Proposals they accompanied.

#### 12. AWARD OF CONTRACT

The Award will be made under one Contract by the Owner on the basis of the Bid from the lowest, responsive, responsible Bidder. The Owner may award entire Bid or selected line items based on the CITY's best interest and available funds a time of Award.

Within 90 calendar days after the opening of Bids, the Owner will accept one of the Bids or will act in accordance with the following paragraphs. The acceptance of the Bid will be by written notice of award, mailed to the office designated in the Bid, or delivered to the Bidder's representative. In the event of failure of the lowest responsive, responsible Bidder to sign the Contract and provide an acceptable Performance Bond, Payment Bond, insurance certificate(s) and evidence of holding required licenses and certificates, the Owner may award the Contract to the next lowest responsive, responsible Bidder. Such award, if made, will be made within 120 days after the opening of Bids. Bidders will guarantee their Bid price(s) for up to 120 calendar days after Bid opening.

The CITY reserves the right to accept or reject any or all Bids and to waive any formalities and irregularities in said Bids.

#### 13. BASIS OF AWARD

The award will be made by the Owner on the basis of the BID from the lowest, responsive, responsible BIDDER which, in the Owner's sole and absolute judgment will best serve the interest of the Owner. THE TOTAL OF BASE BID PLUS THE SUM OF OWNER SELECTED BID ALTERNATES WILL BE THE BASIS OF EVALUATING LOW BIDDER.

#### 14. <u>EXECUTION OF CONTRACT</u>

The successful Bidder shall, within 10 working days after receiving Notice of Award, sign and deliver to the Owner a Contract in the form hereto attached together with the insurance certificate examples of the bonds as required in the Contract Documents and evidence of holding required licenses and certificates. Within 10 working days after receiving the

signed Contract from the successful Bidder, the Owner's authorized agent will sign the Contract. Signature by both parties constitutes execution of the Contract.

#### 15. CONTRACT BONDS

#### A. PERFORMANCE AND PAYMENT BONDS

The successful Bidder shall file with the CITY, at the time of delivery of the signed Contract, a Performance Bond and Payment Bond on the form bound herewith, each in the full amount of the Contract price in accordance with the requirements of Florida Statutes Section 255.05 or 713.23, as applicable, as security for the faithful performance of the Contract and the payment of all persons supplying labor and materials for the construction of the work, and to cover all guarantees against defective workmanship or materials, or both, during the warranty period following the date of final acceptance of the work by the CITY. The Surety furnishing this bond shall have a sound financial standing and a record of service satisfactory to the CITY, shall be authorized to do business in the State of Florida, and shall be listed on the current U.S. Department of Treasury Circular Number 570, or amendments thereto in the Federal Register, of acceptable Sureties for federal projects. The CONTRACTOR shall supply the OWNER with phone numbers, addresses, and contacts for the Surety and their agents. Pursuant to Section 255.05(7), Florida Statutes, in lieu of the bond required by law, the contractor may file with the city an alternative form of security in the form of cash, a money order, a certified check, a cashier's check or an irrevocable letter of credit.

#### B. POWER-OF-ATTORNEY

The Attorney-in-Fact (Resident Agent) who executes this Performance and Payment Bond in behalf of the Surety must attach a notarized copy of his power-of-attorney as evidence of his authority to bind the Surety on the date of execution of the bond. All Contracts, Performance and Payment Bonds, and respective powers-of-attorney will have the same date

#### 16. FAILURE TO EXECUTE CONTRACT AND FURNISH BOND

The Bidder who has a Contract awarded to him and who fails to promptly and properly execute the Contract or furnish the required Bonds shall forfeit the bid security that accompanied his bid, and the bid security shall be retained as liquidated damages by the CITY, and it is agreed that this said sum is a fair estimate of the amount of damages the CITY will sustain in case the Bidder fails to enter into a Contract or furnish the required Bonds. Bid security deposited in the form of cash, a certified check, or cashier's check shall be subject to the same requirement as a Bid Bond.

#### 17. TIME OF COMPLETION

The time of completion of the work to be performed under this Contract is the essence of the Contract. Delays and extensions of time may only be allowed in accordance with the

provisions stated in the General Conditions. Time allowed for completion of the work authorized is stated in Bid.

When the Contractor receives a Notice to Proceed, he shall commence work as soon as possible and shall complete all work within the number of calendar days stipulated in the Contract.

The term of this contract will be **three-hundred (300)** calendar days.

#### 18. PUBLIC RECORDS REQUIREMENTS

In addition to other contract requirements provided by law, each public agency contract for services must include a provision that requires the contractor to comply with public records laws, specifically to (a) keep and maintain public records that ordinarily and necessarily would be required by the public agency in order to perform the service; (b) provide the public with access to public records on the same terms and conditions that the public agency would provide the records and at a cost that does not exceed the cost provided in this chapter or as otherwise provided by law; (c) ensure that public records that are exempt or confidential and exempt from public records disclosure requirements are not disclosed except as authorized by law; and (d) meet all requirements for retaining public records and transfer, at no cost, to the public agency all public records in possession of the contractor upon termination of the contract and destroy any duplicate public records that exempt or confidential and exempt from public records disclosure requirements. All records stored electronically must be provided to the public agency in a format that is compatible with the information technology systems of the public agency. If a contractor does not comply with a public records request, the public agency shall enforce the contract provisions in accordance with the contract.

#### 19. PERMITS AND FEES

The Bidder awarded this project shall procure and pay all permits and licenses, charges, and fees, and give notices necessary and incidental to the due and lawful prosecution of the work, the cost of which shall be included in the fee received for the work, except where separately agreed.

#### 20. BIDDER'S DECLARATION AND UNDERSTANDING

The undersigned, hereinafter called the Bidder, declares that the only persons or parties interested in this Bid are those named herein, that this Bid is, in all respects, fair and without fraud, that it is made without collusion with any official of the Owner, and that the Bid is made without any connection or collusion with any person submitting another Bid on this Contract.

The Bidder further declares that he has carefully examined the Contract Documents and that this Bid is made according to the provisions and under the terms of the Contract Documents, which Documents are hereby made a part of this Bid.

#### 21. <u>ADDENDA</u>

#### 22. SALES AND USE TAXES

The Bidder agrees that all federal, state, and local sales and use taxes are included in the stated lump sum for the work.

\* \* \* \* \*

#### **BID FORM**

To:	City of Key West, Florida		
Address:	1300 White Street, Key West, Florida 33040		
Project Title:	REBID: TRANSIENT RESTROOMS / DOCKMASTER BUILDING CITY MARINA @ GARRISON BIGHT		
Project No.:	ITB #18-010		
Bidder's person to contact for additional information on this Bid:			
Company Name:			
Contact Name & Telephone #:			
Email Address:			

#### BIDDER'S DECLARATION AND UNDERSTANDING

The undersigned, hereinafter called the Bidder, declares that the only persons or parties interested in this Bid are those named herein, that this Bid is, in all respects, fair and without fraud, that it is made without collusion with any official of the Owner, and that the Bid is made without any connection or collusion with any person submitting another Bid on this Contract.

The Bidder further declares that he has carefully examined the Contract Documents, that he has personally inspected the Project, that he has satisfied himself as to the quantities involved, including materials and equipment, and conditions of work involved, including the fact that the description of the quantities of work and materials, as included herein, is brief and is intended only to indicate the general nature of the work and to identify the said quantities with the detailed requirements of the Contract Documents, and that this Bid is made according to the provisions and under the terms of the Contract Documents, which Documents are hereby made a part of this Proposal.

The Bidder further agrees that the Owner may "non-perform" the work in the event that the low bid is in excess of available funding. Non-performance will be determined prior to Notice of Award.

The intent of the Bid Documents is to describe a functionally complete project (or part thereof) to be constructed in accordance with the Contract Documents. Any work, materials, or equipment that may reasonably be inferred from the Contract Documents, as being required to produce the intended result shall be supplied, whether or not specifically called for in the Contract Documents.

#### **CONTRACT EXECUTION**

The Bidder agrees that if this Bid is accepted, he will, within 10 days, not including Sundays and legal holidays, after Notice of Award, sign the Contract in the form annexed hereto, and will at that time, deliver evidence of holding required licenses and certificates, and will, to the extent of his Bid, furnish all machinery, tools, apparatus, and other means of construction and do the work and furnish all the materials necessary to complete all work as specified or indicated in the Contract Documents.

#### CERTIFICATES OF INSURANCE

Bidder agrees to furnish the Owner, before commencing the work under this Contract, the certificates of insurance as specified in these Documents.

#### START OF CONSTRUCTION AND CONTRACT COMPLETION TIME

The Bidder agrees to begin work within fourteen (14) calendar days after the date of the Notice to Proceed and to fully complete all work under this contract within **three-hundred (300)** calendar days, including construction of the foundation and assembly of the structure.

#### LIQUIDATED DAMAGES

In the event the Bidder is awarded the Contract and fails to complete the work within the time limit or extended time limit agreed upon, as more particularly set forth in the Contract Documents, liquidated damages shall be paid to the Owner at the rate of \$1,000.00 per day for all work awarded until the work has been satisfactorily completed as provided by the Contract Documents. Sundays and legal holidays shall be excluded in determining days in default.

Owner will recover such liquidated damages by deducting the amount owed from the final payment or any retainage held by Owner.

#### **ADDENDA**

#### SALES AND USE TAXES

The Bidder agrees that all federal, state, and local sales and use taxes are included in the stated bid prices for the work.

#### LUMP SUM WORK ITEMS

The Bidder further proposes to accept as full payment for the Work proposed herein the amounts computed under the provisions of the Contract Documents and based on the following lump sum amounts.

The Bidder agrees that the lump sums represent a true measure of labor and materials required to perform the Work, including all allowances for overhead and profit for each type of work called for in these Contract Documents. The amounts shall be shown in both words and figures. In case of discrepancy, the amount shown in words shall govern.

#### **BID SCHEDULE**

# REBID: TRANSIENT RESTROOMS / DOCKMASTER BUILDING CITY MARINA at GARRISON BIGHT

#### **LUMP SUM BID PRICE**

Bidder will complete the Work in accordance with the Contract Documents for a fixed fee price.

Pricing for each line item to be broken out into percentages as indicated.

1 /	ricing for each time tiem to be broken out this percentages as indicated.				
1.	. Mobilization, General Conditions, Permit Fees and Demobilization				
	Dockmaster / Maintenance	1LS (68%)	\$		
	Transient Restroom	1LS (32%)	\$		
2.	Grant Requirements				
	Transient Restroom	1LS (100%)	\$		
3.	Demolition (includes all lab	or, equipment and disposal for a complete j	product)		
	Dockmaster / Maintenance	1LS (50%)	\$		
	Transient Restroom	1LS (50%)	\$		
4.	Foundation (includes all lab	or, equipment and material for a complete	product)		
	Dockmaster / Maintenance	1LS (68%)	\$		
	Transient Restroom	1LS (32%)	\$		
5.	Lift, Stairs, Decking, Railin material for a complete production	gs & Building Signage (includes all labor, duct)	equipment and		
	Dockmaster / Maintenance	1LS (50%)	\$		
	Transient Restroom	1LS (50%)	\$		
6.	Building Structure & Roof	(includes all labor, equipment & material for	or a complete product)		
	Dockmaster / Maintenance	1LS (68%)	\$		
	Transient Restroom	1LS (32%)	\$		

	1 LS		\$ 25,000
14.	General Allowance (only to	be used with owner's written directive)	
-	Transient Restroom	1LS (50%)	\$
]	Dockmaster / Maintenance	1LS (50%)	\$
13.	Landscaping (includes all l	abor, equipment and material for a comple	te product)
-	Transient Restroom	1LS (50%)	\$
]	Dockmaster / Maintenance	1LS (50%)	\$
12.	Site Work (includes all laborated)	or, equipment and material for a complete p	product)
-	Transient Restroom	1LS (69%)	\$
]	Dockmaster / Maintenance	1LS (31%)	\$
11.	Plumbing (includes all laboration)	or, fixtures, equipment and material for a co	omplete product)
-	Transient Restroom	1LS (32%)	\$
]	Dockmaster / Maintenance	1LS (68%)	\$
10.	Electrical (includes all laborated)	or, fixtures, equipment and material for a co	omplete product)
-	Transient Restroom	1LS (28%)	\$
]	Dockmaster / Maintenance	1LS (72%)	\$
9. <u>N</u>	Mechanical (includes all lab	or, equipment and material for a complete	product)
-	Transient Restroom	1LS (21%)	\$
]	Dockmaster / Maintenance	1LS (79%)	\$
8. ]	Doors & Windows (include	s all labor, equipment and material for a co	mplete product)
-	Transient Restroom	1LS (32%)	\$
]	Dockmaster / Maintenance	1LS (68%)	\$
7. ]	Interior Finishes (includes a	all labor, equipment, material and disposal to	for a complete product)

#### TOTAL OF ALL EXTENDED LINE ITEMS LISTED ABOVE:

Total o	of BASE BID	lump sum items 1 - 14		\$
	(amoui	nt written in words)	Dollars &	Cents
BID A	LTERNATE	S		
ALTE	RNATE ITEN LTERNATES	IS. THE TOTAL OF BA	ACCEPT OR REJECT A SE BID PLUS THE SUM OF EVALUATING LOW I	OF OWNER SELECTED
1.			em roofing on entry canopyed in section 07617 sheet A	
	1	LS		\$
2.		wood decking (5/4 x 6 prohe stairs and covered pore	emium, square edge) in-lie ch decks.	eu of 5/4 x 6 PT wood
	1	LS		\$
3.		wood louver panels in-lies for the crawl space.	eu of PT wood lattice pa	anels (vertical pattern) to
	1	LS		\$
ALLO	OWANCE ITI	EM:		
1.	Secondary ur	nderground electrical over	specified 75'-0".	
	Per F	OOT unit price \$	<u>/lf</u>	

The Bidder shall submit a Schedule of Values with the Proposal. It shall be broken down by Technical Specification Divisions included in the Base Bid and it shall be used as a basis for payment. The Bidder will be considered non-responsive if Schedule of Values not included in Bid package.

Payment for materials and equipment authorized by the Owner in a written Change Order but not listed in the above Proposal will be provided at the suppliers invoice plus 10 %.

#### CONTRACTOR'S PROJECTED OPERATIONS LOAD AND COST ESTIMATE

List items to be performed by Contractor's own (Use additional sheets if necessary.)	forces and the estimated total cost of these items.
	-

#### **SUBCONTRACTORS**

Street

The Bidder further proposes that the following subcontracting firms or businesses will be awarded subcontracts for the following portions of the work in the event that the Bidder is awarded the Contract: Portion of Work Name City Zip Street State Portion of Work Name Street State City Zip Portion of Work Name City Street State Zip Portion of Work Name

State

Zip

City

#### **SURETY**

		whose add	dress is
Street ,	City	State	Zip
Phone	Resident Agent		
<u>BIDDER</u>			
The name of the Bidder submitting this Bid is _			
		dc	oing business
at			
Street	City	State	Zip
email address			
which is the address to which all communicationshall be sent.	ons concerned with	this Bid and with the	Contract
The names of the principal officers of the corporal of all persons interested in this Bid as principal		his Bid, or of the par	tnership, or
Name		Title	

#### If Sole Proprietor or Partnership

IN WITNESS hereto the undersign 20	ed has set his (its) hand this day of
20	
Signature of Bidder	
Signature of Bidder	
Title	
	If Corporation
	dersigned corporation has caused this instrument to be executed orized officers this day of
(SEAL)	
Name of Corporation	
	Ву
	Title
	Attest
Sworn and subscribed before this	day of
NOTARY PUBLIC, State of	
My Commission Expires:	

#### EXPERIENCE OF BIDDER

The Bidder states that he is an experienced Contractor and has completed similar projects within the last five (5) years. (List similar projects, with types, names of owners, construction costs, Engineers, and references with phone numbers. Use additional sheets if necessary.)

\* \* \* \* \* \*

#### FLORIDA BID BOND

	BOND NO
	AMOUNT: \$
KNOW ALL MEN BY THESE PRESENTS, that	
hereinafter called the PRINCIPAL, and	
a corporation duly organized under the laws of the	e State of
having its principal place of business at	
	in the State of
and authorized to do business in the State of Florida	
hereinafter called the OBLIGEE, in the sum of	
DOLLARS (\$	) for the payment for which we bind
ourselves, our heirs, executors, administrators, s	successors, and assigns, jointly and severally,
firmly by these present.	
THE CONDITION OF THIS BOND IS SUCH TI	HAT:
WHEREAS, the PRINCIPAL is herewith subm	itting his or its Bid Proposal for REBID:
Transient Restrooms / Dockmaster Building, said	d Bid Proposal, by reference thereto, being
hereby made a part hereof.	

WHEREAS, the PRINCIPAL contemplates submitting or has submitted a bid to the OBLIGEE for the furnishing of all labor, materials (except those to be specifically furnished by the CITY), equipment, machinery, tools, apparatus, means of transportation for, and the performance of the work covered in the Proposal and the detailed Specifications, entitled:

# ITB #18-010 REBID: TRANSIENT RESTROOMS / DOCKMASTER BUILDING – CITY MARINA at GARRISON BIGHT

WHEREAS, it was a condition precedent to the submission of said bid that a cashier's check, certified check, or bid bond in the amount of 5 percent of the base bid be submitted with said bid as a guarantee that the Bidder would, if awarded the Contract, enter into a written Contract with the CITY for the performance of said Contract, within 10 working days after written notice having been given of the award of the Contract.

NOW, THEREFORE, the conditions of this obligation are such that if the PRINCIPAL within 10 consecutive calendar days after written notice of such acceptance, enters into a written Contract with the OBLIGEE and furnishes the Performance and Payment Bonds, each in an amount equal to 100 percent of the base bid, satisfactory to the CITY, then this obligation shall be void; otherwise the sum herein stated shall be due and payable to the OBLIGEE and the Surety herein agrees to pay said sum immediately upon demand of the OBLIGEE in good and lawful money of the United States of America, as liquidated damages for failure thereof of said PRINCIPAL.

Signed and sealed this	day of	, 2018.	
	ByPR	RINCIPAL	
	SU	JRETY	
	By	ttornev-In-Fact	

#### ANTI-KICKBACK AFFIDAVIT

STATE OF)		
: 5	SS	
COUNTY OF)		
I, the undersigned hereby duly sworn, of be paid to any employees of the City of directly or indirectly by me or any mem	of Key West as a commission, kic	kback, reward or gift,
By:		
Sworn and subscribed before this	day of	, 20
NOTARY PUBLIC, State of	, at Large	
My Commission Expires:		

\* \* \* \* \* \*

# SWORN STATEMENT UNDER SECTION 287.133(3)(A) FLORIDA STATUTES. ON PUBLIC ENTITY CRIMES

THIS FORM MUST BE SIGNED IN THE PRESENCE OF A NOTARY PUBLIC OR OTHER OFFICER AUTHORIZED TO ADMINISTER OATHS.

nis sworn statement is sub-	mitted by
	(name of entity submitting sworn statement)
hose business address is	
and (if applicable) its Federa	al Employer Identification Number (FEIN) is
nd (if applicable) its Federa	al Employer Identification Number (FEIN) is
and (if applicable) its Federa	al Employer Identification Number (FEIN) is
	al Employer Identification Number (FEIN) is
If the entity has no FEIN, in	nclude the Social Security Number of the individual
If the entity has no FEIN, in	

- 4. I understand that a "public entity crime" as defined in Paragraph 287.133(1)(g), Florida Statutes, means a violation of any state or federal law by a person with respect to and directly related to the transaction of business with any public entity or with an agency or political subdivision of any other state or with the United States, including but not limited to, any bid or contract for goods or services to be provided to any public or an agency or political subdivision of any other state or of the United States and involving antitrust, fraud, theft, bribery, collusion, racketeering, conspiracy, material misrepresentation.
- 5. I understand that "convicted" or "conviction" as defined in Paragraph 287.133(1)(b), <u>Florida Statutes</u>, means a finding of guilt or a conviction of a public entity crime, with or without an adjudication guilt, in any federal or state trial court of record relating to charges brought by indictment information after July 1, 1989, as a result of a jury verdict, nonjury trial, or entry of a plea of guilty or nolo contendere.
- 6. I understand that an "affiliate" as defined in Paragraph 287.133(1)(a), Florida Statutes, means
  - 1. A predecessor or successor of a person convicted of a public entity crime; or
  - 2. An entity under the control of any natural person who is active in the management of the entity and who has been convicted of a public entity crime. The term "affiliate" includes those officers, directors, executives, partners, shareholders, employees, members, and agents who are active in the management of an affiliate. The ownership by one person of shares constituting controlling interest in another person, or a pooling of equipment or income among persons when not for fair market value under an arm's length agreement, shall be a prima facie case that one person controls another person. A person who knowingly enters into a joint venture with a person who has been convicted of a public entity crime in Florida during the preceding 36 months shall be considered an affiliate.

7.	I understand that a "person" as defined in Paragraph 287.133(1)(8), Florida Statutes, means any natural person or entity organized under the laws of any state or of the United States with the legal power to enter into a binding contract and which bids or applies to bid on contracts for the provision of goods or services let by a public entity, or which otherwise transacts or applies to transact business with public entity. The term "person" includes those officers, directors, executives, partners, shareholders, employees, members, and agents who are active in management of an entity.
8.	Based on information and belief, the statement which I have marked below is true in relation to the entity submitting this sworn statement. (Please indicate which statement applies).
	Neither the entity submitting this sworn statement, nor any officers, directors, executives, partners, shareholders, employees, members, or agents who are active in management of the entity, nor any affiliate of the entity have been charged with and convicted of a public entity crime subsequent to July 1, 1989, AND (Please indicate which additional statement applies.)
	There has been a proceeding concerning the conviction before a hearing of the State of Florida, Division of Administrative Hearings. The final order entered by the hearing officer did not place the person or affiliate on the convicted vendor list. (Please attach a copy of the final order.)
	The person or affiliate was placed on the convicted vendor list. There has been a subsequent proceeding before a hearing officer of the State of Florida, Division of Administrative Hearings. The final order entered by the hearing officer determined that it was in the public interest to remove the person or
	affiliate from the convicted vendor list. (Please attach a copy of the final order.)
	The person or affiliate has not been put on the convicted vendor list. (Please describe any action taken by or pending with the Department of General Services.)
	(signature)
	(Signature)
	(date)
STATE	OF
COLINI	
COUN	ΓΥ OF
	PERSONALLY APPEARED BEFORE ME, the undersigned authority,
	Who, after first being sworn by me, affixed his/her
(name o	f individual signing)
signatu	re in the space provided above on this, 20
My con	nmission expires:

NOTARY PUBLIC

#### CITY OF KEY WEST INDEMNIFICATION FORM

To the fullest extent permitted by law, the CONTRACTOR expressly agrees to indemnify and hold harmless the City of Key West, their officers, directors, agents and employees (herein called the "indemnitees") from liabilities, damages, losses and costs, including but not limited to, reasonable attorney's fees and court costs, such legal expenses to include costs incurred in establishing the indemnification and other rights agreed to in this Paragraph, to persons or property, to the extent caused by the negligence, recklessness, or intentional wrongful misconduct of the CONTRACTOR, its Subcontractors or persons employed or utilized by them in the performance of the Contract. Claims by indemnitees for indemnification shall be limited to the amount of CONTRACTOR's insurance or \$1 million per occurrence, whichever is greater. The parties acknowledge that the amount of the indemnity required hereunder bears a reasonable commercial relationship to the Contract and it is part of the project specifications or the bid documents, if any.

The indemnification obligations under the Contract shall not be restricted in any way by any limitation on the amount or type of damages, compensation, or benefits payable by or for the CONTRACTOR under Workers' Compensation acts, disability benefits acts, or other employee benefits acts, and shall extend to and include any actions brought by or in the name of any employee of the CONTRACTOR or of any third party to whom CONTRACTOR may subcontract a part or all of the Work. This indemnification shall continue beyond the date of completion of the work.

CONTRACTOR:			SEAL:	
	Address			
	Signature			
	Print Name			
DATE:	Title			
Sworn and subsc	ribed before this	day of		, 20
NOTARY PUBL	LIC, State of	, at Larg	ge	
My Commission	Expires:			

#### **EQUAL BENEFITS FOR DOMESTIC PARTNERS AFFIDAVIT**

STATE OF)	ı	
:	SS	
COUNTY OF)		
I, the undersigned hereby duly sworn,	depose and say that the firm of _	
provides benefits to domestic partners to employees' spouses, per City of Ke		-
By:		
Sworn and subscribed before this	day of	, 20
NOTARY PUBLIC, State of	, at Large	
My Commission Expires:		

\* \* \* \* \* \*

#### **CONE OF SILENCE AFFIDAVIT**

STATE OF)	
: SS	
COUNTY OF)	
I, the undersigned hereby duly sworn, depose directors, employees and agents representing the f	· · · · ·
have read and understand the limitations and pro	
City of Key West Code of Ordinances Sec. 2-773	Cone of Silence.
Ву:	
Sworn and subscribed before me this	
day of	20
NOTARY PUBLIC, State of	at Large
My Commission Expires:	

\* \* \* \* \* \*

# LOCAL VENDOR CERTIFICATION PURSUANT TO CITY OF KEY WEST CODE OF ORDINANCES SECTION 2-798

The undersigned, as a duly authorized representative of the vendor listed herein, certifies to the best of his/her knowledge and belief, that the vendor meets the definition of a "Local Business." For purposes of this section, "local business" shall mean a business which:

- a. Principle address as registered with the FL Department of State located within 30 miles of the boundaries of the city, listed with the chief licensing official as having a business tax receipt with its principle address within 30 miles of the boundaries of the city for at least one year immediately prior to the issuance of the solicitation.
- b. Maintains a workforce of at least 50 percent of its employees from the city or within 30 miles of its boundaries.
- c. Having paid all current license taxes and any other fees due the city at least 24 hours prior to the publication of the call for bids or request for proposals.
  - Not a local vendor pursuant to Code od Ordinances Section 2-798
  - Qualifies as a local vendor pursuant to Code od Ordinances Section 2-798

If you qualify, please complete the following in support of the self-certification & submit copies of your County and City business licenses. Failure to provide the information requested will result in denial of certification as a local business.

Business Name	Phone:
Current Local Address: (P.O Box numbers may not be used to establish status)	Fax:
Length of time at this address	
Signature of Authorized Representative	Date
STATE OFCOUNTY OF	
The foregoing instrument was acknowledged before me	, of
(Name of officer or agent, title of officer or agent)	· · · · · · · · · · · · · · · · · · ·
or has produced(type of identification)	
	Signature of Notary
Return Completed form with Supporting documents to: City of Key West Purchasing	Print, Type or Stamp Name of Notary
•	Title or Rank

#### **BIDDER'S CHECKLIST**

(Note: The purpose of this checklist is to serve as a reminder of major items to be addressed in submitting a bid and is not intended to be all inclusive. It does not alleviate the Bidder from the responsibility of becoming familiar with all aspects of the Contract Documents and Proper completion and submission of his bid.)

1.	All Contract Documents thoroughly read and understood	
2.	All blank spaces in Bid filled in black ink.	
3.	Total and unit Prices added correctly.	
4.	Addenda acknowledged.	
5.	Subcontractors are named as indicated in the Bid.	
6.	Experience record included.	
7.	Bid signed by authorized officer.	
8.	Bid Bond completed and executed, including power-of-attorney, dated the same date as Bid Bond.	
9.	Bidder familiar with federal, state, and local laws, ordinances, rules and regulations affecting performance of the work.	
10.	Bidder, if successful, able to obtain and/or demonstrate possession of required licenses and certificates within (10) ten days after receiving a Notice of Award.	
11.	Bid submitted intact with the volume containing the Bidding Requirements, Contract Forms and Conditions of the Contract, one (1) original, two (2) USB drives.	
12.	Bid Documents submitted in sealed envelope and addressed and labeled in conformance with the instructions in the Invitation to Bid.	
13.	Anti-kickback Affidavit; Public Entity Crime Form; City of Key West Indemnification Equal Benefits for Domestic Partners Affidavit; Local Vendor Certification; G Requirements	_

# PART 2 CONTRACT FORMS

#### **CONTRACT AGREEMENT**

I his Contract, made and entered into day of	20,
by and between the City of Key West, hereinafter called the "Owner", ar	nd
, hereinafte	er called the "Contractor";
	,
WITNESSETH:	
The Contractor, in consideration of the sum to be paid him by the Owner	
agreements herein contained, hereby agrees at his own proper cost and e	1
and furnish all the materials, tools, labor, and all appliances, machinery,	* *
#18-010 REBID: TRANSIENT RESTROOMS / DOCKMASTER	BUILDING – CITY
MARINA at GARRISON BIGHT	
Key West, Florida to the extent of the Bid made by the Contractor, dated	the day of
, 20 , all in full comp	pliance with the Contract
Documents referred to herein.	

The PROCUREMENT REQUIREMENTS, including the signed copy of the BID FORM, the CONTRACT FORMS, the CONDITIONS OF THE CONTRACT, SPECIFICATIONS, DRAWINGS and/or SUPPLEMENTAL INFORMATION (if any), are hereby referred to and by reference made part of this Contract as fully and completely as if the same were fully set forth herein and are mutually cooperative therewith.

In consideration of the performance of the work as set forth in these Contract Documents, the Owner agrees to pay to the Contractor the amount bid in the Bid as adjusted in accordance with the Contract Documents, or as otherwise herein provided, and to make such payments in the manner and at the times provided in the Contract Documents.

The Contractor agrees to complete the work within **three-hundred (300)** calendar days and to accept as full payment hereunder the amounts computed as determined by the Contract Documents and based on the said Bid.

The Contractor agrees to remedy all defects appearing in the work or developing in the materials furnished and the workmanship performed under this Contract during the warranty period after the date of final acceptance of the work by the Owner, and further agrees to indemnify and save the Owner harmless from any costs encountered in remedying such defects.

Liquidated Damages: The Contractor recognizes that time is of the essence and that the Owner will suffer financial loss if the Work is not completed within the dates specified in the paragraphs above, plus any extensions thereof allowed, in accordance with Article 58 of the General Conditions.

In the event the Contractor fails to complete the work within the time limit or extended time limit agreed upon, as more particularly set forth in the Contract Documents, liquidated damages shall be paid at a rate of \$1,000.00 per day. Sundays and legal holidays shall be included in determining days in default.

20

This contract will automatically expire obligations remain in affect.	e upon completion of the project. Contractors warranty		
IN WITNESS WHEREOF, we, the partie	es hereto, each herewith subscribe the same this		
day of	, A.D., 20		
	CITY OF KEY WEST		
	By		
	Printed		
	Title		
	CONTRACTOR		
	By		
	Printed		
	Title		
APPROVED AS TO FORM			
Attorney for Owner			

#### **FLORIDA PERFORMANCE BOND**

BOND NO
AMOUNT: \$
KNOW ALL MEN BY THESE PRESENTS, that in accordance with Florida Statutes Section
255.05
with offices at
nereinafter called the CONTRACTOR (Principal), and
with offices at
a corporation duly organized and existing under and by virtue of the laws of the State of Florida, hereinafter called the SURETY, and authorized to transact business within the State of Florida, as SURETY, are held and firmly bound unto the CITY OF KEY WEST, hereinafter called the CITY (Obligee), in the sum of:
THE CONDITION OF THE ABOVE OBLIGATION IS SUCH THAT: WHEREAS, the CONTRACTOR has executed and entered into a certain Contract hereto attached, with the CITY, dated
<b>NOW THEREFORE,</b> the conditions of this obligation are such that if the above bounden CONTRACTOR:
1. Shall in all respects comply with the terms and conditions of said Contract and his obligation here under, including the Contract Documents (which include the scope of work and conditions as prepared by the CITY, invitation to bid, instructions to bidders, the CONTRACTOR'S bid as

accepted by the above CITY, the bid and contract performance and payment bonds (Not required as part of this contract), and all addenda, if any, issued prior to the opening of bids), being made a part of this bond by reference, at the times and in the manner prescribed in the contract; and

- 2. Promptly makes payments to all claimants, as defined in Section 255.05(1), Florida Statutes, supplying PRINCIPAL with labor, materials, or supplies, used directly or indirectly by PRINCIPAL in the prosecution of the work provided for in the contract; and
- 3. Pays CITY all losses, costs, expenses, damages, attorney's fees, including appellate proceedings, injury or loss of whatever kind and however arising including, without limitation, delay damages to which said CITY may be subject by reason of any wrongdoing, misconduct, want of care or skill, negligence, failure of performance, breach, failure to petition within the prescribed time, or default, including patent infringements, on the part of said CONTRACTOR, his agents or employees, in the execution or performance of said Contract; and
- 4. Performs the guarantee of all work and materials furnished under the contract for the time specified in the contract, then this obligation shall be void; otherwise, to remain in full force and effect for the term of said Contract.

**AND**, the said Surety for value received, hereby stipulates and agrees that no change involving any extension of time, or addition to the terms of the Contract Documents, or to the work to be performed, or materials to be furnished there under shall affect said obligation of said Surety on this Bond, and the said Surety does hereby waive notice of any such changes, extension of time, alterations, or additions of the terms of the Contract Documents, or to the work.

Any action instituted by a claimant under this bond for payment must be in accordance with the notice and time limitation provisions in Section 255.05(2), Florida Statutes.

IN WITNE	<b>ESS WHEREOF,</b> the above	e parties bonded together have executed t	his instrument
this	day of		and corporate seal
	porate party being hereto ve, pursuant to authority of	affixed and those presents duly signed be fits governing body.	y its undersigned
		CONTRACTOR	
		By:	
(SEAL)			
ATTEST			
		SURETY	
		By:	
(SEAL)			
ATTEST			

#### **FLORIDA PAYMENT BOND**

BOND NO	
AMOUNT: \$	
KNOW ALL MEN BY THESE PRESENTS, that in accordance with Florida Statutes Section	
255.05,	
with offices at	
with offices at hereinafter called the CONTRACTOR, (Principal), and	
with offices at	
a corporation duly organized and existing under and by virtue of the laws of the State of	
hereinafter called the SURETY, and authorized to transact business within the State of Florida, as SURETY, are held and firmly bound unto CITY OF KEY WEST, hereinafter called the City (Obligee), in the sum of:	
DOLLARS(), lawful money of the United States of America, for the payment of which, well and truly be made to the CITY, and the CONTRACTOR and the SURETY bind themselves and each of their heirs, executors, administrators, successors, and assigns, jointly and severally, firmly by these presents as follows:	
THE CONDITION OF THE ABOVE OBLIGATION IS SUCH THAT:	
WHEREAS, the CONTRACTOR has executed and entered into a certain Contract for	
ITB #18-010 REBID: TRANSIENT RESTROOMS / DOCKMASTER BUILDING – CITY MARINA at GARRISON BIGHT attached hereto, with	
the CITY, dated	

**NOW THEREFORE,** the conditions of this obligation are such that if the above bounden CONTRACTOR shall in all respects comply with the terms and conditions of said Contract and his obligation thereunder, including the Contract Documents, which include Scope of work and conditions prepared by the CITY, invitation to bid, instructions to bidders, the

CONTRACTOR'S bid as accepted by the CITY, the bid and contract and payment bonds (Not required as part of this contract), and all addenda, if any, issued prior to the opening of bids), and further that if said CONTRACTOR shall promptly make payments to all persons supplying materials, equipment, and/or labor, used directly or indirectly by said CONTRACTOR or subcontractors in the prosecution of the work for said contract is accordance with Florida Statutes, Section 255.05 or Section 713.23, then this obligation shall be void; otherwise to remain in full force and effect for the term of said contract, including and all guarantee periods as specifically mentioned in said Contract Documents.

**AND,** the said SURETY for value received, hereby stipulates and agrees that no change involving any extension of time, or addition to the terms of the Contract or to the work to be performed, or materials to be furnished thereunder, or in the Contract Documents and specifications accompanying the said contract shall affect said obligation of said SURETY on this Bond, and the said SURETY does hereby waive notice of any such changes, extension of time, alternations, or additions of the terms of the Contract, or to the work, to the Contract Documents, or to the specifications.

Claimant shall give written notice to the CONTRACTOR and the SURETY as required by Section 255.05 or Section 713.23, Florida Statutes. Any action instituted against the CONTRACTOR or SURETY under this bond for payment must be in accordance with the notice and time limitation provisions in Section 255.05(2) or Section 713.23, Florida Statutes.

IN WITNESS WHEREOF, the above parties bou	nded together have executed this instrument
this day of corporate party being hereto affixed and those representative, pursuant to authority of its governing	e presents duly signed by its undersigned
	CONTRACTOR
(SEAL)	By:
ATTEST	SURETY
(SEAL)	By:
ATTEST	

37

#### **LICENSE REQUIREMENT AND COST**

License required for this Project. Contractor must be general contractor or building contractor.

Cost Not To Exceed \$410.00.

#### <u>CITY OF KEY WEST BUSINESS LICENSE TAX RECEIPT</u>

A City of Key West Business License Tax Receipt is required for this project. Contractor must be general contractor, building contractor or engineering contractor. Cost not to exceed \$410.00.

City of Key West Business License Tax Receipt may be found on the city website.

http://www.keywestcity.com/egov/docs/1162843921181.htm

## PART 3 CONDITIONS OF CONTRACT

#### **GENERAL CONDITIONS**

#### **Article**

#### **DEFINITIONS**

- 1. AS APPROVED
- 2. AS SHOWN, AND AS INDICATED
- 3. BIDDER
- 4. CONTRACT DOCUMENTS
- 5. CONTRACTOR
- 6. CONTRACT COMPLETION
- 7. DAYS
- 8. DRAWINGS
- 9. ENGINEER
- 10. NOTICE
- 11. OR EQUAL
- 12. OWNER
- 13. PLANS
- 14. SPECIFICATIONS
- 15. NOTICE TO PROCEED
- 16. SUBSTANTIAL COMPLETION
- 17. WORK

#### **CONTRACT DOCUMENTS**

- 18. INTENT OF CONTRACT DOCUMENTS
- 19. DISCREPANCIES AND OMISSIONS
- 20. CHANGES IN THE WORK
- 21. EXAMINATION AND VERIFICATION OF CONTRACT DOCUMENTS
- 22. DOCUMENTS TO BE KEPT ON THE JOBSITE
- 23. ADDITIONAL CONTRACT DOCUMENTS
- 24. OWNERSHIP OF CONTRACT DOCUMENTS

#### THE ENGINEER

- 25. AUTHORITY OF THE ENGINEER
- 26. DUTIES AND RESPONSIBILITIES OF THE ENGINEER
- 27. LIMITATIONS ON ENGINEER'S RESPONSIBILITIES
- 28. REJECTED WORK
- 29. LINES AND GRADES
- 30. SUBMITTALS
- 31. DETAIL DRAWINGS AND INSTRUCTIONS

#### THE CONTRACTOR AND HIS EMPLOYEES

- 32. CONTRACTOR, AN INDEPENDENT AGENT
- 33. SUBCONTRACTING
- 34. INSURANCE AND LIABILITY
  - A. GENERAL
  - B. CONTRACTOR AND SUB-CONTRACTOR INSURANCE
  - C. COMPENSATION AND EMPLOYER'S LIABILITY INSURANCE
  - D. GENERAL LIABILITY INSURANCE (INCLUDING AUTOMOBILE)

- E. BUILDER'S RISK ALL RISK INSURANCE
- F. NO PERSONAL LIABILITY OF PUBLIC OFFICIALS
- 35. INDEMNITY
- 36. EXCLUSION OF CONTRACTOR CLAIMS
- 37. TAXES AND CHARGES
- 38. REQUIREMENTS OF STATE LAW FOR PUBLIC WORKS PROJECTS
- 39. CODES, ORDINANCES, PERMITS, AND LICENSES
- 40. SUPERINTENDENCE
- 41. RECEPTION OF ENGINEER'S

COMMUNICATIONS

- 42. SAFETY
- 43. PROTECTION OF WORK AND PROPERTY
- 44. RESPONSIBILITY OF CONTRACTOR TO ACT IN AN EMERGENCY
- 45. MATERIALS AND APPLIANCES
- 46. CONTRACTOR'S AND MANUFACTURERS' COMPLIANCE WITH STATE SAFETY, OSHA, AND OTHER CODE REQUIREMENTS
- 47. SUBSTITUTION OF MATERIALS
- 48. TESTS, SAMPLES, AND OBSERVATIONS
- 49. ROYALTIES AND PATENT
- 50. CONTRACTOR'S RIGHT TO TERMINATE CONTRACT
- 51. CORRECTION OF DEFECTIVE WORK DURING WARRANTY PERIOD

#### **PROGRESS OF THE WORK**

- 52. BEGINNING OF THE WORK
- 53. SCHEDULES AND PROGRESS REPORTS
- 54. PROSECUTION OF THE WORK
- 55. OWNER'S RIGHT TO RETAIN IMPERFECT WORK
- 56. OWNER'S RIGHT TO DO WORK
- 57. OWNER'S RIGHT TO TRANSFER

**EMPLOYMENT** 

- 58. DELAYS AND EXTENSION OF TIME
- 59. DIFFERING SITE CONDITIONS

#### **Article**

- **60. LIQUIDATED DAMAGES**
- 61. OTHER CONTRACTS
- 62. USE OF PREMISES
- 63. SUBSTANTIAL COMPLETION DATE
- 64. PERFORMANCE TESTING
- 65. OWNER'S USE OF PORTION OF THE WORK
- 66. CUTTING AND PATCHING
- 67. CLEANING UP

#### **PAYMENT**

- 68. CHANGE ORDERS
  - A. UNIT PRICE
  - B. LUMP SUM
  - C. COST REIMBURSEMENT WORK
- 69. PARTIAL PAYMENTS
  - A. GENERAL
  - B. ESTIMATE
  - C. DEDUCTION FROM ESTIMATE
  - D. QUALIFICATIONS FOR PARTIAL PAYMENT FOR MATERIALS DELIVERED
  - E. PAYMENT
- 70. CLAIMS FOR EXTRA WORK
- 71. RELEASE OF LIENS OR CLAIMS
- 72. FINAL PAYMENT
- 73. NO WAIVER OF RIGHTS
- 74. ACCEPTANCE OF FINAL PAYMENT CONSTITUTES RELEASE

\* \* \* \* \* \* \*

#### **DEFINITIONS**

Whenever in the Contract Documents the following terms are used, the intent and meaning shall be interpreted as follows:

#### 1. AS APPROVED

The words "as approved", unless otherwise qualified, shall be understood to be followed by the words "by the ENGINEER for conformance with the Contract Document".

#### 2. AS SHOWN, AND AS INDICATED

The words "as shown" and "as Indicated" shall be understood to be followed by the words "on the Drawings".

#### 3. BIDDER

The person or persons, partnership, firm, or corporation submitting a Bid for the work contemplated.

#### 4. CONTRACT DOCUMENTS

The "Contract Documents" consist of the Bidding Requirements, Contract Forms, Conditions of the Contact, Specifications, Drawings, all modifications thereof incorporated into the Documents before their execution, Change Orders, and all other requirements incorporated by specific reference thereto. These form the Contract.

#### 5. CONTRACTOR

The person or persons, partnership, firm, or corporation who enters into the Contract awarded him by the OWNER.

#### 6. CONTRACT COMPLETION

The "Contract Completion" is the date the OWNER accepts the entire work as being in compliance with the Contract Documents, or formally waives nonconforming work to extent of nonconformity, and issues the final payment in accordance with the requirements set forth in Article, "Final Payment" of these General Conditions.

#### 7. DAYS

Unless otherwise specifically stated, the term "days" will be understood to mean calendar days. Business day or working day means any day other than Saturday, Sunday, or legal holiday.

#### 8. DRAWINGS

The term "Drawings" refers to the official Drawings, profiles, cross sections, elevations, details, and other working drawings and supplementary drawings, or reproductions thereof, signed by the ENGINEER, which shows the location, character, dimensions, and details of the work to be performed. Drawings may either be bound in the same book as the balance of the Contract Documents, or bound in separate sets, and are a part of the Contract Documents, regardless of the method of binding.

#### 9. ENGINEER

The person or organization identified as such in the Contract Documents. The Term "ENGINEER" means ENGINEER or his authorized representative.

#### 10. NOTICE

The term "notice" or the requirement to notify, as used in the Contract Documents or applicable state or federal statutes, shall signify a written communication delivered in person or by registered mail to the individual, or to a member of the firm, or to an officer of the corporation for whom it is intended. Certified or registered mail shall be addressed to the last business address known to him who gives the notice.

#### 11. OR EQUAL

The term "or equal" shall be understood to indicate that the "equal" product is equivalent to or better than the product named in function, performance, reliability, quality, and general configuration. Determination of equality in reference to the project design requirements will be made by the ENGINEER. Such equal products shall not be purchased or installed by the CONTRACTOR without written authorization.

#### 12. OWNER

The person, organization, or public body identified as such in the Contract Documents.

#### 13. PLANS (See Drawings)

#### 14. SPECIFICATIONS

The term "Specifications" refers to those portions of the Contract Documents consisting of written technical descriptions of materials, equipment, construction systems, standards, and workmanship as applied to the work and certain

administrative details applicable thereto. Where standard specifications, such as those of ASTM, AASHTO, etc., have been referred to, the applicable portions of such standard specifications shall become a part of these Contract Documents. If referenced specifications conflict with specifications contained herein, the requirements contained herein shall prevail.

#### 15. NOTICE TO PROCEED

A written notices given by the OWNER to the CONTRACTOR (with a copy to the ENGINEER) fixing the date on which the Contract time will commence to run and on which the CONTRACTOR shall start to perform his obligation under the Contract Documents. The Notice to Proceed will be given within 30 days following the execution of the Contract by the OWNER.

#### 16. SUBSTANTIAL COMPLETION

"Substantial Completion" shall be that degree of completion of the Project or a defined portion of the Project, as evidenced by the ENGINEER's written notice of Substantial Completion, sufficient to provide the OWNER, at his discretion, the full-time use of the Project or defined portion of the Project for the purposes for which it was intended. "Substantial Completion" of an operating facility shall be that degree of completion that has provided a minimum of 7 continuous days of successful, trouble-free, operation, which period shall begin after all performance and acceptance testing has been successfully demonstrated to the ENGINEER. All equipment contained in the work, plus all other components necessary to enable the OWNER to operate the facility in a manner that was intended, shall be complete on the substantial completion date.

#### 17. WORK

The word "work" within these Contract Documents shall include all material, labor, tools, and all appliances, machinery, transportation, and appurtenances necessary to perform and complete the Contract, and such additional items not specifically indicated or described which can be reasonably inferred as belonging to the item described or indicated and as required by good practice to provide a complete and satisfactory system or structure. As used herein, "provide" shall be understood to mean "furnish and install, complete in-place ".

#### **CONTRACT DOCUMENTS**

#### 18. INTENT OF CONTRACT DOCUMENTS

The Contract Documents are complementary, and what is called for by one shall be as binding as if called for by all. The intent of the Documents is to describe a functionally complete Project (or part thereof) to be constructed in accordance with the Contract Documents. Any work, materials, or equipment that may reasonably be inferred from the Contract Documents as being required to produce the intended result shall be supplied whether or not specifically called for. When words which have a well-known technical or trade meaning are used to describe work, materials, or equipment, such words shall be interpreted in accordance with that meaning.

Reference to standard specifications, manuals, or codes of any technical society, organization or association, or to the laws or regulations of any governmental authority, whether such reference be specific or by implication, shall mean the latest standard specification, manual, code or laws or regulations in effect on the first published date of the Invitation to Bid, except as may be otherwise specifically stated. However, no provision of any referenced standard specification, manual or code (whether or not specifically incorporated by reference in the Contract Documents) shall be effective to change the duties and responsibilities of OWNER, CONTRACTOR, or ENGINEER, or any of their consultants, agents, or employees from those set forth in the Contract Documents, nor shall it be effective to assign to ENGINEER, or any ENGINEER's consultants, agents, or employees, any duty or authority to supervise or direct the furnishing or performance of the work or any duty or authority to undertake responsibility contrary to the provisions of Article LIMITATIONS ON ENGINEER'S RESPONSIBILITIES.

#### 19. DISCREPANCIES AND OMISSIONS

Any discrepancies or omissions found in the Contract Documents shall be reported to the ENGINEER immediately. The ENGINEER will clarify discrepancies or omissions, in writing, within a reasonable time.

In resolving inconsistencies among two or more sections of the Contract Documents, precedence shall be given in the following order:

- A. CONTRACT
- B. BID
- C. SUPPLEMENTARY CONDITIONS
- D. INVITATION TO BID
- E. INSTRUCTIONS TO BIDDERS

- F. GENERAL CONDITIONS
- G. SPECIFICATIONS
- H. DRAWINGS

Addenda shall take precedence over all sections referenced therein. Figure dimensions on Drawings shall take precedence over scale dimensions. Detailed Drawings shall take precedence over general Drawings.

#### 20. CHANGES IN THE WORK

The OWNER, without notice to the Sureties and without invalidating the Contract, may order changes in the work within the general scope of the Contract by altering, adding to, or deducting from the work, the Contract being adjusted accordingly. All such work shall be executed under the conditions of the original Contract, except as specifically adjusted at the time of ordering such change.

In giving instructions, the ENGINEER may order minor changes in the work not involving extra cost and not inconsistent with the purposes of the Project, but otherwise, except in an emergency endangering life and property, additions or deductions from the work shall be performed only in pursuance of an approved Change Order from the OWNER, countersigned by the ENGINEER.

If the work is reduced by alterations, such action shall not constitute a claim for damages based on loss of anticipated profits.

### 21. EXAMINATION AND VERIFICATION OF CONTRACT DOCUMENTS

The CONTRACTOR shall thoroughly examine and become familiar with all of the various parts of these Contract Documents and determine the nature and location of the work, the general and local conditions, and all other matters, which can in any way affect the work under this Contract. Failure to make an examination necessary for this determination shall not release the CONTRACTOR from the obligations of this Contract. No verbal agreement or conversation with any officer, agent, or employee of the OWNER or with the ENGINEER either before or after the execution of this Contract shall affect or modify any of the terms or obligations herein contained.

### 22. DOCUMENTS TO BE KEPT ON THE JOBSITE

The CONTRACTOR shall keep one copy of the Contract

Documents on the job- site, in good order, available to the ENGINEER and to his representatives.

The CONTRACTOR shall maintain on a daily basis at the jobsite, and make available to the ENGINEER on request, one current record set of the Drawings which have been accurately marked to indicate all modifications in the completed work that differ from the design information shown on the Drawings. Upon Substantial completion of the work, the CONTRACTOR shall give the ENGINEER one complete set of these marked up record Drawings.

### 23. ADDITIONAL CONTRACT DOCUMENTS

Copies of Contract Documents or Drawings may be obtained on request from the ENGINEER and by paying the actual cost of reproducing the Contract Documents or Drawings.

### 24. OWNERSHIP OF CONTRACT DOCUMENTS

All portions of the Contract Documents, and copies thereof furnished by the ENGINEER are instruments of service for this Project. They are not to be used on other work and are to be returned to the ENGINEER on request at the completion of the work. Any reuse of these materials without specific written verification or adaptation by the ENGINEER will be at the risk of the user and without liability or legal expense to the ENGINEER. Such user shall hold the ENGINEER harmless from any and all claims arising from any such reuse. Any such verification and adaptation shall entitle the ENGINEER to further compensation at rates to be agreed upon by the user and the ENGINEER.

#### THE ENGINEER

#### 25. AUTHORITY OF THE ENGINEER

The ENGINEER will be the OWNER's representative during the construction period. His authority and responsibility will be limited to the provisions set forth in these Contract Documents. The ENGINEER will have the Authority to reject work that does not conform to the Contract Documents. However, neither the ENGINEER's authority to act under this provision, nor any decision made by him in good faith either to exercise or not to exercise such authority, shall give rise to any duty or responsibility of the ENGINEER to the CONTRACTOR, any SUBCONTRACTOR, their

respective Sureties, any of their agents or employees, or any other person performing any of the work.

### 26. DUTIES AND RESPONSIBILITIES OF THE ENGINEER

The ENGINEER will make visits to the site at intervals appropriate to the various stages of construction to observe the progress and quality of the work and to determine, in general, if the work is proceeding in accordance with the intent of the Contract Documents. He will not make comprehensive or continuous review or observation to check quality or quantity of the work, and he will not be responsible for construction means, methods, techniques, sequences, or procedures, or for safety precautions and programs in connection with the work. Visits and observations made by the ENGINEER shall not relieve the CONTRACTOR of his obligation to conduct comprehensive inspections of the work and to furnish materials and perform acceptable work, and to provide adequate safety precautions, in conformance with the intent of the Contract.

The ENGINEER will make recommendations to the OWNER, in writing, on all claims of the OWNER or the CONTRACTOR arising from interpretation or execution of the Contract Documents. Such recommendations will be of factual and/or technical nature, and will not include the legal interpretation of the Contract Documents. Any necessary legal interpretation of the Contract Document will be made by the OWNER. Such recommendation shall be necessary before the CONTRACTOR can receive additional money under the terms of the Contract. Changes in work ordered by the ENGINEER shall be made in compliance with Article CHANGES IN THE WORK.

One or more project representatives may be assigned to observe the work. It is understood that such project representatives shall have the authority to issue notice of nonconformance and make decisions within the limitations of the authority of the ENGINEER. The CONTRACTOR shall furnish all reasonable assistance required by the ENGINEER or project representatives for proper observation of the work. The above-mentioned project representatives shall not relieve CONTRACTOR of his obligations to conduct comprehensive inspections of the work and to furnish materials and perform acceptable work, and to provide adequate safety precautions, in conformance with the intent of the Contract.

### 27. LIMITATIONS ON ENGINEER'S RESPONSIBILITIES

**ENGINEER** will responsible for not be CONTRACTOR's methods, techniques, means, sequences, or procedures of construction, or the safety precautions and programs incident thereto, **ENGINEER** will not be responsible CONTRACTOR's failure to perform or furnish the work in accordance with the Contract Documents.

ENGINEER will not be responsible for the acts or omissions of CONTRACTOR or of any SUBCONTRACTOR, any supplier, or of any other person or organization performing or furnishing any of the work.

Whenever in the Contract Documents the terms "as ordered", "as directed", "as required", "as allowed", "as approved", or terms of like effect or import are used, or the adjectives "reasonable", "suitable", "acceptable", "proper", or "satisfactory", or adjectives of like effect or import are used to describe a requirement, direction, review or judgment of ENGINEER as to the work, it is intended that such requirement, direction, review or judgment will be solely to evaluate the work for compliance with the Contract Documents (unless there is a specific statement indicating otherwise). The use of any such term or adjective shall not be effective to assign to ENGINEER any duty or authority to supervise or direct the furnishing or performance of the work or any duty or authority to undertake responsibility contrary to the provisions of this Article.

#### 28. REJECTED WORK

Any defective work or nonconforming materials or equipment that may be discovered at any time prior to expiration of the warranty period shall be removed and replaced by work which shall conform to the provisions of the Contract Documents. Any material condemned or rejected shall be removed at once from the project site.

Failure on the part of the ENGINEER to condemn or reject bad or inferior work or to note nonconforming materials or equipment on CONTRACTOR submittals shall not be construed to imply acceptance of such work. The OWNER shall reserve and retain all of its rights and remedies at law against the CONTRACTOR and its Surety for correction of any and all latent defects discovered after the guarantee period.

#### 29. LINES AND GRADES

Lines and grades shall be established as provided in the supplementary conditions. All stakes, marks, and other reference information shall be carefully preserved by the CONTRACTOR, and in case of their careless or unnecessary destruction or removal by him or his employees, such stakes, marks, and other information shall be replaced at the CONTRACTOR's expense.

#### 30. SUBMITTALS

After checking and verifying all field measurements and after complying with applicable procedures specified in Division I. **GENERAL** REQUIREMENTS, CONTRACTOR shall submit to ENGINEER, in accordance with the schedule for submittals for review, shop drawings, electrical diagrams, and catalog cuts for fabricated items and manufactured items (including mechanical and electrical equipment), which shall bear a stamp or specific written indication that CONTRACTOR has satisfied CONTRACTOR's responsibilities under the Contract Documents with respect to the review of the All submittals shall be identified as ENGINEER may require. The data shown shall be complete with respect to quantities, dimensions specified, performance and design criteria, materials, and similar data to enable ENGINEER to review the CONTRACTOR shall also submit to information. ENGINEER for review, with such promptness as to cause no delay in work, all samples required by the Contract Documents. All samples shall have been checked by and accompanied by a specific written CONTRACTOR has indication that satisfied CONTRACTOR's responsibilities under the Contract Documents with respect to the review of the submission and shall be identified clearly as to material, supplier, pertinent data such as catalog numbers and the use for which intended

Before submission of each submittal, CONTRACTOR shall have determined and verified all quantities, dimensions, specified performance criteria, installation requirements, materials, catalog numbers, and similar data with respect thereto and reviewed or coordinated each submittal with other submittals and with the requirements of the work and the Contract Documents.

At the time of each submission, CONTRACTOR shall give ENGINEER specific written notice of each variation that the submittal may have from the requirements of the Contract Documents, and, in addition, shall cause a specific notation to be made on each shop drawing submitted to ENGINEER for review and approval of each variation.

ENGINEER will review submittals with reasonable promptness, but ENGINEER's review will be only for

conformance with the design concept of the Project and for compliance with the information given in the Contract Documents and shall not extend to means, methods, techniques, sequences, or procedures of construction (except where a specific means, method, technique, sequence, or procedure of construction is indicated in or required by the Contract Documents) or to safety precautions or programs incident thereto. The review of a separate item as such will not indicate review of the assembly in which the item functions. CONTRACTOR shall make corrections required by ENGINEER, and shall return the required number of corrected copies of shop drawings and submit as required new samples for review. CONTRACTOR shall direct specific attention in writing to revisions other than the corrections called for by ENGINEER on previous submittals.

ENGINEER's review of submittals shall not relieve CONTRACTOR from the responsibility for any variation from the requirements of the Contract Documents unless CONTRACTOR has in writing called ENGINEER's attention to each such variation at the time of submission and ENGINEER has given written approval of each such variation by a specific written notation thereof incorporated therein or accompanying the shop drawing or sample approval; nor will any approval by ENGINEER relieve CONTRACTOR from responsibility for errors or omissions in the shop drawings or from responsibility for having complied with the provisions herein.

Where a shop drawing or sample is required by the specifications, any related work performed prior to ENGINEER's review and approval of the pertinent submission shall be at the sole expense and responsibility of the CONTRACTOR.

### 31. DETAIL DRAWINGS AND INSTRUCTIONS

The ENGINEER will furnish, with reasonable promptness, additional instructions by means of Drawings or otherwise, if, in the ENGINEER's opinion, such are required for the proper execution of the work. All such Drawings and instructions will be consistent with the Contract Documents, true developments thereof, and reasonably inferable there from.

#### THE CONTRACTOR AND HIS EMPLOYEES

#### 32. CONTRACTOR, AN INDEPENDENT AGENT

The CONTRACTOR shall independently perform all

work under this Contract and shall not be considered as an agent of the OWNER or of the ENGINEER, nor shall the CONTRACTOR's SUBCONTRACTOR's or employees be subagents of the OWNER or of the ENGINEER.

#### 33. SUBCONTRACTING

Unless modified in the Supplementary Conditions, within 10 days after the execution of the Contract, the CONTRACTOR shall submit to the ENGINEER the names of all SUBCONTRACTORS proposed for the work, including the names of any SUBCONTRACTORS that were submitted with the Bid. The CONTRACTOR shall not employ any SUBCONTRACTORS to which the OWNER may object to as lacking capability to properly perform work of the type and scope anticipated.

The CONTRACTOR is as fully responsible to the OWNER for the acts and omissions of his SUBCONTRACTORS and of persons either directly or indirectly employed by them as he is for the acts and omissions of persons directly employed by him.

Nothing contained in the Contract Documents shall create any contractual relationship between any SUBCONTRACTOR and the OWNER or ENGINEER.

#### 34. INSURANCE AND LIABILITY

#### A. GENERAL

The CONTRACTOR shall provide (from insurance companies acceptable to the OWNER) the insurance coverage designated hereinafter and pay all costs.

Before commencing work under this Contract, the CONTRACTOR shall furnish the OWNER with certificates of insurance specified herein showing the type, amount class of operations covered, effective dates, and date of expiration of policies, and containing substantially the following statement:

"The insurance covered by this certificate shall not be canceled or materially altered, except after 30 days' written notice has been received by the OWNER."

In case of the breach of any provision of this Article, the OWNER, at his option, may take out and maintain, at the expense of the CONTRACTOR, such insurance as the OWNER may deem proper and may deduct the cost of such insurance from any monies which may be due or become due the CONTRACTOR under this Contract.

### B. CONTRACTOR AND SUBCONTRACTOR INSURANCE

The CONTRACTOR shall not commence work under this Contract until he has obtained all the insurance required hereunder and such insurance has been reviewed by the OWNER, nor shall the CONTRACTOR allow any SUBCONTRACTOR to commence work on his subcontract until insurance specified below has been obtained. Review of the insurance by the OWNER shall not relieve or decrease the liability of the CONTRACTOR hereunder.

### C. COMPENSATION AND EMPLOYER'S LIABILITY INSURANCE

The CONTRACTOR shall maintain during the life of this Contract the statutory amount of Workmen's Compensation Insurance, in addition, Employer's Liability Insurance in an amount as specified in the Supplementary Conditions, for each occurrence, for all of his employees to be engaged in work on the project under this Contract. In case any such work is subcontracted, the Contractor shall require the Subcontractor to provide similar Worker's Compensation and Employer's Liability Insurance for all of the Subcontractor's employees to be engaged in such work.

### D. GENERAL LIABILITY INSURANCE (INCLUDING AUTOMOBILE)

The CONTRACTOR shall maintain during the life of this Contract such general liability, completed operations and products liability, and automobile liability insurance as will provide coverage for claims for damages for personal injury, including accidental death, as well as for claims for property damage, which may arise directly or indirectly from performance of the work under this Contract. The general liability policy shall include contractual liability assumed by the CONTRACTOR under Article INDEMNITY. Coverage for property damage shall be on a "broad form" basis with no exclusions for "X, C & U". The amount of insurance to be provided shall be as specified in the Supplementary Conditions.

In the event any work under this Contract is performed by a SUBCONTRACTOR, the CONTRACTOR shall be responsible for any liability directly or indirectly arising out of the work performed by the SUBCONTRACTOR, to the extent such liability is not covered by the SUBCONTRACTOR's insurance.

The OWNER and ENGINEER, their officers, agents, and employees shall be named as Additional Insured's on the CONTRACTOR's and any SUBCONTRACTOR's general liability and automobile liability insurance policies for any claims arising out of work performed under this Contract.

#### E. BUILDERS RISK ALL RISK INSURANCE

Unless otherwise modified in the Supplementary Conditions, the CONTRACTOR shall secure and maintain during the life of this Contract, Builders Risk All Risk Insurance coverage in an amount equal to the full value of the facilities/ work performed under this Contract/ construction. Such insurance shall include coverage for earthquake, landslide, flood, collapse, loss due to the results of faulty workmanship or design, and all other normally covered risks, and shall provide for losses to be paid to the CONTRACTOR, OWNER, and ENGINEER as their interests may appear.

The OWNER and ENGINEER, their officers, agents, and employees shall be named as additional insured's on the CONTRACTOR's and any SUBCONTRACTOR's Builders Risk All Risk insurance policies for any claims arising out of work performed under this Contract.

This insurance shall include a waiver of subrogation as to the ENGINEER, the OWNER, the CONTRACTOR, and their respective officers, agents, employees and SUBCONTRACTORS.

### F. NO PERSONAL LIABILITY OF PUBLIC OFFICIALS

In carrying out any of the provisions hereof in exercising any authority granted by the Contract, there will be no personal liability upon any public official.

#### 35. INDEMNITY

To the maximum extent permitted by law, the Contractor shall indemnify and defend the Owner and the Engineer, and their officers, employees, agents, and subconsultants, from all claims and losses, including attorney's fees and litigation costs arising out of property losses or health, safety, personal injury, or death claims by the Contractor, its subcontractors of any tier, and their employees, agents, or invitees regardless of the fault, breach of Contract, or negligence of the Owner or Engineer, excepting only such claims or losses that have been adjudicated to have been caused solely by the negligence of the Owner or the Engineer and regardless of whether or not the Contractor is or can be named a

party in a litigation.

### 36. EXCLUSION OF CONTRACTOR CLAIMS

In performing its obligations, the ENGINEER and its consultants may cause expense for the CONTRACTOR or its SUBCONTRACTORs and equipment or material suppliers. However, those parties and their sureties shall maintain no direct action against the ENGINEER, its officers, employees, agents, and consultants for any claim arising out of, in connection with, or resulting from the engineering services performed or required to be performed.

#### 37. TAXES AND CHARGES

The CONTRACTOR shall withhold and pay any and all sales and use taxes and all withholding taxes, whether State or Federal, and pay all Social Security charges and also all State Unemployment Compensation charges, and pay or cause to be withheld, as the case may be, any and all taxes, charges, or fees or sums whatsoever, which are now or may hereafter be required to be paid or withheld under any laws.

### 38. REQUIREMENTS OF STATE LAW FOR PUBLIC WORKS PROJECTS

When the Contract Documents concern public works of the state or any county, municipality, or political subdivision created by its laws, the applicable statutes shall apply. All parties to this Contract shall determine the contents of all applicable statutes and comply with their provisions throughout the performance of the Contract.

### 39. CODES, ORDINANCES, PERMITS AND LICENSES

The CONTRACTOR shall keep himself fully informed of all local codes and ordinances, as well as state and federal laws, which in any manner affect the work herein specified. The CONTRACTOR shall at all times comply with said codes and ordinances, laws, and regulations, and protect and indemnify the OWNER, the ENGINEER and their respective employees, and its officers and agents against any claim or liability arising from or based on the violation of any such laws, ordinances, or regulations. All permits, licenses and inspection fees necessary for prosecution and completion of the work shall be secured and paid for by the CONTRACTOR, unless otherwise specified.

#### 40. SUPERINTENDENCE

The CONTRACTOR shall keep at the project site, competent supervisory personnel. The CONTRACTOR shall designate, in writing, before starting work, a project superintendent who shall be an employee of the CONTRACTOR and shall have complete authority to represent and to act for the CONTRACTOR. ENGINEER shall be notified in writing prior to any change in superintendent assignment. CONTRACTOR shall give efficient supervision to the work, using his best skill and attention. The CONTRACTOR shall be solely responsible for all methods. construction means. techniques, procedures, and for providing adequate safety precautions and coordinating all portions of the work under the Contract. It is specifically understood and agreed that the ENGINEER, its employees and agents, shall not have control or charge of and shall not be responsible for the construction means, methods, techniques, procedures, or for providing adequate safety precautions in connection with the work under Contract.

### 41. RECEPTION OF ENGINEER'S COMMUNICATIONS

The superintendent shall receive for the CONTRACTOR all communications from the ENGINEER. Communications of major importance will be confirmed in writing upon request from the CONTRACTOR.

The ENGINEER may schedule project meetings for the purposes of discussing and resolving matters concerning the various elements of the work. Time and place for these meetings and the names of persons required to be present shall be as determined by the ENGINEER. CONTRACTOR shall comply with these attendance requirements and shall also require his SUBCONTRACTORS to comply.

#### 42. SAFETY

The CONTRACTOR shall be solely and completely responsible for conditions of the jobsite, including safety of all persons (including employees) and property during performance of the work. This requirement shall apply continuously and not be limited to normal working hours. Safety provisions shall conform to U.S. Department of Labor (OSHA), and all other applicable federal, state, county, and local laws, ordinances, codes, and regulations. Where any of these are in conflict, the more stringent requirement shall be followed. The CONTRACTOR's failure to thoroughly familiarize himself with the aforementioned safety provisions shall

not relieve him from compliance with the obligations and penalties set forth therein.

The CONTRACTOR shall develop and maintain for the duration of this Contract, a safety program that will effectively incorporate and implement all required safety provisions. The CONTRACTOR shall appoint an employee who is qualified and authorized to supervise and enforce compliance with the safety program.

The duty of the ENGINEER to conduct construction review of the work does not include review or approval of the adequacy of the CONTRACTOR's safety program, safety supervisor, or any safety measures taken in, on, or near the construction site.

The CONTRACTOR, as a part of his safety program, shall maintain at his office or other well-known place at the jobsite, safety equipment applicable to the work as prescribed by the aforementioned authorities, all articles necessary for giving first-aid to the injured, and shall establish the procedure for the immediate removal to a hospital or a doctor's care of persons (including employees) who may be injured on the jobsite.

If death or serious injuries or serious damages are caused, the accident shall be reported immediately by telephone or messenger to both the ENGINEER and the OWNER. In addition, the CONTRACTOR must promptly report in writing to the ENGINEER all accidents whatsoever arising out of, or in connection with, the performance of the work whether on, or adjacent to, the site, giving full details and statements of witnesses.

If a claim is made by anyone against the CONTRACTOR or any SUBCONTRACTOR on account of any accident, the CONTRACTOR shall promptly report the facts in writing to the ENGINEER, giving full details of the claim.

### 43. PROTECTION OF WORK AND PROPERTY

The CONTRACTOR shall at all times safely guard and protect from damage the OWNER's property, adjacent property, and his own work from injury or loss in connection with this Contract. All facilities required for protection by federal, state, or municipal laws and regulations and local conditions must be provided and maintained.

The CONTRACTOR shall protect his work and materials from damage due to the nature of the work, the

elements, carelessness of other CONTRACTORS, or from any cause whatever until the completion and acceptance of the work. All loss or damages arising out of the nature of the work to be done under these Contract Documents, or from any unforeseen obstruction or defects which may be encountered in the prosecution of the work, or from the action of the elements, shall be sustained by the CONTRACTOR.

### 44. RESPONSIBILITY OF CONTRACTOR TO ACT IN AN EMERGENCY

In case of an emergency which threatens loss or injury of property, and/or safety of life, the CONTRACTOR shall act, without previous instructions from the OWNER or ENGINEER, as the situation may warrant. The CONTRACTOR shall notify the ENGINEER thereof immediately thereafter. Any claim for compensation by the CONTRACTOR, together with substantiating documents in regard to expense, shall be submitted to the OWNER through the ENGINEER and the amount of compensation shall be determined by agreement.

#### 45. MATERIALS AND APPLIANCES

Unless otherwise stipulated, the CONTRACTOR shall provide and pay for all materials, labor, water, tools, equipment, heat, light, fuel, power, transportation, construction equipment and machinery, appliances, telephone, sanitary facilities, temporary facilities and other facilities and incidentals necessary for the execution and completion of the work.

Unless otherwise specified, all materials shall be new, and both workmanship and materials shall be of good quality. The CONTRACTOR shall, if required, furnish satisfactory evidence as to the kind and quality of materials.

In selecting and/or approving equipment for installation in the Project, the OWNER and ENGINEER assume no responsibility for injury or claims resulting from failure of the equipment to comply with applicable federal, state, and local safety codes or requirements, or the safety requirements of a recognized agency, or failure due to faulty design concepts, or defective workmanship and materials.

## 46. CONTRACTORS' AND MANUFACTURERS' COMPLIANCE WITH STATE SAFETY, OSHA, AND OTHER CODE REQUIREMENTS

The completed work shall include all necessary permanent safety devices, such as machinery guards and similar ordinary safety items required by the state and federal (OSHA) industrial authorities and applicable local and national codes. Further, any features of the work subject to such safety regulations shall be fabricated, furnished, and installed (including OWNERfurnished equipment) in compliance with these requirements. CONTRACTORS and manufacturers of equipment shall be held responsible for compliance with the requirements included herein. **CONTRACTORS** shall notify equipment suppliers SUBCONTRACTORS of the provisions of this Article.

#### 47. SUBSTITUTION OF MATERIALS

Except for OWNER-selected equipment items, and items where no substitution is clearly specified, whenever any material, article, device, product, fixture, form, type of construction, or process is indicated or specified by patent or proprietary name, by name of manufacturer, or by catalog number, such specifications shall be deemed to be used for the purpose of establishing a standard of quality and facilitating the description of the material or process desired. This procedure is not to be construed as eliminating from competition other products of equal or better quality by other manufacturers where fully suitable in design, and shall be deemed to be followed by the words "or equal". The CONTRACTOR may, in such cases, submit complete data to the ENGINEER for consideration of another material, type, or process that shall be substantially equal in every respect to that so indicated or specified. Substitute materials shall not be used unless approved in writing. The ENGINEER will be the sole judge of the substituted article or material.

#### 48. TESTS, SAMPLES, AND OBSERVATIONS

The CONTRACTOR shall furnish, without extra charge, the necessary test pieces and samples, including facilities and labor for obtaining the same, as requested by the ENGINEER. When required, the CONTRACTOR shall furnish certificates of tests of materials and equipment made at the point of manufacture by a recognized testing laboratory.

The OWNER, ENGINEER, and authorized government agents, and their representatives shall at all times be provided safe access to the work wherever it is in

preparation or progress, and the CONTRACTOR shall provide facilities for such access and for observations, including maintenance of temporary and permanent access.

If the Specifications, laws, ordinances, or any public authority require any work, to be specially tested or approved, the CONTRACTOR shall give timely notice of its readiness for observations. If any work should be covered up without approval or consent of the ENGINEER, it shall, if required by the ENGINEER, be uncovered for examination at the CONTRACTOR's expense.

Reexamination of questioned work may be ordered by the ENGINEER, and, if so ordered, the work shall be uncovered by the CONTRACTOR. If such work is found to be in accordance with the Contract Documents, the OWNER will pay the cost of uncovering, exposure, observation, inspection, testing and reconstruction. If such work is found to be not in accordance with the Contract Documents, the CONTRACTOR shall correct the defective work, and the cost of reexamination and correction of the defective work shall be paid by the CONTRACTOR.

#### 49. ROYALTIES AND PATENTS

The CONTRACTOR shall pay all royalty and licenses fees, unless otherwise specified. The CONTRACTOR shall defend all suits or claims for infringement of any patent rights and shall save the OWNER and the ENGINEER harmless from any and all loss, including reasonable attorneys' fees, on account thereof.

### 50. CONTRACTOR'S RIGHT TO TERMINATE CONTRACT

If the work should be stopped under an order of any court or other public authority for a period of more than 3 months, through no act or fault of the CONTRACTOR, its SUBCONTRACTORS, or respective employees or if the ENGINEER should fail to make recommendation for payment to the OWNER or return payment request to CONTRACTOR for revision within 30 days after it is due, or if the OWNER should fail to pay the CONTRACTOR within 30 days after time specified in Article PARTIAL PAYMENTS, any sum recommended by the ENGINEER, then the CONTRACTOR may, upon 15 days' written notice to the OWNER and the ENGINEER, stop work or terminate this Contract and recover from the OWNER payment for all acceptable work performed and reasonable termination expenses, unless said default has been remedied.

### 51. CORRECTION OF DEFECTIVE WORK DURING WARRANTY PERIOD

The CONTRACTOR hereby agrees to make, at his own expense, all repairs or replacements necessitated by defects in materials or workmanship, provided under terms of this Contract, and pay for any damage to other works resulting from such defects, which become evident within 2 years after the date of final acceptance of the work or within 2 years after the date of substantial completion established by the ENGINEER for specified items of equipment, or within such longer period as may be prescribed by law or by the terms of any applicable special guarantee required by the Contract Documents. Un-remedied defects identified for correction during the warranty period but remaining after its expiration shall be considered as part of the obligations of the warranty. Defects in material, workmanship, or equipment which are remedied as a result of obligations of the warranty shall subject the remedied portion of the work to an extended warranty period of 2 years after the defect has been remedied.

The CONTRACTOR further assumes responsibility for a similar guarantee for all work and materials provided by SUBCONTRACTORS or manufacturers of packaged equipment components. The effective date for the start of the guarantee or warranty period for equipment qualifying as substantially complete is defined in Article SUBSTANTIAL COMPLETION, AND Article SUBSTANTIAL COMPLETION DATE, in these General Conditions.

The CONTRACTOR also agrees to hold the OWNER and the ENGINEER harmless from liability of any kind arising from damage due to said defects. The CONTRACTOR shall make all repairs and replacements promptly upon receipt of written order for same from the OWNER. If the CONTRACTOR fails to make the repairs and replacements promptly, or in an emergency where delay would cause serious risk, or loss, or damage, the OWNER may have the defective work corrected or the rejected work removed and replaced, and the CONTRACTOR and his Surety shall be liable for the cost thereof.

#### PROGRESS OF THE WORK

#### **52.** BEGINNING OF THE WORK

Following execution of the Contract, the CONTRACTOR shall meet with the OWNER and ENGINEER relative to his arrangements for prosecuting

the work.

#### 53. SCHEDULES AND PROGRESS REPORTS

Prior to starting the construction, the CONTRACTOR shall prepare and submit to the ENGINEER, a progress schedule showing the dates on which each part or division of the work is expected to be started and finished, and a preliminary schedule for submittals. The progress schedule for submittals shall be brought up to date and submitted to the ENGINEER at the end of each month or at such other times the ENGINEER may request.

The CONTRACTOR shall forward to the ENGINEER, at the end of each month, an itemized report of the delivery status of major and critical items of purchased equipment and material, including shop drawings and the status of shop and field fabricated work. These progress reports shall indicate the date of the purchase order, the current percentage of completion, estimated delivery, and cause of delay, if any.

If the completion of any part of the work or the delivery of materials is behind the submitted progress schedule, the CONTRACTOR shall submit in writing a plan acceptable to the OWNER and ENGINEER for bringing the work up to schedule.

The OWNER shall have the right to withhold progress payments for the work if the CONTRACTOR fails to update and submit the progress schedule and reports as specified.

#### 54. PROSECUTION OF THE WORK

It is expressly understood and agreed that the time of beginning, rate of progress, and time of completion of the work are the essence of this Contract. The work shall be prosecuted at such time, and in or on such part or parts of the Project as may be required, to complete the Project as contemplated in the Contract Documents and the progress schedule.

If the CONTRACTOR desires to carry on work at night or outside the regular hours, he shall give timely notice to the ENGINEER to allow satisfactory arrangements to be made for observing the work in progress.

### 55. OWNER'S RIGHT TO RETAIN IMPERFECT WORK

If any part or portion of the work completed under this

Contract shall prove defective and not in accordance with the Drawings and Specifications, and if the imperfection in the same shall not be of sufficient magnitude or importance as to make the work dangerous or unsuitable, or if the removal of such work will create conditions which are dangerous or undesirable, the OWNER shall have the right and authority to retain such work but will make such deductions in the final payment therefore as may be just and reasonable.

#### 56. OWNER'S RIGHT TO DO WORK

Should the CONTRACTOR neglect to prosecute the work in conformance with the Contract Documents or neglect or refuse at his own cost to remove and replace work rejected by the ENGINEER, then the OWNER may notify the Surety of the condition, and after 10 days' written notice to the CONTRACTOR and the Surety, or without notice if an emergency or danger to the work or public exists, and without prejudice to any other right which the OWNER may have under Contract, or otherwise, take over that portion of the work which has been improperly or non-timely executed, and make good the deficiencies and deduct the cost thereof from the payments then or thereafter due the CONTRACTOR.

### 57. OWNER'S RIGHT TO TRANSFER EMPLOYMENT

If the CONTRACTOR should abandon the work or if he should persistently or repeatedly refuse or should fail to make prompt payment to SUBCONTRACTORS for material or labor, or to persistently disregard laws, ordinances, or to prosecute the work in conformance with the Contract Documents, or otherwise be guilty of a substantial violation of any provision of the Contract or any laws or ordinance, then the OWNER may, without prejudice to any other right or remedy, and after giving the CONTRACTOR and Surety 10 days' written notice, transfer the employment for said work from the CONTRACTOR to the Surety. Upon receipt of such notice, such Surety shall enter upon the premises and take possession of all materials, tools, and appliances thereon for the purpose of completing the work included under this Contract and employ by Contract or otherwise, any qualified person or persons to finish the work and provide the materials therefore, in accordance with the Contract Documents, without termination of the continuing full force and effect of this Contract. In case of such transfer of employment to such Surety, the Surety shall be paid in its own name on estimates according to the terms hereof without any right of the CONTRACTOR to make any claim for the same or any part thereof.

If, after the furnishing of said written notice to the Surety, the CONTRACTOR and the Surety still fail to make reasonable progress on the performance of the work, the OWNER may terminate the employment of the CONTRACTOR and take possession of the premises and of all materials, tools, and appliances thereon and finish the work by whatever method he may deem expedient and charge the cost thereof to the CONTRACTOR and the Surety. In such case, the CONTRACTOR shall not be entitled to receive any further payment until the work is finished. If the expense of completing the Contract, including compensation for additional managerial and administrative services, shall exceed such unpaid balance, the CONTRACTOR and the Surety shall pay the difference to the OWNER.

#### 58. DELAYS AND EXTENSION OF TIME

If the CONTRACTOR is delayed in the progress of the work by any act or neglect of the OWNER or the ENGINEER, or by any separate CONTRACTOR employed by the OWNER, or by strikes, lockouts, fire, adverse weather conditions not reasonably anticipated, or acts of Nature, and if the CONTRACTOR, within 48 hours of the start of the occurrence, gives written notice to the OWNER of the cause of the potential delay and estimate of the possible time extension involved, and within 10 days after the cause of the delay has been remedied, the CONTRACTOR gives written notice to the OWNER of any actual time extension requested as a result of the aforementioned occurrence, then the Contract time may be extended by change order for such reasonable time as the ENGINEER determines. It is agreed that no claim shall be made or allowed for any damages, loss, or expense which may arise out of any delay caused by the above referenced acts or occurrences other than claims for the appropriate extension of time.

No extension of time will be granted to the CONTRACTOR for delays occurring to parts of the work that have no measurable impact on the completion of the total work under this Contract. No extension of time will be considered for weather conditions reasonably anticipated for the area in which the work is being performed. Reasonably anticipated weather conditions will be based on official records of monthly precipitation and other historical data. Adverse weather conditions, if determined to be of a severity that would impact progress of the work, may be considered as cause for an extension of Contract completion time.

Delays in delivery of equipment or material purchased by the CONTRACTOR or his SUBCONTRACTORS, including OWNER-selected equipment shall not be considered as a just cause for delay, unless the OWNER determines that for good cause the delay is beyond the control of the CONTRACTOR. The CONTRACTOR shall be fully responsible for the timely ordering, scheduling, expediting, delivery, and installation of all equipment and materials.

Within a reasonable period after the Contractor submits to the Owner a written request for an extension of time, the Engineer will present his written opinion to the Owner as to whether an extension of time is justified, and, if so, his recommendation as to the number of days for time extension. The Owner will make the final decision on all requests for extension of time.

#### 59. DIFFERING SITE CONDITIONS

The CONTRACTOR shall promptly, and before the conditions are disturbed, give a written notice to the OWNER and ENGINEER of:

- A. Subsurface or latent physical conditions at the site which differ materially from those indicated in this contract,
- B. Unknown physical conditions at the site, of an unusual nature, which differ materially from those ordinarily encountered and generally recognized as inherent in work of the character provided for in the Contract.

The ENGINEER will investigate the site conditions promptly after receiving the notice. If the conditions do materially so differ and cause an increase or decrease in the CONTRACTOR's cost of, or the time required for, performing any part of the work under this Contract, whether or not changed as a result of the conditions, and equitable adjustment shall be made under this Article and the Contract modified in writing accordingly.

No request by the CONTRACTOR for an equitable adjustment to the Contract under this Article will be allowed, unless the CONTRACTOR has given the written notice required; provided that the time prescribed above for giving written notice may be extended by the OWNER.

No request by the CONTRACTOR for an equitable adjustment to the Contract for differing site conditions will be allowed if made after final payment under this Contract.

#### 60. LIQUIDATED DAMAGES

Should the CONTRACTOR fail to complete the work, or any part thereof, in the time agreed upon in the Contract or within such extra time as may have been allowed for delays by extensions granted as provided in the Contract, the CONTRACTOR shall reimburse the OWNER for the additional expense and damage for each calendar day, Sundays and legal holidays included, that the Contract remains uncompleted after the Contract completion date. It is agreed that the amount of such additional expense and damage incurred by reason of failure to complete the work is the per-diem rate, as stipulated in the Bid. The said amount is hereby agreed upon as a reasonable estimate of the costs which may be accrued by the OWNER after the expiration of the time of completion. It is expressly under- stood and agreed that this amount is not to be considered in the nature of a penalty, but as liquidated damages which have accrued against the CONTRACTOR. The OWNER shall have the right to deduct such damages from any amount due, or that may become due the CONTRACTOR, or the amount of such damages shall be due and collectible from the CONTRACTOR or Surety.

#### 61. OTHER CONTRACTS

The OWNER reserves the right to let other Contracts in connection with the work. The CONTRACTOR shall afford other CONTRACTORS reasonable opportunity for the introduction and storage of their materials and the execution of their work and shall properly connect and coordinate his work with theirs.

If any part of the work under this Contract depends for proper execution or results upon the work of any other CONTRACTOR, utility service company or OWNER, the CONTRACTOR shall inspect and promptly report to the ENGINEER in writing any patent or apparent defects to deficiencies in such work that render it unsuitable for such proper execution and results. The CONTRACTOR's failure to so report shall constitute and acceptance of the work by others as being fit and proper for integration with work under this Contract, except for latent or non-apparent defects and deficiencies in the work

#### 62. USE OF PREMISES

The CONTRACTOR shall confine his equipment, the storage of materials and the operation of his workers to limits shown on the Drawings or indicated by law, ordinances, permits, or directions of the ENGINEER, and shall not unreasonably encumber the premises with

his materials. The CONTRACTOR shall provide, at his own expense, the necessary rights-of-way and access to the work, which may be required outside the limits of the OWNER's property and shall furnish the ENGINEER copies of permits and agreements for use of the property outside that provided by the OWNER.

The CONTRACTOR shall not load nor permit any part of the structure to be loaded in any manner that will endanger the structure, nor shall CONTRACTOR subject any part of the work or adjacent property to stresses or pressures that will endanger it.

#### **63.** SUBSTANTIAL COMPLETION DATE

The ENGINEER may issue a written notice of substantial completion for the purpose of establishing the starting date for specific equipment guarantees, and to establish the date that the OWNER will assume the responsibility for the cost of operating such equipment. Said notice shall not be considered as final acceptance of any portion of the work or relieve the CONTRACTOR from completing the remaining work within the specified time and in full compliance with the Contract Documents. See SUBSTANTIAL COMPLETION under DEFINITIONS of these General Conditions.

#### 64. PERFORMANCE TESTING

Operating equipment and systems shall be performance tested in the presence of the ENGINEER to demonstrate compliance with the specified requirements. Performance testing shall be conducted under the specified design operating conditions or under such simulated operating conditions as recommended or approved by the ENGINEER. Schedule such testing with the ENGINEER at least one week in advance of the planned date for testing.

### 65. OWNER'S USE OF PORTIONS OF THE WORK

Following issuance of the written notice of Substantial Completion, the OWNER may initiate operation of the facility. Such use shall not be considered as final acceptance of any portion of the work, nor shall such use be considered as cause for an extension of the Contract completion time, unless authorized by a Change Order issued by the OWNER.

#### 66. CUTTING AND PATCHING

The CONTRACTOR shall do all cutting, fitting, or patching of his work that may be required to make its

several parts come together properly and fit it to receive or be received by work of other CONTRACTORS shown upon or reasonably implied by the Drawings.

#### 67. CLEANING UP

The CONTRACTOR shall, at all times, keep property on which work is in progress and the adjacent property free from accumulations of waste material or rubbish caused by employees or by the work. Upon completion of the construction, the CONTRACTOR shall remove all temporary structures, rubbish, and waste materials resulting from his operations.

#### **PAYMENT**

#### 68. PAYMENT FOR CHANGE ORDERS

The OWNER's request for quotations on alterations to the work shall not be considered authorization to proceed with the work prior to the issuance of a formal Change Order, nor shall such a request justify any delay in existing work. Quotations for alterations to the work shall include substantiating documentation with an itemized breakdown of CONTRACTOR and SUBCONTRACTOR costs, including labor, material, rentals, approved services, overhead, and profit. OWNER may require detailed cost data in order to substantiate the reasonableness of the proposed costs.

Any compensation paid in conjunction with the terms of a Change Order shall comprise total compensation due the CONTRACTOR for the work or alteration defined in the Change Order. By signing the Change Order, the CONTRACTOR acknowledges that the stipulated compensation includes payment for the work or alteration plus all payment for the interruption of schedules, extended overhead, delay, or any other impact claim or ripple effect, and by such signing specifically waives any reservation or claim for additional compensation in respect to the subject Change Order.

At the OWNER's option, payment or credit for any alterations covered by a Change Order shall be determined by one or a combination of the methods set forth in A, B, or C below, as applicable:

#### A. UNIT PRICES

Those unit prices stipulated in the Bid shall be utilized where they are applicable. In the event the Change Order results in a change in the original quantity that is materially and significantly different from the original bid quantity, a new unit price shall be negotiated upon demand of either party. Unit prices for new items included in the Change Order shall be negotiated and mutually agreed upon.

#### B. LUMP SUM

A total lump sum for the work negotiated and mutually acceptable to the CONTRACTOR and the OWNER.

Lump sum quotations for modifications to the work shall include substantiating documentation with an itemized breakdown of CONTRACTOR and SUBCONTRACTOR costs, including labor, material, rentals, approved services, overhead, and profit, all calculated as specified under "C" below.

#### C. COST REIMBURSEMENT WORK

The term "cost reimbursement" shall be understood to mean that payment for the work will be made on a time and expense basis, that is, on an accounting of the CONTRACTOR's forces, materials, equipment, and other items of cost as required and used to do the work.

If the method of payment cannot be agreed upon prior to the beginning of the work, and the OWNER directs by written Change Order that the work be done on a cost reimbursement basis, then the CONTRACTOR shall furnish labor, and furnish and install equipment and materials necessary to complete the work in a satisfactory manner and within a reasonable period of time. For the work performed, payment will be made for the documented actual cost of the following:

- 1. Labor including foremen for those hours they are assigned and participating in the cost reimbursement work (actual payroll cost, including wages, fringe benefits as established by negotiated labor agreements, labor insurance, and labor taxes as established by law). No other fixed labor burdens will be considered, unless approved in writing by the OWNER.
- 2. Material delivered and used on the designated work, including sales tax, if paid by the CONTRACTOR or his SUBCONTRACTOR.
- 3. Rental or equivalent rental cost of equipment, including necessary transportation for items having a value in excess of \$100.

Rental or equivalent rental cost will be allowed for only those days or hours during which the equipment is in actual use. Rental and transportation allowances shall not exceed the current rental rates prevailing in the locality. The rentals allowed for equipment will, in all cases, be understood to cover all fuel, supplies, repairs, and renewals, and no further allowances will be made for those items, unless specific agreement to that effect is made.

- Additional bond, as required and approved by the OWNER.
- 5. Additional insurance (other than labor insurance) as required and approved by the OWNER.

In addition to items 1 through 5 above, an added fixed fee for general overhead and profit shall be negotiated and allowed for the CONTRACTOR (or approved SUBCONTRACTOR) actually executing the Cost Reimbursement work.

An additional fixed fee shall be negotiated and allowed the CONTRACTOR for the administrative handling of portions of the work that are executed by an approved SUBCONTRACTOR. No additional fixed fee will be allowed for the administrative handling of work executed by a SUBCONTRACTOR of a SUBCONTRACTOR, unless by written permission from the OWNER.

The added fixed fees shall be considered to be full compensation, covering the cost of general supervision, overhead, profit, and any other general expense. The CONTRACTOR's records shall make clear distinction between the direct costs of work paid for on a cost reimbursement basis and the costs of other work. The CONTRACTOR shall furnish the ENGINEER report sheets in duplicate of each day's cost reimbursement work no later than the working day following the performance of said work. The daily report sheets shall itemize the materials used, and shall cover the direct cost of labor and the charges for equipment rental, whether furnished by the CONTRACTOR, SUBCONTRACTOR or other forces. The daily report sheets shall provide names or identifications and classifications of workers, the hourly rate of pay and hours worked, and also the size, type, and identification number of equipment and hours operated.

Material charges shall be substantiated by valid copies of vendors' invoices. Such invoices shall be submitted with the daily report sheets, or, if not available, they shall be submitted with subsequent daily report sheets. Said daily report sheets shall be signed by the CONTRACTOR or his authorized agent.

The OWNER reserves the right to furnish such materials and equipment as he deems expedient and the

CONTRACTOR shall have no claim for profit or added fees on the cost of such materials and equipment. To receive partial payments and final payment for cost reimbursement work, the CONTRACTOR shall submit to the ENGINEER, detailed and complete documented verification of the CONTRACTOR's and any of his SUBCONTRACTORS' actual costs involved in the cost reimbursement work. Such costs shall be submitted within 30 days after said work has been performed.

#### 69. PARTIAL PAYMENTS

#### A. GENERAL

Nothing in this Article shall be construed to affect the right, hereby reserved, to reject the whole or any part of the aforesaid work, should such work be later found not to comply with the provisions of the Contract Documents. All estimated quantities of work for which partial payments have been made are subject to review and correction on the final estimate. Payment by the OWNER and acceptance by the CONTRACTOR of partial payments based on periodic estimates of quantities of work performed shall not, in any way, constitute acceptance of the estimated quantities used as a basis for computing the amounts of the partial payments.

#### B. ESTIMATE

At least 30 days before each progress payment falls due, as specified in the Supplementary Conditions, the CONTRACTOR shall submit to the ENGINEER a detailed estimate of the amount earned during the preceding month for the separate portions of the work, and request payment. As used in this Article, the words "amount earned" means the value, on the date of the estimate for partial payment, of the work completed in accordance with the Contract Documents, and the value of approved materials delivered to the project site suitable stored and protected prior to incorporation into the work.

ENGINEER will, within 7 days after receipt of each request for payment, either indicate in writing a recommendation of payment and present the request to OWNER, or return the request to CONTRACTOR indicating in writing ENGINEER's reasons for refusing to recommend payment. In the latter case, CONTRACTOR may, within 7 days, make the necessary corrections and resubmit the request.

ENGINEER may refuse to recommend the whole or any

part of any payment if, in his opinion, it would be incorrect to make such representations to OWNER. ENGINEER may also refuse to recommend any such payment, or, because of subsequently discovered evidence or the results of subsequent inspections or tests, nullify any such payment previously recommended to such an extent as may be necessary in ENGINEER's opinion to protect the OWNER from loss because:

- 1. The work is defective, or completed work has been damaged requiring correction or replacement;
- Written claims have been made against OWNER or Liens have been filed in connection with the work:
- 3. The Contract Price has been reduced because of Change Orders;
- OWNER has been required to correct defective work or complete the work in accordance with Article OWNER'S RIGHT TO DO WORK;
- 5. Of CONTRACTOR's unsatisfactory prosecution of the work in accordance with the Contract Documents; or
- CONTRACTOR's failure to make payment to SUBCONTRACTORS or for labor, materials, or equipment.

#### C. DEDUCTION FROM ESTIMATE

Unless modified in the Supplementary Conditions, deductions from the estimate will be as described below:

1. The OWNER will deduct from the estimate, and retain as part security, 10 percent of the amount earned for work satisfactorily completed. A deduction and retainage of 10 percent will be made on the estimated amount earned for approved items of material delivered to and properly stored at the jobsite but not incorporated into the work. When the work is 50 percent complete, the OWNER may reduce the retainage to 5 percent of the dollar value of all work satisfactorily completed to date provided the CONTRACTOR is making satisfactory progress and there is no specific cause for a greater retainage. The OWNER may reinstate the retainage up to 10 percent if the OWNER determines, at his discretion, that the CONTRACTOR is not making satisfactory

progress or where there is other specific cause for such withholding.

### D. QUALIFICATION FOR PARTIAL PAYMENT FOR MATERIALS DELIVERED

Unless modified in the Supplementary Conditions, qualification for partial payment for materials delivered but not yet incorporated into the work shall be as described below:

- 1. Materials, as used herein, shall be considered to be those items which are fabricated and manufactured material and equipment. No consideration shall be given to individual purchases of less than \$200 for any one item.
- 2. To receive partial payment for materials delivered to the site, but not incorporated in the work, it shall be necessary for the CONTRACTOR to include a list of such materials on the Partial Payment Request. At his sole discretion, the ENGINEER may approve items for which partial payment is to be made. Partial payment shall be based on the CONTRACTOR's actual cost for the materials as evidenced by invoices from the supplier. Proper storage and protection shall be provided by the CONTRACTOR, and as approved by the ENGINEER. Final payment shall be made only for materials actually incorporated in the work and, upon acceptance of the work, all materials remaining for which advance payments had been made shall revert to the CONTRACTOR, unless otherwise agreed, and partial payments made for these items shall be deducted from the final payment for the work.
- CONTRACTOR warrants and guarantees that title to all work, materials, and equipment covered by any Application for Payment, whether incorporated in the Project or not, will pass to OWNER at the time of payment free and clear of all liens, claims, security interests, and encumbrances.
- 4. If requested by the ENGINEER, the CONTRACTOR shall provide, with subsequent pay requests, invoices receipted by the supplier showing payment in full has been made.

#### E. PAYMENT

After deducting the retainage and the amount of all

previous partial payments made to the CONTRACTOR from the amount earned, the amount due will be made payable to the CONTRACTOR. Recommendations for payment received by the OWNER less than 9 days prior to the scheduled day for payment will not be processed or paid until the following month.

#### 70. CLAIMS FOR EXTRA WORK

In any case where the CONTRACTOR deems additional time or compensation will become due him under this Contract for circumstances other than those defined in Article DELAYS AND EXTENSION OF TIME, the CONTRACTOR shall notify the ENGINEER, in writing, of his intention to make claim for such time or compensation before he begins the work on which he bases the claim, in order that such matters may be settled, if possible, or other appropriate action taken. The notice of claim shall be in duplicate, in writing, and shall state the circumstances and the reasons for the claim, but need not state the amount. If such notification is not given or if the ENGINEER is not afforded proper facilities by the CONTRACTOR for keeping strict account of actual cost, then the CONTRACTOR hereby agrees to waive the claim for such additional time or compensation. Such notice by the CONTRACTOR, and fact that the ENGINEER has kept account of the cost as aforesaid, shall not in any way be construed as proving the validity of the claim.

No extension of time will be granted to the CONTRACTOR for delays resulting from extra work that have no measurable impact on the completion of the total Work under this Contract. Claims for additional time or compensation shall be made in itemized detail and submitted, in writing, to the OWNER and ENGINEER within 10 days following completion of that portion of the work for which the CONTRACTOR bases his claim. Failure to make the claim for additional compensation in the manner and within the time specified above shall constitute waiver of that claim. In case the claim is found to be just, it shall be allowed and paid for as provided in Article PAYMENT FOR CHANGE ORDERS.

#### 71. RELEASE OF LIENS OR CLAIMS

The CONTRACTOR shall indemnify and hold harmless the OWNER from all claims for labor and materials furnished under this Contract. Prior to the final payment, the CONTRACTOR shall furnish to the OWNER, as part of his final payment request, a certification that all of the CONTRACTOR's obligations on the Project have been satisfied and that all monetary claims and

indebtedness have been paid. The CONTRACTOR shall furnish complete and legal effective releases or waivers, satisfactory to the OWNER, of all liens arising out of or filed in connection with the work.

#### 72. FINAL PAYMENT

Upon completion of all the work under this Contract, the CONTRACTOR shall notify the ENGINEER, in writing, that he has completed his part of the Contract and shall request final payment. Upon receipt of such notice the ENGINEER will inspect and, if acceptable, submit to the OWNER his recommendation as to acceptance of the completed work and as to the final estimate of the amount due the CONTRACTOR. Upon approval of this final estimate by the OWNER and compliance by the CONTRACTOR with provisions in Article RELEASE OF LIENS OR CLAIMS, and other provisions as may be the OWNER shall pay applicable, to CONTRACTOR all monies due him under the provisions of these Contract Documents.

#### 73. NO WAIVER OF RIGHTS

Neither the inspection by the OWNER, through the ENGINEER or any of his employees, nor any order by the OWNER for payment of money, nor any payment for, or acceptance of, the whole or any part of the work by the OWNER or ENGINEER, nor any extension of time, nor any possession taken by the OWNER or its employees, shall operate as a waiver of any provision of this Contract, or any power herein reserved to the OWNER, or any right to damages herein provided, nor shall any waiver of any breach in this Contract be held to be a waiver of any other or subsequent breach. Acceptance or final payment shall not be final and conclusive with regards to latent defects, fraud, or such gross mistakes as may amount to fraud, or as regards the OWNER's rights under the warranty.

### 74. ACCEPTANCE OF FINAL PAYMENT CONSTITUTES RELEASE

The acceptance by the CONTRACTOR of the final payment shall release the OWNER and the ENGINEER, as representatives of the OWNER, from all claims and all liability to the CONTRACTOR for all things done or furnished in connection with the work, and every act of the OWNER and others relating to or arising out of the work except claims previously made in writing and still unsettled. No payment, however, final or otherwise, shall operate to release the CONTRACTOR or his Sureties from obligations under this Contract and the Performance Bond, Payment Bond, and other bonds and warranties, as herein provided.

#### **SUPPLEMENTARY CONDITIONS**

The General Conditions are hereby revised as follows:

**ARTICLE 9 "ENGINEER"** 

Add the following:

Wherever in these Documents the word "Engineer" appears, it shall be understood to mean the City of Key West, Port & Marine Services Director or authorized representatives.

**ARTICLE 12 "OWNER"** 

Add the following:

Wherever in these Documents the word "Owner" appears, it shall be understood to mean the City of Key West whose address is 1300 White Street, Key West, Florida 33040.

ARTICLE 32 "CONTRACTOR, AN INDEPENDENT AGENT"

Add the following:

Assignment of any part or the whole of this Contract shall be subject to review and approval of the City Commission.

ARTICLE 34 "INSURANCE AND LIABILITY"

Delete Article 34 Sections A, B, C and D and replace with the following:

CONTRACTOR is to secure, pay for, and file with the City of Key West, prior to commencing any work under the Contract, all certificates for workers' compensation, public liability, and property damage liability insurance, and such other insurance coverages as may be required by specifications and addenda thereto, in at least the following minimum amounts with specification amounts to prevail if greater than minimum amounts indicated. Notwithstanding any other provision of the Contract, the CONTRACTOR shall provide the minimum limits of liability insurance coverage as follows:

Auto Liability	\$1,000,000	Combined Single Limit
General Liability	\$2,000,000	Aggregate (Per Project)
	\$2,000,000	Products Aggregate
	\$1,000,000	Any One Occurrence
	\$1,000,000	Personal Injury
	\$ 300,000	Fire Damage/Legal
Additional Umbrella Liability	\$1,000,000	Occurrence / Aggregate

CONTRACTOR shall furnish an original Certificate of Insurance indicating, and such policy providing coverage to, City of Key West named as an additional insured on a PRIMARY and NON CONTRIBUTORY basis utilizing an ISO standard endorsement at least as broad as CG 2010 (11/85) or its equivalent, (combination of CG 20 10 07 04 and CG 20 37 07 04, providing coverage for completed operations, is acceptable) including a waiver of subrogation clause in favor of City of Key West on all policies. CONTRACTOR will maintain the General Liability and Umbrella Liability insurance coverages summarized above with coverage continuing in full force including the additional insured endorsement until at least 3 years beyond completion and delivery of the work contracted herein.

Notwithstanding any other provision of the Contract, the CONTRACTOR shall maintain complete workers' compensation coverage for each and every employee, principal, officer, representative, or agent of the CONTRACTOR who is performing any labor, services, or material under the Contract. Further, CONTRACTOR shall additionally maintain the following minimum limits of coverage:

Bodily Injury Each Accident \$1,000,000 Bodily Injury by Disease Each Employee \$1,000,000 Bodily Injury by Disease Policy Limit \$1,000,000

If the work is being done on or near a navigable waterway, CONTRACTOR's workers compensation policy shall be endorsed to provide USL&H Act (WC 00 01 06 A) and Jones Act (WC 00 02 01 A) coverage if specified by the City of Key West. CONTRACTOR shall provide the City of Key West with a Certificate of Insurance verifying compliance with the workman's compensation coverage as set forth herein and shall provide as often as required by the City of Key West such certification which shall also show the insurance company, policy number, effective and expiration date, and the limits of workman's compensation coverage under each policy.

CONTRACTOR's insurance policies shall be endorsed to give 30 days written notice to the City of Key West in the event of cancellation or material change, using form CG 02 24, or its equivalent.

Certificates of Insurance submitted to the City of Key West will not be accepted without copies of the endorsements being requested. This includes additional insured endorsements, cancellation/material change notice endorsements, and waivers of subrogation. Copies of USL&H Act and Jones Act endorsements will also be required if necessary. PLEASE ADVISE YOUR INSURANCE AGENT ACCORDINGLY.

CONTRACTOR will comply with any and all safety regulations required by any agency or regulatory body including but not limited to OSHA. CONTRACTOR will notify City of Key West immediately by telephone at (305) 809-3811 any accident or injury to anyone that occurs on the jobsite and is related to any of the work being performed by the CONTRACTOR.

Article 34.E add "wind" to the list of covered perils.

#### Add the following Section:

#### G. SURETY AND INSURER QUALIFICATIONS

All bonds, insurance contracts, and certificates of insurance shall be either executed by or countersigned by a licensed resident agent of the Surety or insurance company, having his place of business in the State of Florida, and in all ways complying with the insurance laws of the State of Florida. Further, the said Surety or insurance company shall be duly licensed and qualified to do business in the State of Florida.

#### **ARTICLE 35 "INDEMNITY"**

Delete Article 35 in its entirety and replace with the following:

#### **INDEMNITY**

To the fullest extent permitted by law, the CONTRACTOR expressly agrees to indemnify and hold harmless the City of Key West, their officers, directors, agents, and employees (herein called the "indemnitees") from liabilities, damages, losses and costs, including, but not limited to, reasonable attorney's fees and court costs, such legal expenses to include costs incurred in establishing the indemnification and other rights agreed to in this Paragraph, to persons or property, to the extent caused by the negligence, recklessness, or intentional wrongful misconduct of the CONTRACTOR, its Subcontractors or persons employed or utilized by them in the performance of the Contract. Claims by indemnitees for indemnification shall be limited to the amount of CONTRACTOR's insurance or \$1 million per occurrence, whichever is greater. The parties acknowledge that the amount of the indemnity required hereunder bears a reasonable commercial relationship to the Contract and it is part of the project specifications or the bid documents, if any.

The indemnification obligations under the Contract shall not be restricted in any way by any limitation on the amount or type of damages, compensation, or benefits payable by or for the CONTRACTOR under workers' compensation acts, disability benefits acts, or other employee benefits acts, and shall extend to and include any actions brought by or in the name of any employee of the CONTRACTOR or of any third party to whom CONTRACTOR may subcontract a part or all of the Work. This indemnification shall continue beyond the date of completion of the work.

#### ARTICLE 39 "CODES, ORDINANCES, PERMITS, AND LICENSES"

#### Add the following:

#### A. NOISE ORDINANCE

City of Key West has a noise ordinance that allows working hours between 8:00 AM to 7:00 PM, Monday through Friday. No work should be performed during weekends or

City Holidays, State Holidays and National Holidays. Any construction operations outside these hours and these days will require a variance from the City of Key West Commission.

#### B. "LICENSES"

Within 10 days of Notice of Award, the successful Bidder must represent that he holds all applicable state, county, and City of Key West licenses and permits required to do business as a contractor with respect to the work described in the Contract Documents.

Further, the successful Bidder must, within 10 days of Notice of Award, furnish documentation showing that, as a minimum, he has complied with the provisions of Chapter 18 of the Code of Ordinances of the City of Key West in order to enter into the Agreement contained in the Contract Documents.

Specifically, within 10 days after Notice of Award, the successful Bidder must demonstrate that he holds, as a minimum, the following licenses and certificates:

- 1. City of Key West, Engineering Contractor Class I license.
- 2. Holds a valid occupational license issued by the City of Key West.

#### ARTICLE 42 "SAFETY"

Add the following sub article:

#### OCCUPATIONAL SAFETY AND HEALTH

The Contractor shall observe and comply with all applicable local, state, and federal occupational safety and health regulations during the prosecution of work under this Contract. In addition, full compliance by the Contractor with the U.S. Department of Labor's Occupational Safety and Health Standards, as established in Public Law 91-596, will be required under the terms of this Contract.

#### ARTICLE 43 "PROTECTION OF WORK AND PROPERTY"

Add the following Article:

#### HISTORIC PRESERVATION

The Contractor shall comply with Florida's Archives and Historic Act (Florida Statutes, Chapter 267) and the regulations of the local historic preservation board as applicable and protect against the potential loss or destruction of significant historical or archaeological data, sites, and properties in connection with the project.

#### ARTICLE 45 "MATERIALS AND APPLIANCES"

Add the following to the end of the first paragraph:

At contract completion, no equipment purchased by the contractor shall be transferred to state or local ownership.

#### ARTICLE 57 "OWNERS RIGHT TO TRANSFER EMPLOYMENT"

Add the following Article:

#### TERMINATION FOR CONVENIENCE AND RIGHT OF SUSPENSION

- A. Owner shall have the right to terminate this Contract without cause by written notice of Termination to the Contractor. In the event of such termination for convenience, the Contractor's recovery against the Owner shall be limited to that portion of the Contract amount earned through the date of termination, together with any retainage withheld and reasonable termination expenses incurred. Contractor shall not be entitled to any other or further recovery against the Owner, including, but not limited to, damages or any anticipated profit on portions of the Work not performed.
- B. Owner shall have the right to suspend all or any portions of the Work upon giving the Contractor prior written notice of such suspension. If all or any portion of the Work is so suspended, the Contractor shall be entitled to reasonable costs, expenses and time extension associated with the suspension.

#### ARTICLE 60 "LIQUIDATED DAMAGES"

Delete Article "LIQUIDATED DAMAGES" in its entirety and substitute the following:

#### LIQUIDATED DAMAGES

Should the Contractor fail to complete the work or any part thereof in the time agreed upon in the Contract Documents or within such extra time as may have been allowed for delays by extensions granted as provided in the Contract, the Contractor shall reimburse the Owner for the additional expense and damage for each calendar day, Sundays and legal holidays included, that project outlined in Contract Documents remains uncompleted after the completion date. Liquidated damages shall be assessed. It is agreed that the amount of such additional expense and damage incurred by reason of failure to complete the work is the per diem rate as stipulated in the Proposal. The said amount is hereby agreed upon as a reasonable estimate of the costs which may be accrued by the Owner after the expiration of the time of completion. It is expressly understood and agreed that this amount is not to be considered in the nature of a penalty but as liquidated damages, which have accrued against the Contractor. The Owner shall have the right to deduct such damages from any amount due or that may become due the Contractor or the amount of such damages shall be due and collectible from the Contractor or Surety.

#### ARTICLE 69 "PARTIAL PAYMENTS"

Delete the first paragraph of Article "PARTIAL PAYMENTS" and substitute the following:

No more than once each month the Contractor shall submit to the Engineer a detailed estimate of the amount earned during the preceding month for the separate portions of the work and request payment. As used in this Article the words "amount earned" means the value, on the date of the estimate, for partial payment of the work completed in accordance with the Contract Documents and the value of approved materials delivered to the project site suitably stored and protected prior to incorporation into the work.

#### Add the following:

Payment will be made by the Owner to the Contractor within 40 days receipt of the written recommendation of payment from the Engineer.

Delete Sub-Article C "DEDUCTION FROM ESTIMATE" in its entirety and substitute the following:

#### **DEDUCTION FROM ESTIMATE**

The Owner will deduct from the estimate and retain as part security 10 percent of the amount earned for work satisfactorily completed. A deduction and retainage of 10 percent will be made on the estimated amount earned for approved items of material delivered to and properly stored at the jobsite but not incorporated into the work.

Delete Sub-Article E "PAYMENT" in its entirety and substitute the following:

#### **PAYMENT**

After deducting the retainage and the amount of all previous partial payments made to the Contractor from the amount earned the amount due will be made payable to the Contractor. Recommendations for payment received by the Owner less than 40 days prior to the scheduled day for payment will not be processed or paid until the following month.

#### ARTICLE 72 "FINAL PAYMENT"

#### Add the following;

#### A. Acceptance and Final Payment.

Whenever the Contractor has completely performed the work provided for under the Contract and the Engineer has performed a final inspection and made final acceptance and subject to the terms of the Engineer will prepare a final estimate showing the value of the work as soon as the Engineer makes the necessary measurements and computations.

The Engineer will correct all prior estimates and payments in the final estimate and payment. The OWNER will pay the estimate, less any sums that the OWNER may have deducted or retained under the provisions of the Contract, as soon as practicable after final acceptance of the work, provided the Contractor has met the requirements of (1) through (5) below.

- The Contractor has agreed in writing to accept the balance due or refund the overpayment, as determined by the OWNER, as full settlement of his account under the Contract and of all claims in connection therewith, or the Contractor, accepted the balance due or refunded the overpayment, as determined by the OWNER, with the stipulation that his acceptance of such payment or the making of such refund does not constitute any bar, admission, or estoppel, or have any effect as to those payments in dispute or the subject of a pending claim between the Contractor and the OWNER. To receive payment based on a FINAL PAYMENT CERTIIFCATE, The Contractor further agrees, by submitting a FINAL PAYMENT CERTIFICATE that any pending or future arbitration claim or suit is limited to those particulars, including the itemized amounts, defined in the original FINAL PAYMENT CERTIFICATE, and that he will commence with any such arbitration claim or suit within 15 calendar days from and after the time of final PAYMENT of the work and that his failure to file a formal claim within this period constitutes his full acceptance of the Engineer's final estimate and payment. The overpayment refund check from the Contractor, if required, will be considered a part of any Acceptance Letter executed.
- 2 The Contractor has properly maintained the project, as specified hereinbefore.
- 3 The Contractor has furnished a sworn affidavit to the effect that the Contractor has paid all bills and no suits are pending (other than those exceptions listed, if any) in connection with work performed under the Contract and that the Contractor has not offered or made any gift or gratuity to, or made any financial transaction of any nature with, any employee of the OWNER in the performance of the Contract.
- 4 The surety on the contract bond consents, by completion of their portion of the affidavit and surety release subsequent to the Contractor's completion of his portion, to final payment to the Contractor and agrees that the making of such payment does not relieve the surety of any of its obligations under the bond.
- 5 The Contractor has furnished all required mill tests and analysis reports to the Engineer.

\* \* \* \* \*

# PART 4 SPECIFICATIONS

# SECTION 01001 GENERAL REQUIREMENTS

#### PART 1 - GENERAL

### 1.01 GENERAL

A. The Work under this Contract shall be performed by the Contractor as required by the City. Work will be authorized in the form of a Notice to Proceed issued to the Contractor. The Contractor shall complete all Work in the Contract within the number of calendar days stipulated in the Contract, unless an extension in the time of completion is granted by the City. Upon completion of the Work and compliance with applicable provisions in the Contract Documents, the Contractor will receive final payment for all Work done.

### B. Contractor's Duties:

- 1. In addition to provisions stipulated in other portions of the Contract Documents, Contractor shall secure permits as necessary for proper execution and completion of the work.
- 2. Contractor shall be totally responsible for all permits required and shall ensure that construction complies with all applicable local, state, and federal codes.
- 3. Provide an experienced, qualified, and competent Superintendent to oversee the Work. Superintendent shall be expected to remain for the duration of the Project.

## 1.02 MOBILIZATION AND DEMOBILIZATION

CONTRACTOR shall be responsible for mobilization and demobilization of labor, materials and equipment. Payment for mobilization and demobilization will be included in the lump sum price indicated in the BID.

## 1.03 CONTRACTOR'S USE OF PREMISES

- A. Work shall be scheduled as to not interfere with on-going area activities.
- B. Coordinate use of premises and requirements with the City.
- C. Assume full responsibility for the protection and safekeeping of products under this Contract, stored on the site.
- E. Contractor shall provide drinking water for construction personnel.
- F. Contractor shall provide waste solutions for construction personnel.

### 1.04 FINISHING OF SITE AND STORAGE AREAS

Upon completion of the Project, all areas used by the Contractor shall be properly cleared of all temporary structures, rubbish, and waste materials and properly graded to drain and blend in with the abutting property. Areas used for the deposit of waste materials shall be finished to properly drain and blend with the surrounding terrain.

# PART 2 - TEMPORARY CONSTRUCTION UTILITIES AND FACILITIES

#### 2.1 TEMPORARY WATER

CONTRACTOR shall make his own arrangements to obtain suitable water and shall pay all costs.

#### 2.2 SANITARY FACILITIES

CONTRACTOR shall provide and maintain sanitary facilities for his employees and his subcontractors' employees that will comply with the regulations of local and state departments of health, and as directed by the CH2M/OMI and the City.

### 2.3 STORAGE OF MATERIALS

A. Materials shall be so stored as to ensure the preservation of their quality and fitness for the work. When considered necessary, materials shall be placed on wooden platforms or other hard, clean surfaces, not on the ground.

# PART 3 - PRESERVATION, RESTORATION, AND CLEANUP

# 3.1 SITE RESTORATION AND CLEANUP

At all times during the work keep the premises clean and orderly. Upon completion of the day's work, repair all damage caused by equipment and leave the project clean and free of rubbish or excess materials of any kind.

Stockpile excavated materials in a manner that will cause the least damage to the area and near shore waters.

Upon completion of the Work, all areas used by the Contractor shall be cleared of temporary structures, rubbish, and waste materials, and properly graded to finished surface similar to the original surface, free-draining and free from holes, ruts, rough spots, leaving the area like original condition.

\* \* \* \* \*

# SECTION 01010 SCOPE OF WORK

The City of Key West is seeking BIDS from qualified, experienced contractors for the construction and permitting of a new Dockmaster building with transient restrooms and laundry facility and maintenance shop located at the City Marina at Garrison Bight. All improvements must meet requirements of the Florida Building Code, current edition, the Boating Infrastructure Grant (BIG), and must be constructed to meet Florida Green Building Coalition (FGBC) certification. The project will include, but not be limited to, the following:

# **Demolition**

Removal of existing \*restroom/ shower building, \*office trailer, ramp, deck, relocated light post, and removal of some asphalt, curbs and walkways to accommodate new work. Cap all plumbing lines, not to be re-used. All of the unused electrical wiring is to be pulled back to panel. (50% Dockmaster/ Maintenance, 50% Transient Restroom).

# **Foundation**

Construct the building foundation with reinforced concrete piers and auger cast piles. Reinforced concrete piles to be at depths as determined by the structural drawings. (68% Dockmaster/ Maintenance, 32% Transient Restroom).

# Lift, Stairs, Decking, and Railing

Lift- Handicap lift to be as specified on the drawings.

Stairs- Stairs to be constructed of P.T. wood with P.T. wood finish.

Decking- P.T. wood decking. The decking is to be installed per specification.

Railing- Aluminum guardrails and handrails. Some guardrail sections are to have prefabricated aluminum decorative x-bracing in front of cable rails. Refer to construction documents for further information.

Aluminum Building Signage w/ integrated lighting.

(50% Dockmaster/ Maintenance, 50% Transient Restroom).

### **Building Structure and Roof**

Construct the new Dockmaster Building using concrete and concrete block as described in the drawing set. The roof structure is to be constructed with P.T. wood roof rafters and preengineered wood roof trusses. The roof finish is to be a single-ply membrane roof over sloped insulation for the main building and a white galvanized standing seam metal roof for the entry canopy (alternate option- White alum. standing seam roof). The building structure and roof is to be constructed per the specifications and in a water tight manner. Refer to construction documents for further information. (68% Dockmaster/ Maintenance, 32% Transient Restroom).

### **Interior Finishes**

Interior finishes are to be provided and installed by the contractor in accordance with the construction documents. The interior finishes encompass approximately 2,137 SF. (68% Dockmaster/ Maintenance, 32% Transient Restroom).

# **Doors and Windows**

Provide and install all doors, windows, storefronts, and louvers per the construction documents. All fenestrations are to be installed to provide water and air tightness per the construction documents design criteria.

Doors- Refer to door schedule

Windows- Basis of design or equal CGI Commercial Series. Refer to window schedule Storefronts- Basis of design YKK: YHS 50 TU. Refer to storefront schedule.

Louvers- Louver manufacturer varies. Refer to louver schedule.

Hardware- All hardware is to be heavy-duty commercial grade. Refer to hardware schedule. (79% Dockmaster/ Maintenance, 21% Transient Restroom).

# Mechanical

Install new HVAC system as per the construction documents. Transient restroom/ shower facility, transient laundry facility, and public restrooms to be naturally ventilated with exhaust system. Refer to construction documents for system requirements and finishing. (72% Dockmaster/ Maintenance, 28% Transient Restroom).

# **Electrical**

Install all electrical and power needs as directed by the construction documents. The building is to be powered by a 400 amp 120/240V Single Phase main breaker with one 400 amp panel and one 100 amp panel. Provide and install Tru-Portal Access System with card readers, all lighting requirements for building signage, site/ exterior lighting fixtures, exterior outlets, interior light fixtures and outlets, and security system with cameras as denoted in the construction documents. The anticipated footage, for the secondary feeder to power the building, is 75'-0". (68% Dockmaster/ Maintenance, 32% Transient Restroom).

# **Plumbing**

New plumbing fixtures for the transient restrooms, shower facility, office restroom, laundry facility, and public restrooms. Install hot water heater system. Refer to construction documents for further information. (31% Dockmaster/ Maintenance, 69% Transient Restroom).

#### Site Work

Install new driveway, ADA curb ramps, new walkways, new bike racks, new parking spaces, and some reconfigured grading. Existing catch basins and water retention systems are to be utilized. Refer to construction documents. (50% Dockmaster/ Maintenance, 50% Transient Restroom).

### Landscaping

Install plants along the right-of-way, at parking lot perimeter, and in non-vehicular open spaces per the landscape drawings. The irrigation system is to be provided by the contractor through shop drawings and must comply with FGBC irrigation system requirements. (50% Dockmaster/Maintenance, 50% Transient Restroom).

\*NOTE: The existing Dockmaster office trailer is to remain fully functional and have uninterrupted operations throughout the project. Site work which occurs in the existing trailer location is be phased and scheduled in order to avoid any interruptions to the function and

service of the Dockmaster facility. Contractor is to provide and maintain one unisex ADA compliant portable toilet for the duration of the project.

The intent of these Documents is to describe a functionally complete project (or part thereof) to be constructed in accordance with the Contract Documents. Any work, materials, or equipment that may reasonably be inferred from the Contract Documents, as being required to produce the intended result shall be supplied whether or not specifically called for.

#### **SECTION 01025**

#### MEASUREMENT AND PAYMENT

### 1.1 GENERAL

- A. Contractor shall receive and accept compensation as provided in the Bid and Contract in full payment for performing all operations necessary to complete the work under the Lump Sum portions of this Contract, and also in full payment for all loss or damages arising from the nature of the work, until the final acceptance by the City.
- B. Lump Sum Price stated in the Bid shall include all costs and expenses for performing and completing the work as ordered. Measurement and payment for an item at a Unit Price or Lump Sum shown in the Bid shall be in accordance with the description of the item in this section.
- C. Contractor's attention is called to the fact that the quotations for various items of work are intended to establish a total price for completing the work in its entirety. It is the intent of this Contract that the Contractor provide a completed structure, and any item required to accomplish this shall be included to establish a total cost.

#### 1.2 PAYMENT

- A. General: Progress payments will be made monthly on the date established at the preconstruction meeting.
- B. Payment for all Work as ordered or specified in the Contract Documents is included in the Contract Price. No measurement or payment will be made for individual items except as itemized herein as unit price items or lump sum.

### 1.4 NONPAYMENT FOR REJECTED OR UNUSED PRODUCTS

Payment will not be made for following:

- 1. Loading, hauling, and disposing of rejected material.
- 2. Quantities of material wasted or disposed of in manner not called for under Contract Documents.
- 3. Rejected loads of material, including material rejected after it has been placed by reason of failure of Contractor to conform to provisions of Contract Documents.
- 4. Material not unloaded from transporting vehicle.
- 5. Defective Work not accepted by the City.
- 6. Material remaining on hand after completion of Work.

# 1.5 PARTIAL PAYMENT FOR STORED MATERIALS AND EQUIPMENT

- A. Partial Payment: No partial payments will be made for materials and equipment delivered or stored unless Shop Drawings are acceptable to the City and materials are properly stored at a site as agreed to by the City.
- B. Final Payment: Will be made only for products incorporated in Work; remaining products, for which partial payments have been made, shall revert to Contractor unless

otherwise agreed, and partial payments made for those items will be deducted from final payment.

\* \* \* \* \*

# SECTION 01300 SUBMITTALS

## PART 1 GENERAL

#### 1.01 GENERAL

- A. Inquiries: Direct to ENGINEER regarding procedure, purpose, or extent of Submittal.
- B. Submittal Submission Procedures: As provided in General Conditions, as specified herein, and as may otherwise be established during the preconstruction conference.
- C. OWNER's Authorization: At any time, OWNER may authorize changes to procedures and requirements for Submittals, as necessary to accomplish specific purpose of each Submittal. Such authorization will be by Field Order or Work Change Directive.
- D. Timeliness: Make submissions in such sequence as to cause no delay in Work or in work of other contractors.

### E. Identification of Submittals:

- 1. Identify each Submittal with numbering and tracking system reviewed by ENGINEER:
  - a. Sequentially number each Submittal.
  - b. Resubmission of a Submittal will have original number with sequential alphabetic suffix.
- 2. Show date of submission and dates of previous submissions.
- 3. Show Project title and OWNER's contract identification and contract number.
- 4. Show names of CONTRACTOR, Subcontractor or Supplier, and manufacturer as appropriate.
- 5. Clearly identify revisions from previous submissions.

# F. Incomplete Submittal Submissions:

- 1. At ENGINEER's sole discretion, ENGINEER will either (i) return the entire Submittal for CONTRACTOR's revision/correction and resubmission, or (ii) retain portions of the Submittal and request submission/resubmission of specified items or as noted thereon.
- 2. Submittals which do not clearly bear CONTRACTOR's specific written indication of CONTRACTOR review and approval of Submittal will be returned to CONTRACTOR unreviewed for resubmission in accordance with Contract Documents.
- 3. Delays, resequencing or other impact to Work resulting from CONTRACTOR's submission of unchecked or unreviewed, incomplete, inaccurate or erroneous, or nonconforming Submittals, which will require CONTRACTOR's resubmission of

- a Submittal for ENGINEER's review, shall not constitute a basis of claim for adjustment in Contract Price or Contract Times.
- F. Non-specified Submissions: Submissions not required under these Contract Documents and not shown on schedule of Submittals submissions will not be reviewed and will be returned to CONTRACTOR.
- G. Transmit Submittals in accordance with current accepted schedule of Submittal submissions, and deliver as follows:
  - 1. Submittals to: Key West Historic Seaport, Deputy Port and Marine Services Director.
- H. Disposition of Submittals: As specified herein for administrative Submittals. ENGINEER will review, stamp, and indicate requirements for resubmission or acceptance on Submittal as follows:
  - 1. Reviewed as Noted:
    - a. Reference the General Conditions for intent.
    - b. CONTRACTOR may proceed to perform Submittal related Work.
    - c. One copy for ENGINEER's file.
    - d. One copy returned to CONTRACTOR.
  - 2. Revise and Resubmit (Revise/Correct or Develop Replacement and Resubmit):
    - a. Revise/correct in accordance with ENGINEER's comments and resubmit.
    - b. One copy to ENGINEER's file.
    - c. One copy returned to CONTRACTOR appropriately annotated.
- I. ENGINEER's Review: ENGINEER will act upon CONTRACTOR's Submittal and transmit response to CONTRACTOR not later than 30 days after receipt, unless: (i) specified otherwise or (ii) accepted by ENGINEER and identified on current accepted schedule of Submittals submissions. Re-submittals will be subject to the same review time.

### 1.02 SHOP DRAWINGS

- A. Description: Reference the General Conditions.
- B. Excessive Shop Drawing Review: Review of the first submission and two resubmissions of Shop Drawings will be performed by ENGINEER at no cost to CONTRACTOR. Subsequent additional resubmissions of that Shop Drawing will be reviewed by ENGINEER, however, ENGINEER will document work hours and other expenses required to perform such additional review(s). OWNER shall deduct these costs from Contractor's contract for reimbursement to the ENGINEER.
- C. Copies: Submit two (2) copies.

D. Submit Shop Drawings to ENGINEER for equipment and materials to be furnished under these Contract Documents.

# E. Identify and Indicate:

- 1. Pertinent products, units and assemblies, and system or equipment identification or tag numbers.
- 2. Critical field dimensions and relationships to other critical features of Work.
  - a. Each deviation or variation from Contract Documents.
- F. Resubmissions: Clearly identify each correction or change made.
- G. Foreign Manufacturers: When proposed, include following additional information:
  - 1. Names and addresses of at least two companies closest to Project that maintain technical service representatives.
  - 2. Complete inventory of spare parts and accessories for each piece of equipment.

# H. Preparation:

- 1. Format: Whenever possible, schedule for and combine Shop Drawings into a single Submittal package.
- 2. Present in a clear and thorough manner and of sufficient detail to show kind, size, arrangement, and function of components, materials, and devices and compliance with Contract Documents.

#### PART 4 PAYMENT

A. Payment for the work in this section will be incidental

# SECTION 01700 CONTRACT CLOSEOUT

## PART 1 - GENERAL

# 1.01 REQUIREMENTS

Contract completion includes substantial completion, final inspection after completion, final cleaning, contractor's closeout submittals, and final adjustment of accounts

### 1.02 SUBSTANTIAL COMPLETION

- A. When Contractor considers work has reached substantial completion, he shall submit to the OWNER the following:
  - a. Written notice that the work is substantially complete in accordance with Contract Documents.
  - b. A list of items yet to be completed or corrected and explanations thereof.
- B. Within a reasonable time upon receipt of such notice, the OWNER will make an inspection, if necessary, to determine the status of completion.
- C. Should the OWNER determine that the work is not substantially complete:
  - a. The OWNER will promptly notify the Contractor in writing, giving the reasons thereof.
  - b. Contractor shall remedy the deficiencies in the work and send a second written notice of Substantial Completion to the OWNER.
  - c. Upon receipt of the second notice, the OWNER will review the work.
- D. When the OWNER finds that the Work is substantially complete he will issue a Certificate of Substantial Completion with a tentative list of items to be completed or corrected before final inspection.

### 1.03 FINAL INSPECTION AFTER COMPLETION

- A. When Contractor considers the Work is complete with all minor deficiencies completed or corrected, he shall submit written certification that:
  - a. Contract Document requirements have been met.

- b. Work has been inspected for compliance with Contract Documents.
- c. Work has been completed in accordance with Contract Documents.
- d. All minor deficiencies have been corrected or completed and the Work is ready for final inspection.
- e. Project record documents are complete and submitted.
- B. Within a reasonable time upon receipt of such certification, the OWNER will make an inspection to verify the status of completion.
- C. Should the OWNER determine that the work is incomplete or defective:
  - a. The OWNER will promptly notify the Contractor in writing, listing the incomplete or defective work.
  - b. Contractor shall remedy the deficiencies in the work and send a second written certification to the OWNER that the work is complete.
  - c. Upon receipt of the second certification, the OWNER will review the Work.
- D. When the OWNER determines that the work is acceptable, under the Contract Documents, he shall request the Contractor to make closeout submittals.

# 1.04 FINAL CLEANING

- A. Execute final cleaning prior to final inspection.
- B. Clean site; vacuum floors and walls, wash windows, mirrors and plumbing fixtures.
- C. Remove waste and surplus materials, rubbish, and construction facilities from the Project and from the site.

# 1.05 CONTRACTOR'S CLOSEOUT SUBMITTALS

- A. Project Record Documents
  - a. At Contract Closeout, submit documents with transmittal letter containing date, project title, Contractor's name and address, list of documents, and signature of Contractor.
  - b. Changes made by Field Order or by Change Order.
- B. Evidence of payment and Release of Liens.

## 1.06 FINAL ADJUSTMENT OF ACCOUNTS

A. Submit a final statement of accounting to the OWNER.

- B. Statement shall reflect all adjustments to the Contract Sum.
  - a. The original Contract sum.
  - b. Additions and deductions resulting from:
    - Previous change orders or written amendment.
    - Allowances
    - Lump Sum prices
    - Deductions for uncorrected work.
    - Deductions for liquidated damages
    - Other adjustments
  - c. Total contract sum as adjusted
  - d. Previous payments
  - e. Sum remaining due

\*\*\*\*\*\*

# PART 5

# **SUPPLEMENTAL INFORMATION** (BIG Grant Requirements)

# **CERTIFICATIONS AND ASSURANCES**

In performance of this Contract, Contractor provides the following certifications and assurances:

- A. <u>Debarment and Suspension Certification (2 CFR Part 1400)</u>
- B. Certification Regarding Lobbying (31 U.S.C. 1352)
- C. <u>Certification Regarding Drug-Free Workplace Requirements (41 U.S.C. 701 et. seq.) (as applicable)</u>
- D. <u>Certification Regarding the Scrutinized Companies List (section 287.135, F.S.) (as applicable)</u>
- E. Buy American Assurance (43 CFR Part 12, Subpart E)
- F. Trafficking Victims Protection Act Assurance (2 CFR Part 175)
- G. <u>Boating Infrastructure Grant Program (BIGP) Rule Assurance (50 CFR</u> Part 86)

# A. CERTIFICATION REGARDING DEBARMENT, SUSPENSION, AND OTHER RESPONSIBILITY MATTERS -PRIMARY COVERED TRANSACTION.

The undersigned Contractor certifies to the best of its knowledge and belief, that it and its principals:

- 1. Are not presently debarred, suspended, proposed for debarment, declared ineligible, or voluntarily excluded from covered transactions by a Federal department or agency;
- 2. Have not within a three-year period preceding this Contract been convicted of or had a civil judgment rendered against them for commission of fraud or a criminal offense in connection with obtaining, attempting to obtain, or performing a public (Federal, State, or local) transaction or contract under a public transaction; violation of Federal or State antitrust statutes or commission of embezzlement, theft, forgery, bribery, falsification or destruction of records, making false statements, or receiving stolen property;
- 3. Are not presently indicted or otherwise criminally or civilly charged by a government entity (Federal, State or local) with commission of any of the offenses enumerated in paragraph A.2. of this certification; and/or
- 4. Have not within a three-year period preceding this application/proposal had one or more public transactions (Federal, State, or local) terminated for cause of default.

If Contractor is unable to certify to any of the statements in this certification, Contractor shall attach an explanation to this Contract.

**B. CERTIFICATION REGARDING LOBBYING -** Certification for Contracts, Grants, Loans and Cooperative Agreements.

The undersigned Contractor certifies, to the best of his or her knowledge and belief, that:

No Federal appropriated funds have been paid or will be paid, by or on behalf of the undersigned, to any person for influencing or attempting to influence an officer or employee of Congress, or an employee of a Member of Congress in connection with the awarding of any Federal contract, the making of any Federal grant, the making of any Federal loan, the entering into of any cooperative agreement, and the extension, continuation, renewal, amendment or modification of any Federal contract, grant, loan or cooperative agreement.

If any funds other than Federal appropriated funds have been paid or will be paid to any person for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employees of Congress, or employee of a Member of Congress in connection with a Federal contract, grant, loan, or cooperative agreement, the undersigned shall also complete and submit Standard Form - LLL, "Disclosure Form to Report Lobbying," in accordance with its instructions.

The undersigned shall require that language of this certification be included in the documents for all subcontracts at all tiers (including subcontracts, sub-grants and contracts under grants, loans and cooperative agreements) and that all sub-recipients and contractors shall certify and disclose accordingly.

This certification is a material representation of fact upon which reliance was placed when this Contract was made or entered into. Submission of this certification is a prerequisite for making or entering into this Contract imposed by 31 U.S.C. 1352. Any person who fails to file the required certification shall be subject to a civil penalty of not less than \$10,000 and not more than \$100,000 for each such failure.

# C. CERTIFICATION REGARDING DRUG-FREE WORKPLACE REQUIREMENTS

Pursuant to the Drug-Free Workplace Act of 1988, the undersigned attests and certifies that the Contractor (if not an individual) will provide a drug-free workplace by the following actions:

- 1. Publishing a statement notifying employees that the unlawful manufacture, distribution, dispensing, possession or use of a controlled substance is prohibited in Contractor's workplace and specifying the actions that will be taken against employees for violation of such prohibition.
- 2. Establishing an ongoing drug-free awareness program to inform employees concerning:
  - a. The dangers of drug abuse in the workplace.
  - b. The policy of maintaining a drug-free workplace.

- c. Any available drug counseling, rehabilitation and employee assistance programs.
- d. The penalties that may be imposed upon employees for drug abuse violations occurring in the workplace.
- 3. Making it a requirement that each employee to be engaged in the performance of the Contract be given a copy of the statement required by paragraph C.1. of this certification.
- 4. Notifying the employee in the statement required by paragraph C.1. of this certification that, as a condition of employment under the Contract, the employee will:
  - a. Abide by the terms of the statement.
  - b. Notify the employer in writing of his or her conviction for a violation of a criminal drug statute occurring in the workplace no later than five (5) calendar days after such conviction.
- 5. Notifying the City in writing ten (10) calendar days after receiving notice under subparagraph 4.b. from an employee or otherwise receiving actual notice of such conviction. Provide such notice of convicted employees, including position title, to every Contract Manager on whose contract activity the convicted employee was working. The notice shall include the identification number(s) of each affected Contract or Grant.
- 6. Taking one of the following actions, within thirty (30) calendar days of receiving notice under subparagraph 4.b. herein, with respect to any employee who is so convicted:
  - a. Taking appropriate personnel action against such an employee, up to and including termination, consistent with the requirements of the Rehabilitation Act of 1973 as amended.
  - b. Requiring such employee to participate satisfactorily in drug abuse assistance or rehabilitation program approved for such purposes by a Federal, State or local, health, law enforcement, or other appropriate agency.
- 7. Making a good faith effort to continue to maintain a drug-free workplace through implementation of this entire certification.

If the Contractor is an individual, the Contractor certifies that:

- 1. As a condition of the Contract, Contractor will not engage in the unlawful manufacture, distribution, dispensing, possession, or use of a controlled substance in conducting any activity with the contract; and,
- 2. If convicted of a criminal drug offense resulting from a violation occurring during the conduct of any contract activity, Contractor will report the conviction, in writing, within

10 calendar days of the conviction, to the City when notice is made to such a central point, it shall include the identification number(s) of each affected contract.

# D. CERTIFICATION REGARDING the SCRUTINIZED COMPANIES LISTS, SECTION 287.135, F.S.

If this Contract is in the amount of \$1 million or more, in accordance with the requirements of section 287.135, F.S., Contractor hereby certifies that it is not listed on either the Scrutinized Companies with Activities in Sudan List or the Scrutinized Companies with Activities in the Iran Petroleum Energy Sector List. Both lists are created pursuant to section 215.473, F.S. Contractor also hereby certifies that it is not engaged in business operations is Cuba or Syria.

Contractor understands that pursuant to section 287.135, F.S., the submission of a false certification may subject Contractor to civil penalties, attorney's fees, and/or costs.

# E. BUY AMERICAN ASSURANCE (43 CFR PART 12, SUBPART E)

The Contractor agrees to construct the Project according to all provisions of the Buy American Act – Construction Materials which can be found in 43 CRF Part 12, Subpart E, Sections 12.800 through 12.830. It applies to procurement contracts awarded under a grant or cooperative agreement for construction, alteration, or repair of any public building or public work in the United States. As prescribed in 43 CFR Part 12, Subpart E, Section 12.825:

(a) The Buy American Act (41 U.S.C. 10) provides that the Government give preference to domestic construction material.

*Components*, used in this clause, means those articles, materials, and supplies incorporated directly into construction materials.

Construction material, as used in this clause, means an article, material, or supply brought to the construction site for incorporation into the building or work. Construction material also includes an item brought to the site pre-assembled from articles, materials or supplies. However, emergency life safety systems, such as emergency lighting, fire alarm, and audio evacuation systems, which are discrete systems incorporated into a public building or work and which are produced as a complete system, shall be evaluated as a single and distinct construction material regardless of when or how the individual parts or components of such systems are delivered to the construction site.

Domestic construction material, as used in this clause, means (a) an unmanufactured construction material mined or produced in the United States, or (b) a construction material manufactured in the United States, if the cost of its components mined, produced, or manufactured in the United States exceeds 50 percent of the cost of all its components. Components of foreign origin of the same class or kind as the construction materials determined to be unavailable pursuant to §12.810(a)(3) of 43 CFR part 12, subpart E shall be treated as domestic.

(b) The contractor agrees that only domestic construction material will be used by the contractor, subcontractors, materialmen, and suppliers in the performance of this agreement, except for foreign construction materials, if any, listed in this agreement.

# F. TRAFFICKING VICTIMS PROTECTION ACT ASSURANCE

Any grant, contract, or cooperative agreement provided or entered into by a Federal department or agency under which funds are to be provided to a private entity, in whole or in part, shall include a condition that authorizes the department or agency to terminate the grant, contract, or cooperative agreement, without penalty, if the grantee or any subgrantee, or the contractor or any subcontractor (i) engages in severe forms of trafficking in persons or has procured a commercial sex act during the period of time that the grant, contract, or cooperative agreement is in effect, or (ii) uses forced labor in the performance of the grant, contract, or cooperative agreement.

# G. BOATING INFRASTRUCTURE GRANT PROGRAM (BIGP) RULE ASSURANCE

The Contractor agrees to construct the Project according to all provisions of the Boating Infrastructure Grant Program Final Rule, 50 CFR Part 86, attached and made part of this Contract. The Contractor further agrees to comply with all other applicable federal, state, and local rules and regulations in providing services under this Contract. The Contractor further agrees to include this provision in all subcontracts issued as a result of this Contract.

If Contractor is unable to certify to any of the statements in this certification, Contractor shall attach an explanation to this Contract.

By signing below, Contractor certifies the representations outlined in parts A through G above are true and correct.

(Signature and Title of Authorized Representative)		
Contractor Name		
Contractor Street Address		
City, State, ZIP Code		

In addition to the Certifications and Assurances to be signed by the Contractor, the following are applicable to this Contract:

### PUBLIC ENTITY CRIMES.

A. Convicted Vendor List. Pursuant to Subsection 287.133(2)(a), F.S., a person or affiliate who has been placed on the convicted vendor list following a conviction for a public entity crime may not be awarded or perform work as a Grantee, supplier, subcontractor or consultant under a contract with any public entity and may not transact business with any public entity in excess of the threshold amount provided in Section 287.017, F.S., for Category Two for a period of thirty six (36) months from the date of being placed on the convicted vendor list.

The State of Florida, Department of Management Services, Division of State Purchasing provides listings for convicted, suspended, discriminatory and federal excluded parties, as well as the vendor complaint list at:

http://www.dms.myflorida.com/business\_operations/state\_purchasing/vendor\_informatio n/convicted suspended discriminatory complaints vendor lists/convicted vendor list

- B. **Notice of Conviction of Public Entity Crime.** Any person must notify the Department of Management Services and the City of Key West within 30 days after conviction of a public entity crime applicable to that person or an affiliate of that person as defined in Section 287. 133, F.S.
- C. Certifications and Assurances. Upon execution of this Contract by the Contractor, the Contractor shall complete, sign and return to the City's Grant Manager a completed copy of the form entitled "Certifications and Assurances," attached hereto. This includes the Certification Regarding Public Entity Crimes.

#### PUBLIC RECORDS.

Pursuant to Section 119.0701, F.S., A) Keep and maintain public records that ordinarily and necessarily would be required by the City in order to perform the service. B) Provide the public with access to public records on the same terms and conditions that the City would provide the records and at a cost that does not exceed the cost provided in Chapter 119, F.S. or as otherwise provided by law. C) Ensure that public records that are exempt or confidential and exempt from public records disclosure requirements are not disclosed except as authorized by law. D) Meet all requirements for retaining public records and transfer, at no cost, to the City all public records in possession of the Contractor upon termination of the Contract and destroy any duplicate public records that are exempt or confidential and exempt from public records disclosure requirements. All records stored electronically must be provided to the public agency in a format that is compatible with the information technology systems of the City.

# RECORD KEEPING REQUIREMENTS.

**A.** Contractor Responsibilities. The Contractor shall maintain accurate books, records, documents and other evidence that sufficiently and properly reflect all direct and indirect costs of any nature expended in the performance of this Contract, in accordance with generally accepted accounting principles.

- **B. State Access to Contractor Books, Documents, Papers, and Records.** The Contractor shall allow the Florida Fish and Wildlife Conservation Commission, the Chief Financial Officer of the State of Florida, the Auditor General of the State of Florida, the Florida Office of Program Policy Analysis and Government Accountability or authorized representatives of the state or federal government to have access to any of the Contractor's books, documents, papers, and records, including electronic storage media, as they may relate to this Contract, for the purposes of conducting audits or examinations or making excerpts or transcriptions.
- **C.** Contractor Records Retention. Project records shall be maintained for five (5) years following the close of this Contract. The Contractor shall cooperate with the City to facilitate the duplication and transfer of such records upon the City's request.

# **D.** Contractor Responsibility to Include Records Requirements – Subcontractors. In the event any

work is subcontracted under this Contract, the Contractor shall include the aforementioned audit and record keeping requirements in all subsequent contracts.

FEDERAL FUNDS. This Agreement relies on federal funds, therefore, the following terms and conditions apply:

Compliance with all federal laws, rules, and regulations, including but not limited to:

- Executive Order 11246 of September 24, 1965, entitled "Equal Employment Opportunity." as amended by Executive Order 11375 of October 13, 1967, and as supplemented in Department of Labor regulations (41 CFR Chapter 60).
- The Copeland "Anti-Kickback" Act (18 U.S.C. 874) as supplemented in Department of Labor regulations (29 CFR part 3).
- Sections 103 and 107 of the Agreement Work Hours and Safety Standards Act (40 U.S.C. 327-330) as supplemented by Department of Labor regulations (29 CFR part 5). (Construction agreements awarded by grantees and subgrantees in excess of \$2,000, and in excess of \$2,500 for other agreements which involve the employment of mechanics or laborers)
- All applicable standards, orders, or requirements issued under section 306 of the Clean Air Act (42 U.S.C. 1857(h)), section 508 of the Clean Water Act (33 U.S.C. 1368), Executive Order 11738, and Environmental Protection Agency regulations (40 CFR part 15).
- Mandatory standards and policies relating to energy efficiency which are contained in the State energy conservation plan issued in compliance with the Energy Policy and Conservation Act (Pub. L. 94-163, 89 Stat. 871).

# SUBCONTRACTS.

Each subcontractor will be deemed to be an independent contractor and will not be considered or permitted to be an agent, servant, joint venturer, or partner of the State of Florida.

#### NONDISCRIMINATION

- **A. Non-Discrimination in Performance.** No person, on the grounds of race, creed, color, national origin, age, sex, or disability, shall be excluded from participation in, be denied the proceeds or benefits of, or be otherwise subjected to discrimination in performance of this Agreement.
- **B. Discriminatory Vendor List.** In accordance with Section 287.134, F.S., an entity or affiliate who has been placed on the discriminatory vendor list may not submit a bid, proposal, or reply on a contract to provide any goods or services to a public entity; may not submit a bid, proposal or reply on a contract with a public entity for the construction or repair of a public building or public work; may not submit bids, proposals, or replies on leases of real property to a public entity; may not be awarded or perform work as a contractor, supplier, subcontractor, or consultant under a contract with any public entity; and may not transact business with any public entity. The Contractor has a continuing duty to disclose to the City whether they appear on the discriminatory vendor list.

## PROHIBITION OF UNAUTHORIZED ALIENS.

In accordance with federal Executive Order 96-236, the employment of unauthorized aliens is considered a violation of Section 274A(e) of the Immigration and Nationalization Act. Such violation shall be cause for unilateral cancellation of the Contract if the Contractor knowingly employs unauthorized aliens.

# EMPLOYMENT ELIGIBILITY VERIFICATION (E-VERIFY).

- **A. Requirement to Use E-Verify.** Executive Order 11-116, signed May 27, 2011, by the Governor
- of Florida, requires contracts in excess of nominal value to expressly require the: 1.) utilization of the U.S. Department of Homeland Security's E-Verify system to verify the employment eligibility of all new employees hired during the contract term; and,
- 2.) inclusion in all subcontracts under this Contract, of the requirement that subcontractors performing work or providing services pursuant to this contract utilize the E-Verify system to verify the employment eligibility of all new employees hired by the subcontractor during the term of the subcontract.
- **B. E-Verify Online.** E-Verify is an Internet-based system that allows an employer, using information reported on an employee's Form I-9, Employment Eligibility Verification, to determine the eligibility of all new employees hired to work in the United States after the effective date of the required Memorandum of Understanding (MOU); the responsibilities and elections of federal contractors, however, may vary, as stated in Article II.D.1.c. of the MOU. There is no charge to employers to use E-Verify. The Department of Homeland Security's EVerify system can be found online at

http://www.dhs.gov/files/programs/gc 1185221678150.shtm

**C. Enrollment in E-Verify.** If you do not have an E-Verify MOU in effect, you must enroll in the E-Verify system prior to hiring any new employee after the effective date of this Contract.

- **D. E-Verify Recordkeeping.** The Contractor further agrees to maintain records of its participation and compliance with the provisions of the E-Verify program, including participation by its subcontractors as provided above, and to make such records available to the City or other authorized state entity consistent with the terms of the Contractor's enrollment in the program. This includes maintaining a copy of proof of the Contractor's and subcontractors' enrollment in the E-Verify Program (which can be accessed from the "Edit Company Profile" link on the left navigation menu of the E-Verify employer's homepage).
- **E. Employment Eligibility Verification.** Compliance with the terms of the Employment Eligibility Verification provision is made an express condition of this Contract and the City may treat a failure to comply as a material breach of the Contract.



Thursday, January 18, 2001

# Part VII

# Department of the Interior

Fish and Wildlife Service

50 CFR Part 86 Boating Infrastructure Grant Program; Final Rule



# DEPARTMENT OF THE INTERIOR

#### Fish and Wildlife Service

# 50 CFR Part 86

#### RIN 1018-AF38

#### Boating Infrastructure Grant Program

AGENCY: Fish and Wildlife Service, Interior.

ACTION: Final rule.

SUMMARY: This regulation provides for the uniform administration of the national Boating Infrastructure Grant Program and survey authorized by Section 7404 of the Sportfishing and Boating Safety Act of 1998. Through this program, the U.S. Fish and Wildlife Service will provide funds to States to install or upgrade tie-up facilities for transient recreational boats 26 feet or more in length.

DATES: This rule is effective on February 20, 2001.

ADDRESSES: The administrative record for this rule, including copies of comments received, is available for viewing Monday through Friday, 8 am to 4 pm, in the Division of Federal Aid, 4401 North Fairfax Drive, Room 140, Arlington, VA 22203.

## FOR FURTHER INFORMATION CONTACT:

Steve Farrell, Project Officer, U.S. Fish and Wildlife Service, Division of Federal Aid, 4401 North Fairfax Drive, Suite 140, Arlington, Virginia 22203; telephone (703) 358–2156; fax (703) 358–1705; email <steve\_farrell@fws.gov>.

# SUPPLEMENTARY INFORMATION:

#### Background

#### Economic Status of Boating in the United States

Historically, coastal and inland waterways were the first highways along our shores and into the interior of the continent. Americans used boats almost exclusively for transportation of people and goods. Today we use more than 12 million recreational boats to cruise and fish. Recreational boating is now a significant economic activity in many areas of the country and in many respects exceeds that of waterborne commerce. Given the present demographic forces, we expect this rule to have a positive economic impact by adding facilities to accommodate larger cruising boats.

#### Purpose of the Boating Infrastructure Grant Program

Recreational boats 26 feet or more in length, called "nontrailerable" boats, represent about 4 percent, or more than

600,000, of the recreational boats in the United States. Although we have approximately 12,000 marinas in the United States, Congress recognized that insufficient tie-up facilities exist for transient, nontrailerable boats for reasonable and convenient access from our navigable waters. These boaters are unable to enjoy many recreational, cultural, historic, scenic, and natural resources of the United States. We also have an insufficient quantity of marinas or commercial tie-up facilities along extended stretches of our coastlines and rivers that benefit transient, nontrailerable boats. In many parts of the country, the number of places to tieup, moor, or anchor a cruising boat, especially during a storm, is limited. Basic features, such as tie-ups, fuel, utilities, and restrooms, are often nonexistent. As a result, Congress passed the Sport Fishing and Boating Safety Act of 1998 (16 U.S.C. 777g). Under the Act, the U.S. Fish and Wildlife Service conducts the Boating Infrastructure Grant (BIG) Program. The BIG Program will provide \$32 million to States and Territories over 4 years to construct, renovate, or maintain tie-up facilities for recreational boats 26 feet or more in length.

This program will:

- (a) Create dockage for transient recreational boats 26 feet or more in length for recreational opportunities and safe harbors;
- (b) Provide navigational aids for boaters to use these facilities:
- (c) Enhance access to recreational, historic, cultural, natural, and scenic resources:
- (d) Strengthen local ties to the boating community and its economic benefits;
- (e) Promote public/private partnerships and entrepreneurial opportunities;
- (f) Provide continuity of public access to the shore; and
- (g) Promote awareness of transient boating opportunities.

The Act also directs us to:

- (a) Develop a national framework or methodology to conduct a boat access needs assessment or survey to determine the adequacy of facilities for recreational boats of all sizes;
- (b) Encourage States to complete the boat access needs survey; and
- (c) Complete a comprehensive national assessment of boat access needs and facilities (the assessment will be a compilation of information from the States' surveys into a national report of boat access needs).

#### Analysis of Public Comments and Changes Made to the Proposed Rule

On January 20, 2000, the U.S. Fish and Wildlife Service (Service) published a proposed rule in the Federal Register (65 FR 3331) and requested comments on the proposed rule and information collection for the national BIG Program. The Service received 13 written responses by the close of the comment period on March 20, 2000. The responses came from nongovernmental organizations, a private individual, and Federal and State employees and agencies.

We received a total of 170 comments from the 13 written responses covering the following areas: survey (74), criteria for selection (11), State plans (8), clarifications (3), suspected typographical errors (3), and comments of a general nature (71). We may not reflect these actual numbers in the list below due to combining similar comments or questions. We have addressed all of the comments in this section of the preamble and made any necessary changes to the proposed rule. As a result, some sections of the proposed rule were combined or eliminated. Comments pertaining to the proposed information collection are addressed under Substantive Comments, Issues 1 through 9.

The following is clarification or analysis and reporting of any substantial changes to the rule that the Service made in response to these comments. The Service also corrected other minor errors in the proposed rule, as described below.

#### Minor Changes

In § 86.12(n), in response to a comment asking us to define "transient," we are adding the following definition, "Passing through or by a place, staying 10 days or less."

In § 86.13(d), we changed "seasonal" to "transient" to avoid confusing wording.

In § 86.13(f), in response to a suggestion, we added "docks" after the word "floating" to clarify.

In § 86.13, in response to suggestions, we added at paragraphs (o), (p), and (q), "dockside" to clarify where utilities are placed.

In § 86.13, in response to a suggestion, we added "(r) Debris deflection booms." In § 86.13, in response to suggestions, we added "(s) Marine fueling stations."

In § 86.20(a)(1), in response to a suggestion, we added a second sentence as follows: "You must design new construction and renovations to last at least 20 years."

In § 86.20(e)(1), in response to a suggestion, we deleted from subparagraph (vii) to end, as the activities listed in the proposed rule are not appropriate preconstruction activities.

In § 86.20, in response to a suggestion, we added a new item, "(f) Produce information and education materials such as charts, cruising guides, and brochures."

In § 86.21(d), in response to a suggestion, we changed the sentence to read: "Construct or renovate principal structures not expected to last at least 20 years."

In § 86.44(b), in response to a suggestion, we replaced "\* \* we will" with "\* \* The State must

In § 86.54(f)(1)(i) in the proposed rule (now § 86.53(d)(1)(i)), in response to several comments, we changed the section to clarify that no State Tier One proposal may exceed \$100,000 in any given fiscal year.

#### Substantive Changes

Issue 1. Questions and comments on the survey included issues of respondent burden (how long to complete a section), unclear questions, issues of confidentiality, site-specific questions, and the value of answers to specific questions in determining national need.

Response: In response to all of the comments received on the technical aspects of the proposed survey, we completed significant revisions to the survey instrument. We considered all comments and redesigned the survey improved guidance and questions, and decreased the time burden for completion. The revised survey began its approval process when it was published in the Federal Register (65 FR 63606) on October 24, 2000 for a 60day public comment period. The survey will be ready for use during the FY 2001 grant cycle, and the Service will notify eligible applicants upon receiving OMB approval.

Issue 2. How can States pay for the

Response: States can pay the costs for conducting the survey using Federal Aid in Sport Fish Restoration program funds, and credit expenditures toward the 15% minimum used for motor boat access (16 U.S.C. 777g (g)(4)).

Issue 3. Is the survey required?
Response: The Service does not require States to conduct surveys to receive funding under the national BIG program. However, the Service must produce a comprehensive national assessment of recreational boat access needs and facilities. The States, by conducting and forwarding their survey results to the Service, are ensuring that

their needs are known (16 U.S.C. 777g (g)(2)). States may use existing, recent survey results, approved by the appropriate Service Regional Office, to determine boating infrastructure needs (§ 86.113).

Issue 4. States completing a survey should receive ranking points.

should receive ranking points.

Response: No "points" are assigned in the criteria for completing a survey; the only available points are outlined in § 86.60. States must use the survey results to develop a State plan for boating infrastructure. The plan (any plan certified by the Service Regional Office) is then eligible for 15 ranking points.

Issue 5, Can States add questions to the survey?

Response: States cannot add any questions to the survey. If OMB approves the final survey as written, we will reimburse States only for administering it as approved.

Issue 6. Typographical errors occur in the survey tables.

Response: We fixed typographical errors in both tables describing the type of participant in survey parts.

Issue 7. States should submit survey results electronically.

Response: The Service wants the results transmitted to the Service Regional Offices in a common electronic format, such as Microsoft Word, Word Perfect, Excel or Quattro Pro. We added a requirement regarding electronic transmission of results to § 86.111.

Issue 8. Clarify "boat service providers" as the term relates to public

Response: We changed § 86.102(b)(2)(i) and (ii) to include a phrase that specifies boat service providers "who allow public access."

Issue 9. What are the survey response

rate requirements?

Response: We changed § 86.115 to read "plus or minus 10 percent." We also replaced the language that suggests a 70 percent response rate with language that says the State is responsible for selecting a statistically valid sample

Issue 10. In § 86.60, the points do not add up to 100, or no points are allowed for completing surveys.

Response: This section needed several minor editorial changes to make it clear and concise. Among these, we changed the point total to 105. The issue regarding awarding points for completing surveys is addressed in Issue 4.

Issue 11. According to § 86.54, why are proposals that are awarded less than 60 points when held to the criteria in § 86.60 automatically moved to Tier Two? Hesponse: We removed this "60 point" break when we revised the Tier One/Two break.

Issue 12. By awarding 15 points for a State plan, the Service is making State plans mandatory, penalizing all nonplan proposals.

Response: State plans are desirable under the Act (16 U.S.C. 777g (g)). They help set priorities and describe how a State will fulfill recreational boaters' needs. We can also use plans to help set national priorities in the comprehensive national assessment of recreational boat access needs and facilities described in the Act. The plan nonetheless is optional. The Act states, "A State may develop a plan for," and in the proposed rule, § 86.131 states "Plans are voluntary." However, the Service will award 15 criteria points to any acceptable plan, already existing and current, or a newly developed plan based on the OMB-approved survey or other recent approved survey results. The reason for the 15 criteria points is that projects coming from an accepted plan will establish priorities based on a formal needs assessment.

Issue 13. Add a statement to identify priorities in the State plan in

§ 86.60(b)(1).

Response: We added to § 86.60(b)(1),
after the word "following" "priorities
identified in" your State's program plan.
We also agreed with the second thought
presented in this comment and changed
"construct and renovate" to "construct,
renovate, and maintain."

Issue 14. Change the point values assigned to certain criteria.

Response: We are not making a change to the final rule as a result of this comment. We believe that we did a thorough job of stakeholder involvement when we assigned these point values.

Issue 15. A possible conflict exists in § 86.60(b)(4) about match funds.

Response: We have changed § 86.60(b)(4) to read "Include private, local, or other State funds in addition to the non-Federal match described in § 86.42."

Issue 16. Because of typographical errors in § 86.60(b)(4)(i), the percentage range was left out.

Response: In § 86.60(b)(4), we changed paragraph (i) to read "Twenty-six percent to thirty-five percent—5 points," paragraph (ii) to remove the word "above" at the end, and paragraph (iii) to insert the word "and" so it reads "Fifty percent and above—15 points."

Issue 17. Add the word "proposed" before "tie-up" to the second sentence in § 86.60(b)(5), and delete the word "access" from the same sentence,

Response: We do not believe this suggestion adds to the clarity of the sentence. We are making no changes to the proposed rule as a result of this comment.

Issue 18. In § 86.60(b)(6), change the wording of the sentence to restrict the sites to those near population centers and raise the value of the points assigned.

Hesponse: We believe this change would unnecessarily restrict State proposals. We believe the sentence is clear and the points are appropriate. The sentence remains unchanged.

Issue 19. In § 86.60(b)(7), substitute "enhanced opportunities" for "access." Response: We are not changing this paragraph as a result of this comment because we believe the sentence is clear as written.

Issue 20. In § 86.60(b)(8), add
"adjacent" before the word
"community," and add "as the result of
construction, renovation, or increased
use," at the end of the first sentence.

Response: We believe this change would unnecessarily restrict the program criteria. The proposed sentence stands unchanged

stands unchanged.

Issue 21. Allow Tier One dollars to finance State plans because not to do so creates an unfunded Federal mandate.

Response: The Act states that

"\* \* a State may develop and submit
to the Secretary a plan for \* \* ," not
"must" (16 U.S.C. 777g-1 {c}). A
mandate does not exist. However, States
may do planning under other Sport Fish
Restoration Act grant programs for
purposes eligible under those programs
and apply the resulting plan to the BIG
Programs

Program.

Issue 22, Will the Service accept

existing plans?

Response: The Service will accept any plan certified by the appropriate Service Regional Office that ensures that public boat access is and will be adequate to meet the needs of recreational boaters on your State's waters (§ 86.134).

Issue 23. What are the products of the

National Assessment?

Response: The products of the Comprehensive National Assessment are listed in § 66.124.

are listed in § 86.124.

Issue 24. Complying with the National Environmental Policy Act and the Endangered Species Act could cause up to 2 years of delay.

Response: All grantees must agree to and certify compliance with all applicable Federal laws. We use a certified "Assurances" statement. We are making no changes to the proposed rule as a result of this comment.

Issue 25. Can States use consultant fees, design, permitting, and construction administration costs as match for Tier One projects?

match for Tier One projects? Response: Allowable matching and cost-sharing regulations are in 43 CFR 12.64; specific questions should be directed to the appropriate Service Regional Office.

Issue 26. Reduce match to where local governments or small marinas can compete for funds.

Response: In accordance with the Act, only States are eligible for funding under this grant program.

Issue 27. Are pre-award costs allowable?

Response: Only as described under § 86.20.

Issue 28. Who assumes the administrative burden after the program expires in 3 years?

Response: We discuss maintenance of approved projects in 50 CFR 80.17. Issue 29. How much money can we charge the public to tie-up?

Response: The going rate in the locality determines the amount for the facility. We are making no change to the proposed rule as a result of this comment.

Issue 30. For projects in Tier Two that exceed available funding in Tier Two, what is the methodology for awarding funds other than what States request?

Response: There is none. We cannot exceed established spending levels. If a State requires funds from two or more different programs to obtain full funding, the State is responsible for securing the funds.

Issue 31. In § 86.20(a)(3), substitute "deep enough" for "6 feet of depth at the lowest tide \* \* \* ."

Response: We researched these depths before making this specification, and we are not making the recommended change.

Issue 32. In § 86.20(a)(5)(iii), delete the reference to "this program," so that the sentence reads: "You may use funds from the Clean Vessel Act Program."

Response: We encourage the construction of pumpout stations with either set of funds. We are making no changes to the proposed rule as a result of this comment.

Issue 33. In § 86.20(e)(1), add "contract documents."

Response: No contract should be prepared and awarded until after we sign a grant; therefore, contract documents are not allowed as a preagreement cost. We are making no changes to the proposed rule as a result of this comment.

Issue 34. In § 86.21(e) and (g), add note on one-time dredging to (e) and add dinghy docks to (g).

Response: We address one-time dredging in § 86.20, and dinghy docks are not for transient vessels more than 26 feet in length. We are making no changes to the proposed rule as a result of this comment. Issue 35. In § 86.21, make the State plan eligible for funding under BIG. Response: State plans are eligible activities under the Sport Fish Restoration Act (16 U.S.C. 777).

Issue 36. In accordance with § 86,30, must I allow the public to use grantfunded facilities?

Response: Yes, § 86.30 specifies that reasonable access must be allowed and explains what that means.

Issue 37. In regard to § 86.54, one responder commented that the criterion referenced in § 86.60 was developed for competitive Tier Two projects and was not meant for Tier One projects. Tier One project funding was developed for States meeting the requirements in §§ 86.13 and 86.20.

Response: We agree and will change § 86.54(f)(1)(ii) (now § 86.53(d)(1)(ii) in the final rule) to read "\* \* using the eligibility requirements in §§ 86.14 and 86.20." (Section 86.13 from the proposed rule has become § 86.14 in the final rule.)

Issue 38. In regard to § 86.55(f), why are two sets of \$100,000 proposals required? Why not require one \$200,000

proposal?

\*\*Response: Tier One proposals in the first grant cycle will be funded by 2 different fiscal years. To provide additional clarity, we replaced the second sentence in § 86.55(f) (now § 86.54(f)) with wording similar to the following, "We will fund one proposal for each fiscal year provided that each proposal meets the eligibility requirements in §§ 86.13 (now 86.14) and 86.20."

Issue 39. Sections 86,20 and 86,21 conflict in useful life of some outputs.

Response: We do not agree. Principal structures expected to last 20 years or more are different from navigational aids which may have a shorter lifespan. Only principal structures must be designed to last at least 20 years, therefore a conflict does not exist.

Issue 40. In regard to § 86.56, after the Service awards funds for a project, if it is found not to meet compliance requirements, where will these funds so?

Response: If a State cannot bring the project into compliance according to § 86.56 (now § 86.55), the funds may revert as required by the Act (16 U.S.C. 777c).

Issue 41. Regarding § 86.60(b), a respondent recommended considering construction in remote areas in the scoring priority and adding a statement to the Regulatory Flexibility Act section to give criteria points for remote sites.

Response: Section 86.60(b)(6) and (7) considers remote areas as they relate to significant links to prominent

destinations, way-points, and national, regional, and local significant areas. Therefore, we do not see the need to alter the present guidance

Issue 42. In §86.60(b)(8), how will we assess "significant positive economic

Response: States should assess the economic impact of projects using a cost/benefit analysis as described in the example in § 86.60(b)(8). States should address this issue, along with all the other criteria, in the grant proposal.

Issue 43. In §86,82(b), replace the

word "slippage" with "deviation."
Response: "Slippage" is the term used in 43 CFR 12.80(b)(2)(ii), which provides reporting requirements for assistance programs. We are making no changes to the proposed rule as a result of this comment.

Issue 44. In § 86.80, remove the requirement for quarterly reports.

Response: We are removing the requirement for quarterly reporting for Tier One projects only, according to 43 CFR 12.80, Monitoring and Reporting Program Performance.

Issue 45. In §86.91, make States give credit to the Sport Fish Restoration

Program for BIG grants.

Response: Credit to the Sport Fish Restoration Program remains optional. We have nothing on which to base making this issue mandatory. We are changing "may" in the last sentence to \* are encouraged to \*

Issue 46. Could priority for services at tie-up facilities be given to transient boats over 26 feet in length? If so, how would States enforce the priority?

Response: No. While we intend the program to benefit transient, nontrailerable boats, States must give equal public access to all boaters. Consequently, enforcement is unnecessary

Issue 47. The Service Director should convene a nongovernmental advisory committee to provide expertise on "recreational boating facilities needs."

Response: Under the Federal Advisory Committee Act ( Pub. L. 92-463, 5 U.S.C. App.1), requirements for creating an advisory committee would significantly delay the distribution of the first cycle grant funds. However, opportunities may exist for the nongovernmental community to participate in the grant selection process, and the Service will investigate such opportunities.

Issue 48. Are maintenance and operation costs in remote areas eligible under § 86.20(4)?

Response: We cover maintenance under § 86.20(a); we specifically exclude operation and janitorial costs under § 86.21.

Issue 49. Provide sufficient time to rank and award proposals,

Response: We changed the application period for the first grant cycle to allow applicants 90 days to submit grant proposals to the appropriate Service Regional Office. This change occurs in §86.50.

# Regulatory Planning and Review (E.O.

In accordance with the criteria in Executive Order 12866, this rule is a significant regulatory action according to the following:

- (a) This rule will not have an annual economic effect of \$100 million or adversely affect an economic sector. productivity, jobs, the environment, or other units of government. Program funds total \$8 million per year for 4 years, for a total of \$32 million. Program funds for surveys total \$1,050,000. States must match these amounts with 25 percent or \$2 million per year. State match totals \$8 million over 4 years. The program will not negatively affect an economic sector, productivity, jobs, and other units of government. The program will have a positive effect on these factors.
- (b) This rule will not create inconsistencies with other agencies' actions. We will require the necessary Federal, State, and local reviews and permits before allowing construction of all facilities approved under the program. These reviews will ensure that this rule will not create inconsistencies with other agencies' actions.
- (c) This rule will materially affect entitlements, grants, user fees, loan programs, or the rights and obligations of their recipients. This rule establishes a new grant program using existing funds that are otherwise available to State natural resource agencies under the Sport Fish Restoration Act (16 U.S.C., 777-777m). Recipients will voluntarily accept all stipulations prior to the Service awarding funds for facility construction. The program only obligates the recipient to maintain the facility. User fees are not mandatory and the program allows only enough charges to maintain the facility established by the grant.
- (d) This rule will not raise novel legal or policy issues. This program will award funds to States to install facilities for transient nontrailerable boats. We will review all actions for compliance with the National Environmental Policy Act. This grant program is similar to past Federal Aid grant programs for construction of facilities.

94

#### Regulatory Flexibility Act

The Department certifies that this document will not have a significant economic effect on a substantial number of small entities under the Regulatory Flexibility Act (5 U.S.C. 601 et seq.). Eight million dollars will be available annually for a 4-year period. The effects of these regulations occur to agencies in the States, Puerto Rico, Guarn, the Virgin Islands, American Samoa, the District of Columbia, and the Northern Mariana Islands. These are not small entities under the Regulatory Flexibility Act. Some small entities, mainly marina operators, may receive grant funds. The program will place facilities where none exist now, in remote areas where no competition exists, and in populated areas where marinas have not previously installed them, Employment, investment, productivity, and innovation should all increase because the program will construct facilities to attract transient boaters. The result will be a net transfer of expenditures in the area, U.S.-based enterprises' ability to compete with foreign-based enterprises should not be affected by this rule.

#### Small Business Regulatory Enforcement Fairness Act (SBREFA)

This regulation is not a major rule under 5 U.S.C. 804(2), the Small Business Regulatory Enforcement Fairness Act, as discussed in the Regulatory Planning and Review and Regulatory Flexibility Act sections

#### Unfunded Mandates Reform Act

This regulation does not impose an unfunded mandate on State, local, or tribal governments or the private sector of more than \$100 million per year. This regulation does not have a significant or unique effect on State, local, tribal governments, or the private sector. The rule establishes a grant program that States may participate in voluntarily. A statement containing the information required by the Unfunded Mandates Reform Act (2 U.S.C. 1531 et seq.) is not necessary.

### Takings (E.O. 12630)

In compliance with Executive Order 12630, this regulation does not have significant takings implications. The rule provides standardized procedures for administering a Federal discretionary grant program.

#### Federalism Assessment (E.O. 13132)

In compliance with Executive Order 13132, this regulation does not have sufficient Federalism implications to warrant the preparation of a Federalism Assessment. Through this regulation,

eligible States will receive grants for construction, renovation, maintenance of boating facilities, and public information and education programs. Therefore, the rule is consistent with Executive Order 13132.

#### Civil Justice Reform (E.O. 12988)

In compliance with Executive Order 12988, the Office of the Solicitor has determined that this regulation does not unduly burden the judicial system and meets the requirements of §§ 3(a) and 3(b)(2) of the Order.

#### Paperwork Reduction Act

The information collection requirements contained in this regulation are only those necessary to fulfill applicable grant application requirements of 43 CFR Part 12; see 43 CFR 12.4 for information concerning OMB approval of those requirements. We have submitted the collections of information contained in this rule to the Office of Management and Budget for approval as required under 44 U.S.C. 3501 et seq. The Service will not collect this information until OMB has granted us approval. Additionally, no person may be required to respond to a collection of information unless it displays a currently valid OMB number.

The information collection requirement related to the surveys has a current OMB Approval Number 1018-0106 under the provisions of 44 U.S.C. 3501 et seq. In response to the comments received on the technical aspects of the proposed survey, we revised the survey improving the guidance, questions, and a lower time burden for completion.

What Intergovernmental Review Procedures Must I as a State follow?

Executive Order 12372, "Intergovernmental Review of Federal Programs," and 43 CFR Part 9, "Intergovernmental Review of Department of the Interior Programs and Activities," applies to the national BIG Program. Under the Order, you may design your own processes to review and comment on proposed Federal assistance under covered programs.

What Is My Responsibility as a State if I Participate in the Executive Order Process Having Single Points of Contact?

You should alert your Single Points of Contact (SPOC) to the prospective applications and receive any necessary instructions to provide material the SPOC requires. You must submit all required materials, if any, to the SPOC and show the date of this submittal (or the date of contact if the SPOC does not

require submittal) on the narrative. If you are from a State that chooses to exempt the grants, you need take no action regarding E.O. 12372.

The principal author of this rule is Steve Farrell, Project Officer, U.S. Fish and Wildlife Service, Division of Federal Aid, 4401 North Fairfax Drive, Suite 140, Arlington, Virginia 22203.

#### Regulation Promulgation

Accordingly, the Service hereby establishes part 86, subchapter F of Chapter I, Title 50 Code of Federal Regulations, as set forth below:

#### List of Subjects in 50 CFR Part 86

Administrative practice and procedure, Boats and boating, Grant programs-recreation, Natural resources, Recreation and recreation areas, Reporting and record keeping requirements.

For the reasons set out in the preamble, we amend Subchapter F of Chapter I, Title 50 of the Code of Federal Regulations, by adding a new part 86 to read as follows.

#### PART 86-BOATING INFRASTRUCTURE GRANT (BIG) PROGRAM

#### Subpart A-General Information About the Grant Program

86.10 What does this regulation do? What does the national BIG Program 86.11

86.12 Definitions of Terms Used in Part 86 86.13 What is boating infrastructure?

86.14 Who may apply for these grants? 86.15 How does the grant process work? 86.16 What are the information collection

## Subpart B-Funding State Grant Proposals

86.20 What activities are eligible for funding?

requirements?

86.21 What activities are ineligible for funding?

#### Subpart C-Public Use of the Facility

86.30 Must I allow the public to use the

grant-funded facilities? 86.31 How much money may I charge the public to use tie-up facilities?

#### Subpart D-Funding Availability

86.40 How much money is available for

grants? 86.41 How long will the money be available?

86.42 What are the match requirements? 86.43 May someone else supply the match? 86.44 What are my allowable costs?

86.45 When will I receive the funds?

# Subpart E-How States Apply for Grants

86.50 When must I apply?

86.51 To whom must I apply?

86.52 What information must I include in my grant proposals?

86.53 What are funding tiers?

86.54 How must I submit proposals?

86.55 What are my compliance requirements with Federal laws, regulations, and policies?

#### Subpart F-How the Service Selects Projects To Receive Grants

86.60 What are the criteria used to select projects for grants?

86.61 What process does the Service use to

select projects for grants? 86.62 What must I do after my project has been selected?

86.63 May I appeal if my project is not selected

#### Subpart G-How States Manage Grants

\$6.70 What are my requirements to acquire, install, operate, and maintain real and

personal property? 86.71 How will I be reimbursed?

86.72 Do any other Federal requirements

apply to this program? 86.73 What if I do not spend all the money?

86.74 What if I need more money?

# Subpart H-Reporting Requirements for the

86.80 What are my reporting requirements

for this grant program? 86.81 When are the reports due? 86.82 What must be in the reports?

#### Subpart I-State Use of Signs and Sport Fish Restoration Symbols

86.90 What are my responsibilities for information signs?

86.91 What are my program crediting responsibilities? 86.92 Who can use the SFR logo?

86.93 Where should I use the SFR logo?

86.94 What crediting language should I use?

#### Subpart J-Service Completion of the National Framework

86.100 What is the National Framework? 86.101 What is the Service schedule to

adopt the National Framework? 86.102 How did the Service design the National Framework?

#### Subpart K-How States Will Complete a Boat Access Needs Survey (Survey)

86.110 What does the survey do?

86.111 Must I do a survey? 86.112 What are the advantages of doing a survey? 86.113 What if I have recently completed a

survey?

86.114 Do I need to conduct a survey if I already have a plan for installing tie-up facilities?

86.115 How should I administer the survey? 86.116 May I change the questions in the survey?

86.117 [Reserved] 86.118 What does this survey include?

#### Subpart L-Completing the Comprehensive National Assessment

86.120 What is the Comprehensive National Assessment? 86.121 What does the Comprehensive

National Assessment do?

86.122 Who completes the Comprehensive National Assessment?

86.123 Comprehensive National Assessment schedule. 86.124 What are the Comprehensive

National Assessment products?

#### Subpart M-How States Will Complete the State Program Plans

86.130 What does the State program plan

do?

86.131 Must I do a plan? 86.132 What are the advantages to doing a

plan? 86.133 What are the plan standards? 86.134 What if I am already carrying out a plan?

86.135 [Reserved]

86.136 What must be in the plan? 86.137 What variables should I consider?

Authority: 16 U.S.C. 777g, 777g-1.

#### Subpart A—General Information About the Grant Program

#### §86.10 What does this regulation do?

In this part, the terms "I." "you." "my," and "your" refer to the State agency seeking participation in the national Boating Infrastructure Grant (BIG) Program. "We" and "us" refers to the Fish and Wildlife Service. This part establishes your requirements under the Sportfishing and Boating Safety Act of

- (a) Participate in the national BIG Program,
- (b) Complete your boat access survey, and
- (c) Develop State plans to install tieup facilities for transient nontrailerable recreational vessels

# §86.11 What does the national BIG Program do?

This program provides funds for States to construct, renovate, and maintain tie-up facilities with features for transient boaters in vessels 26 feet or more in length, and to produce and distribute information and educational materials about the program.

# § 86.12 Definitions of terms used in part 86.

For the purposes of this part, the following terms are defined:

Construct means engaging in activities that produce new capital improvements and increase the value or usefulness of existing property. These activities include building new tie-up facilities or replacing or expanding existing tie-up

Grant means financial assistance the Federal Government awards to an eligible applicant.

Grant agreement means a contractual agreement used to obligate Federal Aid funds for carrying out work covered by an approved grant proposal.

Maintain means engaging in activities that allow the facility to continue to function, such as repairing docks. These activities exclude routine janitorial activities

Navigable waters means waters connected to or part of the jurisdictional waters of the United States that transient nontrailerable recreational vessels currently use or can use.

Nontrailerable recreational vessels mean motorized boats 26 feet or more in length manufactured for and operated primarily for pleasure, including vessels leased, rented, or chartered to another person for his or her pleasure.

Project means a specific plan or design.

Proposal means a description of one or more projects for which a State

requests grant funds.

Recreational waters means navigable waters that vessels use for recreational purposes.

Renovate means to rehabilitate or repair a tie-up facility to restore it to its original intended purpose, or to expand its purpose to allow transient nontrailerable recreational vessels.

States means individual States within the United States, the District of Columbia, the Commonwealth of Puerto Rico, Guarn, American Samoa, the Virgin Islands, and the Commonwealth of the Northern Mariana Islands.

Survey instrument means a tool developed by the Service and approved by OMB to assess the need for boating facilities.

Tie-up facilities mean facilities that transient nontrailerable recreational vessels occupy temporarily, not to exceed 10 consecutive days; for example, temporary shelter from a storm; a way station en route to a destination; a mooring feature for fishing; or a dock to visit a recreational, historic, cultural, natural, or scenic site.

Transient means passing through or by a place, staying 10 days or less.

Water-body means the lake, section of river, or specific area of the coast, such as a harbor or cove, where tie-up facilities or boat access sites are located.

### §86.13 What is boating infrastructure?

Boating infrastructure refers to features that provide stopover places for transient nontrailerable recreational vessels to tie up. These features include, but are not limited to:

- (a) Mooring buoys (permanently anchored floats designed to tie up nontrailerable recreational vessels):
- (b) Day-docks (tie-up facilities that do not allow overnight use);
- (c) Navigational aids (e.g., channel markers, buoys, and directional information);

(d) Transient slips (slips that boaters with nontrailerable recreational vessels occupy for no more than 10 consecutive

(e) Safe harbors (facilities protected from waves, wind, tides, ice, currents, etc., that provide a temporary safe anchorage point or harbor of refuge during storms);

(f) Floating docks and fixed piers;

- (g) Floating and fixed breakwaters; (h) Dinghy docks (floating or fixed platforms that boaters with nontrailerable recreational vessels use for a temporary tie-up of their small boats to reach the shore);
  - (i) Restrooms;
  - (j) Retaining walls;
  - (k) Bulkheads;
  - (l) Dockside utilities;
  - (m) Pumpout stations;
  - (n) Recycling and trash receptacles;
  - (o) Dockside electric service; (p) Dockside water supplies;
  - (q) Dockside pay telephones;
  - (r) Debris deflection booms; and
  - (s) Marine fueling stations.

#### §86.14 Who may apply for these grants?

You, with authority from your State Government. You must identify one key contact only and submit proposals through this person.

#### §86.15 How does the grant process work?

To ensure that grants address the highest national priorities identified in the Act, we make funds available on a competitive basis. You must submit your proposals by the appropriate date as specified in §86.50. You must address certain questions and criteria (listed in § 86.52) to be eligible and competitive. We will conduct a panel review of all proposals, and the Service Director will make the final grant awards. You may begin work on your project only after you receive a fully executed grant agreement.

#### §86.16 What are the information collection requirements?

This part contains both routine information collection and survey requirements, as follows:

- (a) The routine information collection requirements for grants applications and associated record keeping contained in this part are only those necessary to fulfill applicable requirements of 43 CFR part 12. These requirements include record keeping and reporting requirements. See 43 CFR 12.4 for information concerning OMB approval of those requirements.
- (b) The revised information collection requirements related to the surveys will be submitted to OMB for approval as changed. They will not be imposed until

we receive OMB approval under the provisions of 44 U.S.C. 3501 et seq. The surveys are voluntary and are for States to determine the adequacy, number, location, and quality of facilities that provide public access for all sizes of recreational boats. The public's burden estimate for the survey is as follows:

Type of information		Average time required per response (minutes)	Annual burden hours
Boat owners: Part A Boat owners: Part B Boat Service Providers: Part C Boat Service Providers: Part D		12 12 20 20	2,240 5,600 2,800 1,333

<sup>\*</sup>These numbers are not additive since some boaters will fill out both Parts A and B, and most of the providers will fill out both Parts C and D.

(c) Send comments regarding this collection of information to the Service Information Collection Clearance Officer, MS-222 ARLSQ, Fish and Wildlife Service, Washington, DC 20240, and the Office of Management and Budget, Department of Interior, Desk Officer, 1849 C Street, NW., Washington, DC 20503. An Agency may not conduct or sponsor, and a person is not required to respond to, a collection of information unless it displays a currently valid OMB control number. OMB has up to 60 days to approve or disapprove the information collection but may respond after 30 days. Therefore, to ensure maximum consideration, you must send your comments to OMB by the above referenced date.

#### Subpart B—Funding State Grant Proposals

#### §86.20 What activities are eligible for funding?

Your project is eligible for funding if you propose to:

(a) Construct, renovate, and maintain either publicly or privately owned boating infrastructure tie-up facilities. To be eligible you must:

- (1) Build these tie-up facilities on navigable waters, available to the public. You must design new construction and renovations to last at least 20 years:
- (2) Design these tie-up facilities for temporary use for transient nontrailerable recreational vessels;
- (3) Build these tie-up facilities in water deep enough for nontrailerable recreational vessels to navigate (a minimum of 6 feet of depth at the lowest tide or other measure of lowest fluctuation);
- (4) Provide security, safety, and service for these boats; and,
- (5) Install a pumpout station, if you construct a facility for overnight stays:
- (i) If there is already a pumpout within a reasonable distance (generally within 2 miles) of the facility, you may not need one;

- (ii) For facilities intended as day stops, we encourage you to install a pumpout; and,
- (iii) You may use funds from the BIG program, or the Clean Vessel Act pumpout grant program also administered by us, to pay for a numerous station

pumpout station.
(b) Do one-time dredging only, to give transient vessels safe channel depths between the tie-up facility and maintained channels or open water.

- (c) Install navigational aids, limited to giving transient vessels safe passage between the tie-up facility and maintained channels or open water.
- (d) Apply funds to grant administration.
- (e) Fund preliminary costs: (1) Preliminary costs may include any of the following activities completed
- before signing a grant agreement: (i) Conducting appraisals; (ii) Administering environmental
- reviews and permitting;
  (iii) Conducting technical feasibility studies, for example, studies about environmental, economic, and construction engineering concerns;
- (iv) Carrying out site surveys and engaging in site planning;
   (v) Preparing cost estimates; and
- (vi) Preparing cost estimates, and (vi) Preparing working drawings, construction plans, and specifications.
- (2) We will fund preliminary costs only if we approve the project.
- (3) If the project is approved, the appropriate Service Regional Director must still approve preliminary costs.
- (f) Produce information and education materials such as charts, cruising guides, and brochures.

# § 86.21 What activities are ineligible for funding?

Your project is ineligible for funding if you propose to: (a) Complete a project that does not

- (a) Complete a project that does not provide public benefits, for instance, a project that is not open to the public for use:
- (b) Involve law enforcement activities;
  (c) Significantly degrade or destroy valuable natural resources or alter the cultural or historic nature of the area;

- (d) Construct or renovate principal structures not expected to last at least 20 years;
- (e) Do maintenance dredging; (f) Fund operations or routine, custodial, and janitorial maintenance of the facility;
- (g) Construct, renovate, or maintain boating infrastructure tie-up facilities for nontrailerable vessels, for example the following:
- the following:
  (1) Tie-up slips available for occupancy for more than 10 consecutive days by a single party;
  - (2) Dryland storage;
  - (3) Haul-out features; and
- (4) Boating features for trailerable or "car-top" boats (boats less than 26 feet in length), such as launch ramps and carry-down walkways.
- (h) Develop a State program plan to construct, renovate, and maintain boating infrastructure tie-up facilities; and
- (i) Conduct surveys to determine boating access needs.
- (1) You may conduct the survey with funds allocated to motorboat access to recreational waters under subsection (b)(1) of section 8 of the Federal Aid in Sport Fish Restoration Act of 1950, as amended (16 U.S.C. 777).
- (2) You may combine surveys under one contractor where feasible if you can realize a cost or other savings.

#### Subpart C-Public Use of the Facility

#### § 86.30 Must I allow the public to use the grant-funded facilities?

(a) You must allow reasonable access to all recreational vessels for the useful life of the tie-up facilities. Accessible to the public means located where the public can reasonably reach the facility and where all boats typical to that facility can easily use it, charging equitable fees, and being open for reasonable periods. You must allow public access to the shore and basic features such as fuel and restrooms in facilities that have them. You must specify precise details of the public access in the contract with the facility manager. We do not require public access to the remainder of a park or marina where the facility is found. Nor do we require any restrictions in that park or marina.

(b) You must comply with Americans with Disabilities Act requirements when you construct or renovate all tie-up facilities under this grant.

#### § 86.31 How much money may I charge the public to use tie-up facilities?

You may charge the public only a reasonable fee, based on the prevailing rate in the area. You must determine a fee that does not pose an unreasonable, competitive amount, based on other publicly and privately owned tie-up facilities in the area. You must approve any proposed changes in fee structure by a sub-grantee.

#### Subpart D-Funding Availability

# §86.40 How much money is available for grants?

There is \$32 million available for grants under the BIG program (\$8 million per year for fiscal years 2000– 2003).

# §86.41 How long will the money be available?

Under the Act, funding for the BIG program is provided for FY 2000–2003.

Each year's funds remain available for obligation for a total of three fiscal years (e.g. FY 2000 funds will remain available through FY 2002) (16 U.S.C. 777c(b)(3)(B)).

#### §86.42 What are the match requirements?

The Act authorizes the Secretary of the Interior (through the Director of the U.S. Fish and Wildlife Service (Service)) to award grants to States to pay up to 75 percent of the cost to construct, renovate, or maintain tie-up facilities for transient nontrailerable recreational vessels. You or a partner must pay the remaining project cost—at least a 25 percent match is required. Title 43 CFR 12.64 applies to cost sharing or matching requirements for Federal grants.

# § 86.43 May someone else supply the match?

Third-party contribution, including property and in-kind services, is allowable, but must be necessary and reasonable to accomplish grant objectives. In-kind contributions must also represent the current market value of noncash contributions that the third party furnishes as part of the grant.

#### § 86.44 What are my allowable costs?

(a) The State may spend grant funds to pay only costs that are necessary and reasonable to accomplish the approved grant objectives. Grant costs must meet the applicable Federal cost principles in 43 CFR 12.62. You may purchase informational and program signs as allowable costs.

(b) If you include purposes other than those eligible under the Act, you must prorate the costs equitably according to Federal cost principles in 43 CFR 12.62 and 50 CFR 80.15.

#### § 86.45 When will I receive the funds?

Once you sign the grant agreement, we will make the funds available.

#### Subpart E—How States Apply for Grants

#### §86.50 When must I apply?

(a) We will accept proposals between February 20, 2001, and May 18, 2001, for the first grant cycle; between July 1, 2001, and September 30, 2001, for the second grant cycle; and between July 1, 2002, and September 30, 2002, for the third grant cycle. This program starts fiscal year 2000 and ends fiscal year 2003. We will have \$16 million to award the first grant cycle, and \$8 million each cycle after that.

(b) The annual schedule follows:

Action	FY 2000-2001	FY 2002	FY 2003
We announce the beginning of the grant cycle.	February 20, 2001	July 1, 2001	July 1, 2002.
You submit your grant proposal by		September 30, 2001	
Regions submit the proposals to Washington by.	June 18, 2001	October 31, 2001	October 31, 2002.
We rank the proposals by	July 17, 2001	November 30, 2001	November 30, 2002. December 31, 2002.
Regions finalize their grant agree- ments by.	October 15, 2001	February 28, 2002	February 28, 2003.

#### §86.51 To whom must I apply?

You must submit your proposals to the appropriate regional office of the U.S. Fish and Wildlife Service. See the chart below for the address you will need.

Region	State	Address	Telephone
1	California, Hawaii, Idaho, Nevada, Oregon, Washington, American Samoa, Com- monwealth of the Northern Mariana Is- lands, and Guam.	Division of Federal Aid, U.S. Fish & Wildlife Service, Eastside Federal Complex, 911 NE 11th Avenue, Portland, OR 97232— 4181.	503-231-6128, Fax: 503-231-6996
2	Arizona, New Mexico, Oklahoma, and Texas.	Division of Federal Aid, U.S. Fish & Wildlife Service, P.O. Box 1306, 625 Silver, SW, Suite 325, Albuquerque, NM 87102.	505-248-7450, Fax: 505-248-7471
3	Illinois, Indiana, Iowa, Michigan, Minnesota, Missouri, Ohio, and Wisconsin.	Division of Federal Aid, U.S. Fish & Wildlife Service, Bishop Henry Whipple Federal Building, 1 Federal Drive, Fort Snelling, MN 55111–4058.	612-713-5130, Fax: 612-713-5290
4	Alabama, Arkansas, Florida, Georgia, Ken- tucky, Louisiana, Mississippi, North Caro- lina, South Carolina, Tennessee, Puerto Rico, and the Virgin Islands.	Division of Federal Aid, U.S. Fish & Wildlife Service, 1875 Century Boulevard, Suite 324, Atlanta, Georgia 30345.	404-679-4159, Fax: 404-679-4160

Region	State	Address	Telephone
5	Connecticut, Delaware, District of Colum- bia, Maine, Maryland, Massachusetts, New Hampshire, New Jersey, New York, Pennsylvania, Rhode Island, Vermont, Virginia, and West Virginia.	Division of Federal Aid, U.S. Fish & Wildlife Service, 300 Westgate Center Drive, Hadley, MA 01035–9589.	413-253-8200, Fax: 413-253-8487
6	Colorado, Kansas, Montana, Nebraska, North Dakota, South Dakota, Utah, and Wyoming.	Division of Federal Aid, U.S. Fish & Wildlife Service, Derwer Federal Center, P.O. Box 25486, Lake Plaza North Building, 134 Union Boulevard, 4th Floor, Denver, Colorado 80225.	303-236-7392, Fax: 303-236-8192
7	Alaska	Division of Federal Aid, U.S. Fish & Wildlife Service, 1011 East Tudor Road, Anchor- age, Alaska 99503.	907-786-3435, Fax: 907-786-3575

# § 86.52 What information must I include in my grant proposals?

You must submit the following standard forms and narrative for all projects (Tier One and Tier Two) (see § 86.53):

(a) Standard Form 424 series as prescribed by the Office of Management and Budget. The SF 424 series consists of the Applications for Federal Assistance (SF 424), Budget Information—Non-Construction Programs (SF 424A), Assurances—Non-Construction Programs (SF 424B), Budget Information—Construction Programs (SF 424C), and Assurances—Construction Programs (SF 424C). Submit forms appropriate for either construction or nonconstruction projects. Forms are available from your appropriate Service Regional Office.

(b) Information requested under OMB Circular A-102 (Application Booklet for Federal Aid Grants—pending approval under the Paperwork Reduction Act).

#### §86.53 What are funding tiers?

- (a) This grant program will consist of two tiers of funding.
- (i) You may apply for one or both
- (ii) The two tiers will allow all States some certainty of base level funding.
- (b) Tier One funding will ensure broad geographical distribution to meet the needs of transient nontrailerable recreational vessels.
- (c) Tier Two funding will allow States with large projects to compete with other States with large projects based on individual project merits.
- (d) We describe the two tiers as follows:
  - (1) Tier One Projects.
- (i) You may submit a proposal with an unlimited number of projects within this tier. However, your total request cannot exceed \$100,000 of Federal funds for any given fiscal year.

- (ii) Tier One projects must meet the eligibility requirements in §§ 86.14 and 86.20.
  - (2) Tier Two Projects.
- (i) While we expect available funds for Tier Two proposals to be between \$3 million and \$4 million per grant cycle, we have no dollar limit for Tier Two proposals. You may submit any number of projects, which we will score and rank separately according to the criteria in \$86.60.
- (ii) Each project will compete nationally against every other project in Tier Two.
- (iii) Tier Two projects must also meet the eligibility requirements in §§ 86.14 and 86.20.

#### §86.54 How must I submit proposals?

- (a) You may apply for either Tier One funding or Tier Two funding or both.
- (b) You may submit more than one project proposal within Tier One and Tier Two.
- (c) You may submit one proposal that includes Tier One and Tier Two projects.
- (d) If your proposal includes Tier One and Tier Two projects, you must describe Tier One projects separately from Tier Two projects.
- (e) You must describe each project in Tier Two separately, so that the Service can rank and score each project in Tier Two separately.
- (f) For the first grant cycle, which includes fiscal years 2000 and 2001, a State may submit one Tier One proposal not to exceed \$100,000 per fiscal year. States should submit proposals between February 20, 2001, and May 18, 2001. We will fund one Tier One proposal per State for each fiscal year provided that each proposal meets the eligibility requirements in §§ 86.14 and 86.20. Fiscal year 2000 funds are available only for Tier One proposals. Tier One proposals need not meet the criteria in § 86.60. We will fund Tier Two

- proposals received between February 20, 2001, and May 18, 2001, that meet the criteria in §§ 86.14, 86.20, and 86.60 with fiscal year 2001 funds and the remainder of fiscal year 2000 funds.
- (g) For the remaining grant cycles, you may submit only one proposal of Tier One projects per fiscal year.
- (h) When we approve projects, the appropriate Service Regional Office will determine how many grant agreements are necessary.

#### § 86.55 What are my compliance requirements with Federal laws, regulations, and policies?

- (a) To receive Federal funds, you must agree to and certify compliance with all applicable Federal laws, regulations, and policies. You must submit an Assurance Statement, as described in 43 CFR part 12.51(c), that describes how you comply with Federal grant requirements; and
- (b) You may have to provide additional documentation to comply with environmental and other laws, as defined in Fish and Wildlife Service Manual 523 FW 1 (available on the internet at http://www.fws.gov/directives/523fw1.html). The Service Regional Office Grant Administrator may request preliminary evidence at the grant proposal stage that the proposed project will meet these compliance requirements. Consult with the appropriate Service Regional Office for specific applicability.

#### Subpart F—How the Service Selects Projects To Receive Grants

# § 86.60 What are the criteria used to select projects for grants?

- (a) We will rank all Tier Two proposals according to the criteria in paragraph (b) of this section and the attached chart, which sets forth points we will ascribe for various factors.
  - (b) We will consider proposals that:

(1) Plan to construct, renovate, and maintain tie-up facilities for transient nontrailerable recreational vessels following priorities identified in your State's program plan (see Subpart M for State program plan information) that the Secretary of the Interior has approved under section 7404(c) of the Sportfishing and Boating Safety Act. (2) Provide for public/private and public/public partnership efforts to develop, renovate, and maintain tie-up facilities.	15 points.
These partners must be other than the Service and lead State agency: (i) One partner	5 points.
(ii) Two parlners	10 points.
(iii) Three or more partners	15 points.
(3) Use innovative techniques to increase the availability of tie-up facilities for transient nontrailemble recreational vessels (includes education/information).	0-15 points.
(4) Include private, local, or other State funds in addition to the non-Federal match, described in § 86.42:  (i) Twenty-six percent to thirty-five percent	5 points.
(ii) Between thirty-six and forty-nine percent	10 points.
(iii) Fifty percent and above	15 points.
(5) Are cost efficient. Proposals are cost efficient when the tie-up facility or access site's features add a high value compared with the funds from the proposal, for example, where you construct a small feature such as a transient mooring dock within an existing harbor that adds high value and opportunity to existing features (restrooms, utilities, etc.). A proposal that requires installing all of the above features would add less value for the cost.	0-10 points.
(6) Provide a significant link to prominent destination way points such as those near metropolitan population centers, cultural or natural areas, or that provide safe harbors from storms.	10 points.
(7) Provide access to recreational, historic, cultural, natural, or scenic opportunities of national, regional, or local significance. Projects that provide access to opportunities of national, regional, or local significance receive 5 points for each, for a maximum of 15 points.	
(8) Provide significant positive economic impacts to a community. For example, a project that costs \$100,000 and attracts a number of bosters who altogether spend \$1 million a year in the community.	1-5 points.
(9) Include multi-State efforts that result in coordinating location of tie-up facilities	5 points.
(10) Total possible points	105 points.

Criteria	Points
(1) Construct Tie-up Facilities	15
(2) Provide Partnership Efforts	5-15
(3) Use Innovative Techniques (4) Include Other Funding	0-15
Sources	5-15
(5) Are Cost Efficient	0-10
(6) Provide Way Point Linkage (7) Provide Access To Oppor-	10
tunities	5-15
nomic Impacts	1-5
(9) Include Multi-State Efforts	5
(10) Total Possible Points	105

#### §86.61 What process does the Service use to select projects for grants?

The Service's Division of Federal Aid convenes a panel of professional staff to review, rank, and recommend funding to the Service Director. This panel will include representatives from the Service's Washington, DC, and Regional Offices. The Director may convene an advisory panel of nongovernmental organizations to advise and make recommendations to the Federal panel. The Service Director will select projects for grants by August 16, 2001, August 10, 2001, and August 10, 2002, for the three grant cycles.

#### § 86.62 What must I do after my project has been selected?

After we approve your award, we will notify you to work with the appropriate Service Regional Office to fulfill the grant documentation requirements and finalize the grant agreement.

#### § 86.63 May I appeal if my project is not selected?

If you have a difference of opinion over the eligibility of proposed activities or differences arising over the conduct of work, you may appeal to the Director. Final determination rests with the Secretary of the Interior (50 CFR 80.7).

#### Subpart G—How States Manage Grants

#### § 86.70 What are my requirements to acquire, install, operate, and maintain real and personal property?

(a) You will find applicable regulations for this subject in 43 CFR 12.71 and 12.72. If you have questions about applicability, contact the appropriate Service Regional Office.
(b) You must ensure that the design

and installation of tie-up facilities provide for substantial structures that will have a significant longevity, at least 20 years.

(c) You must ensure that you operate, maintain, and use the tie-up facilities and features for the stated grant purpose. You must obtain prior written approval from the appropriate Service Regional Director before you can convert these tie-up facilities to other

#### §86.71 How will I be reimbursed?

For details on how we will pay you, refer to 43 CFR 12.61.

# § 86.72 Do any other Federal requirements apply to this program?

For administrative requirements not covered under these specific guidelines. check 43 CFR 12, which generally applies to all Federal grant programs.

#### §86.73 What if I do not spend all the money?

Funds not obligated or expended after 3 fiscal years from the date of the award revert to the Secretary of Transportation for use in State recreational boating safety programs. (16 U.S.C. 777c(b)(3)(B), 16 U.S.C. 777c(b)(4))

#### §86.74 What if I need more money?

Funds for grants are available only on a competitive basis. Therefore, if you need more money, you must compete in the next grant cycle.

# Subpart H—Reporting Requirements for the States

# § 86.80 What are my reporting requirements for this grant program?

(a) For all projects, you must submit to the appropriate Service Regional Office an annual report and a final performance report and otherwise comply with 43 CFR 12.80.

(b) For Tier Two projects, you must submit quarterly reports according to 43 CFR 12.80.

#### §86.81 When are the reports due?

Reports are due as follows: (a) Annual reports are due 90 days

after the grant year ends;

(b) The final performance report is due 90 days after the expiration or termination of grant support;

(c) Tier Two quarterly reports are due January 31, April 30, July 31, and October 31 unless specified otherwise in the grant agreement; and

(d) The State must report certified percentage of completion data and other significant developments in accordance

with the grant agreement or 43 CFR 12.80.

#### §86.82 What must be in the reports?

The reports must include the following:

- (a) You must identify the actual accomplishments compared to the objectives established for the period;
- You must identify the reasons for any slippage if established objectives were not met; and
- (c) You must identify any additional pertinent information including, when appropriate, analysis and explanation of cost overruns or high unit costs.

#### Subpart I—State Use of Signs and Sport Fish Restoration Symbols

#### § 86.90 What are my responsibilities for information signs?

You should install appropriate information signs at boating infrastructure tie-up facilities. You should ensure that this information is clearly visible, directing boaters to the facility. Information should show fees, restrictions, hours of operation, a contact name, and telephone number to report an inoperable facility.

# §86.91 What are my program crediting responsibilities?

You should give public credit to the Federal Aid in Sport Fish Restoration (SFR) program as the source of funding for the BIC Program. You should recognize this program by using the SFR logo. You are encouraged to use the crediting logo identified in § 80.26 of this chapter to identify national BIC Program projects.

#### § 86.92 Who can use the SFR logo?

The States may use the SFR logo and should encourage others to display it. Other people or organizations may use the logo for purposes related to the national BIG Program as authorized in § 80.26 of this chapter.

#### §86.93 Where should I use the SFR logo?

You should display the logo on tie-up facilities that you construct, acquire, develop, or maintain under these grants. You should also use the logo on printed material or other visual representations that relate to project accomplishments or education/information. Refer to § 85.47 of this chapter for logo colors.

# §86.94 What crediting language should I use?

Suggested examples of language to use when crediting the national BIG Program follow:

(a) Example 1: The Federal Aid in Sport Fish Restoration Program funded this facility thanks to your purchase of fishing equipment and motorboat fuels.

- (b) Example 2: The Federal Aid in Sport Fish Restoration Program is funding this construction thanks to your purchase of fishing equipment and motorboat fuels. And.
- (c) Example 3: The Federal Aid in Sport Fish Restoration Program funded this (pamphlet) thanks to your purchase of fishing equipment and motorboat fuels

#### Subpart J—Service Completion of the National Framework

### §86.100 What is the National Framework?

The National Framework is the survey, required by the Act, you must use to determine boating access needs in your State. Through a State survey, you must conduct a boating access needs assessment or data collection to determine the adequacy, number, location, and quality of tie-up facilities and boat access sites providing access to recreational waters for all sizes of recreational boats.

#### § 86.101 What is the Service schedule to adopt the National Framework?

The Secretary of the Interior adopted the National Framework on September 28, 2000 via a notice in the Federal Register (Volume 65, Number 189, Page 58284).

#### §86.102 How did the Service design the National Framework?

The Framework divides the survey into two components; boater survey, and boat access provider survey.

- (a) The purpose of the boater survey component is to identify boat user preferences and concerns for existing and needed access available to the public.
- (1) The nontrailerable boat data set will fulfill informational needs for you to develop your State program plans as called for in the Act.
- (2) The boater survey will survey registered boat owners in your State for two types of boats:
- (i) Part A—for nontrailerable recreational vessels.
- (ii) Part B—for trailerable and "cartop" boats (less than 26 feet long).
- (b) The purpose of the boat access provider component is to identify boat access providers' ideas about current and needed facility and site locations and perceptions of boat user preferences and concerns regarding access. We developed these questions to guide interviews of boat access facility and site managers.
- (1) The nontrailerable boat data set will fulfill the informational needs for you to develop your State plans as called for in the Act.

- (2) The boat access provider survey will survey facility providers in your State for two types of boats:
- (i) Part C—a survey to all providers in your State who allow public access, including State agency and non-State entities (Federal and local government entities, corporate and private/ commercial providers), and operate tieup facilities for nontrailerable recreational vessels.
- (ii) Part D—a survey to all providers in your State who allow public access and operate boat access sites for boats less than 26 feet long.

#### Subpart K—How States Will Complete Access Needs Surveys

#### § 86.110 What does the State survey do?

The State survey determines the status of boating access facilities for all recreational boats in your State and your future boater access needs.

#### §86.111 Must I do a survey?

The Act does not require surveys. They are voluntary. However, if you do a survey, you must complete it following the National Framework to receive funds. You must transmit the results to the Service Regional Offices in a common electronic format, such as Microsoft Word, Word Perfect, Excel or Ouattro Pro.

# §86.112 What are the advantages of doing a survey?

Surveys provide the information necessary to fully understand the needs of boaters in your State. Surveys allow you to develop a meaningful plan to provide better access to boaters. Use surveys to complete the plan.

#### § 86.113 What if I have recently completed a boat access survey?

If the recent survey substantially answers the provisions in § 86.118, the appropriate Service Regional Office will determine if it is sufficient to meet the needs of the program. If the Regional Office determines that the survey is not sufficient, you must complete that portion(s) or an entire new survey to receive credit for completing a recent survey.

#### § 86.114 Do I need to conduct a survey if I already have a plan for installing tie-up

You need not conduct the survey if the appropriate Secretary of the Interior certifies that you have developed and are carrying out a State program plan, as described in Subpart M of this chapter, that ensures that public boat access exists and is adequate to meet the needs of recreational boaters on your waters.

#### §86.115 How should I administer the survey?

Use a consultant or university specializing in administration of such surveys. Use sample sizes large enough to achieve statistical accuracy so the estimate is within plus or minus 10 percent of the true number.

(a) You may use a telephone, mail, or other type of survey for a sample population of boaters within the State. Costs for telephone and mail surveys are roughly similar. However, response rates for mail surveys are generally lower.

(b) For boat access providers, we prefer that you survey all State agency and non-State providers, but you may survey a sample population.

(c) You may develop your own methodology to collect data, which may include telephone, mail, fax, or other inventory means. We do not expect you to use automated, electronic, mechanical, or similar means of information collection.

(d) Data collected are unique to each respondent. Data collection should use standard survey method criteria to gather information from each respondent.

# §86.116 May I change the questions in the survey?

You must not change the questions because we need information that is comparable nationwide. We have developed a survey instrument for completing the surveys. We are seeking approval from OMB on the survey questions and the OMB approval does not extend to additional questions.

#### §86.117 Reserved for survey approval schedule.

# §86.118 What does this survey instrument include?

(a) We divided this survey into four parts. Part A being for transient nontrailerable boat owners. Part B is for trailerable or "car-top" boat owners. Part C is for State agency and non-State providers of facilities for nontrailerable recreational vessels in the State. Part D is for State and non-State providers of access sites for trailerable or "car-top" boats.

(b) Follow these instructions to complete Part A—BOAT OWNER SURVEY FOR TIE-UP FACILITIES FOR NONTRAILERABLE RECREATIONAL VESSELS:

 If the boater owns a nontrailerable recreational vessel, ask the boater to fill out Part A;

(2) If the boater owns more than one boat 26 feet or more in length, ask the boater to provide information for the boat he or she uses most often; (3) If the boater owns at least one boat more than and at least one boat less than 26 feet in length, ask the boater to fill out both Parts A and B: and.

(4) You should use a sample size large enough to achieve statistical accuracy so the estimate is within 10 percent of the true number.

(c) Follow these instructions to complete Part B—BOAT OWNER SURVEY FOR TRAILERABLE OR "CAR-TOP" BOAT ACCESS SITES:

(1) If the boater owns a boat less than 26 feet long, ask the boater to fill out Part B:

(2) If the boater owns more than one boat less than 26 feet long, ask the boater to provide information for the boat he or she uses most;

(3) If the boater owns at least one boat more than and at least one boat less than 26 feet in length, ask the boater to complete both Parts A and B; and,

(4) You should use a sample size large enough to achieve statistical accuracy so the estimate is within 10 percent of the true number.

(d) Parts C and D are surveys for providers of tie-up facilities and boat access sites. Part C is for State agency and non-State providers of facilities for nontrailerable recreational vessels in the State. Part D is for State and non-State providers of boat access sites for boats less than 26 feet in length.

(e) Follow these instructions to complete Part C—STATE AGENCY AND NON-STATE PROVIDER SURVEY FOR TIE-UP FACILITIES:

(1) Ask State agency and non-State providers of tie-up facilities for nontrailerable recreational vessels to fill out Part C.

(2) If more than one State agency manages these facilities, send this survey to all of those agencies.

(3) If the State agency or non-State provider awards grants to others who provide facilities, ask these grantees to respond for these facilities instead of the State agency or non-State provider.

(4) If a State agency or non-State provider operates facilities and sites for both nontrailerable and trailerable boats, ask the provider to fill out both Parts C and D.

(5) Ask State agency and non-State providers to identify all tie-up facilities.

(6) For all questions, if you need additional space, make copies of the appropriate page.

(f) Follow these instructions to complete Part D—STATE AGENCY AND NON-STATE PROVIDER SURVEY FOR TRAILERABLE BOAT ACCESS SITES.

 Ask State agency and non-State providers of access sites for boats less than 26 feet long to fill out Part D. (2) Non-State providers include the Federal Government, local government, corporations, private owners, and others.

(3) If more than one State agency manages these sites, send this survey to all of them.

(4) If the State agency or non-State provider awards grants to others who provide sites, ask these grantees to respond for these sites instead of the State agency or non-State provider.

(5) If a State agency or non-State provider operates facilities and sites for both nontrailerable and trailerable boats, ask the provider to fill out both Parts C and D.

(6) We prefer that the State agency or non-State provider identify all boat access sites and water-bodies, but if he or she has many sites and water-bodies, the provider may group the information together rather than identify each site individually.

(7) For all questions, if you need additional space, make copies of the appropriate page.

#### Subpart L—Completing the Comprehensive National Assessment

#### §86.120 What is the Comprehensive National Assessment?

The Comprehensive National Assessment is a national report integrating the results of State boat access needs and facility surveys.

#### § 86.121 What does the Comprehensive National Assessment do?

The Comprehensive National Assessment determines nationwide the adequacy, number, location, and quality of public tie-up facilities and boat access sites for all sizes of recreational boats.

#### §86.122 Who completes the Comprehensive National Assessment?

The Service completes the Assessment. We will develop standards in consultation with the States.

#### §86.123 Comprehensive National Assessment schedule.

Using the results from the State surveys, the Service will compile the results and produce the Comprehensive National Assessment by September 30, 2003.

#### § 86.124 What are the Comprehensive National Assessment products?

The Comprehensive National Assessment products are:

(a) A single report, including the following information:

 A national summary of all the information gathered in the State surveys.

- (2) A table of States showing the results of the information gathered.
- (3) One-page individual State summaries of the information.
- (4) Appendices that include the survey questions, and names, addresses, and telephone numbers of State contacts.
- (5) An introduction, background, methodology, results, and findings.
  - (6) Information on the following:
- Boater trends, such as what types of boats they own, where they boat, and how often they boat.
- (ii) Boater needs, such as where facilities and sites are now found, where boaters need new facilities and boat access sites, and what changes of features boaters need at these facilities and sites. And
  - (iii) Condition of facilities.
- (b) Summary report abstracting important information from the final national report. And
- (c) A key findings fact sheet suitable for widespread distribution.

#### Subpart M—How States Will Complete the State Program Plans

# § 86.130 What does the State program plan do?

The State program plan identifies the construction, renovation, and maintenance of tie-up facilities needed to meet nontrailerable recreational vessel user needs in the State.

#### § 86.131 Must I do a plan?

The Act does not require plans. Plans are voluntary. However, if you do a plan, you must complete it following these regulations.

# § 86.132 What are the advantages to doing a plan?

Plans provide the information necessary to fully understand the needs of boaters operating nontrailerable recreational vessels in your State. The plan will make you more competitive when you submit grants under this program. We will give you 15 points for having an approved plan.

#### § 86.133 What are the plan standards?

You must hase State program plans on a recent, completed survey following the National Framework.

# §86.134 What if I am already carrying out a plan?

You need not develop a program plan if we certify that you have developed and are carrying out a plan that ensures public boat access is and will be adequate to meet the needs of recreational boaters on your waters.

# § 86.135 Reserved for plan approval schedule.

#### §86.136 What must be in the plan?

The plan must:

 (a) Identify current boat use and patterns of use.

- (b) Identify current tie-up facilities and features open to the public and their condition.
- (c) Identify boat access user needs and preferences and their desired locations. Include repair, replacement, and expansion needs and new tie-up facilities and features needed.
- (d) Identify factors that inhibit boating in specific areas, such as lack of facilities, or conditions attached that inhibit full use of facilities. Identify strategies to overcome these problems.
- (e) Include information about the longevity of current tie-up facilities.

#### §86.137 What variables should I consider?

You should consider the following variables:

- (a) Location of population centers,
- (b) Boat-based recreation demand,
- (c) Cost of development,
- (d) Local support and commitment to maintenance,
  - (e) Water-body size,
- (f) Nature of the fishery and other resources,
- (g) Geographic distribution of existing tie-up facilities,
- (h) How to balance the need for new tie-up facilities with the cost to maintain and improve existing facilities.

Dated: December 8, 2000.

#### Kenneth L. Smith,

Acting Assistant Secretary for Fish and Wildlife and Parks.

[FR Doc. 01-951 Filed 1-17-01; 8:45 am] BILLING CODE 4310-55-P

# PART 6

# **DRAWINGS**